Halfspace Matching: a Domain Decomposition Method for Scattering by 2D Open Waveguides

Der Fakultät für Mathematik des Karlsruher Instituts für Technologie zur Erlangung des akademischen Grades eines Dr. rer. nat.

genehmigte Dissertation

von Herr Dipl.-Math.techn. Julian Ott aus Augsburg

Datum der Einreichung: 26.04.2017

Referent: Prof. Dr. Andreas Kirsch Koreferent: Prof. Dr. Willy Dörfler Koreferentin: Prof. Dr. Sonia Fliss

Tag der mündlichen Prüfung: 24.05.2017

Abstract

We study a scattering problem for the Helmholtz equation in 2D, which involves nonparallel open waveguides, by means of the halfspace matching method. This method has formerly been applied to periodic media and homogeneous anisotropic media, and we extend it to open waveguides. It allows the reformulation of the Helmholtz equation in an exterior domain to a set of equations for particular traces of the solution, reducing the overall dimension of the problem by 1, making it accessible for numerical discretisation. We show the well-posedness of the halfspace matching method for a model problem in the exterior of a triangular domain, assuming the presence of absorption. Furthermore, we introduce a numerical discretisation which allows the realisation of transparent boundary conditions by a system of coupled integral equations. To illustrate the practicality of this method, we study a number of optimisation examples involving junctions of open waveguides by means of material optimisation.

Acknowledgements

This thesis is result of my work at the Karlsruhe Institute of Technology from 2013 to 2017. It started with financial support of the DFG (Deutsche Forschungsgesellschaft) in the Research Training Group 1294 "Analysis, Simulation and Design of Nanotechnological Processes", and was continued afterwards with financial support of the state Baden-Württemberg. Later on, it was embedded in the DFG Collaborative Research Center 1173 on "Wave Phenomena" at KIT. I am very grateful for the excellent environment provided by this CRC and the faculty of Mathematics at KIT.

Foremost, I want the to thank the supervisor of this thesis, Andreas Kirsch, for discussions, advice, and the freedom I enjoyed during research. It feels rather fortunate to have easily accessible advice at hand, and I greatly appreciated it. This is also addressed to Frank Hettlich and Tilo Arens, who were of great help with discussions and advice. I also want to thank the remainder of the Workgroup for Inverse Problems for the pleasant atmosphere and the cake.

As mentioned above, this thesis was associated to the CRC 1173 of the DFG, and in particular project C4 on "Modeling, design and optimization of 3D waveguides", which helped me to put at least some relevance to this work. I want to thank Willy Dörfler for agreeing to act as second examiner of this thesis, as well as discussions and input on many aspects of this work.

Let me also thank a number of my fellow PhD-students, in particular Patrick Krämer, David Hipp, Pascal Kraft, Uwe Zeltmann and Felix Hagemann, for all the small and large discussions on mathematics and all the world and his wife.

Last but not least, I have to thank Anne-Sophie Bonnet-ben Dhia, Sonia Fliss, Christoph Hazard and Antoine Tonnoir: this thesis is nothing but a small extension of their works. My short research stay at ENSTA, Paris, in the spring of 2016 provided critical inputs and ideas, and is a main cause why thesis takes the form it has today. I hope my work will do justice to theirs. I am very grateful that Sonia Fliss agreed to be the third examiner of this thesis.

I want to close with a personal thank to all the living mathematicians, as well as the dead. Ever so often when reading up on a new topic, I was filled with awe in view of the ingenuity, thought, and elegance i glimpsed.

Julian Ott Karlsruhe, April 2017

Contents

1	Intr	oduction	9		
	1.1	Introduction	9		
	1.2	Notation and a few Basics	13		
	1.3	The Limit-Absorption Principle	15		
	1.4	Well-Posedness for Absorptive Media	18		
		-			
2	The	e Expansion Theorem	21		
	2.1	Introduction and References	21		
	2.2	The Operator A and Spectral Family \mathcal{F}_A	22		
	2.3	The Pekeris Profile	29		
	2.4	Proof of the Expansion Theorem	32		
2 Halfanago Brobloma		fenaco Problems	51		
J	2 1	Introduction and References	51		
	0.1 2.0	The Solution Operator S	52		
	0.2 2.2	The Model Perpresentation \mathcal{O}_{Wg}	57		
	0.0 2.4	The Divisiblet to Neumann Operator	60		
	0.4 25	A Laplace Problem	62		
	ວ.ວ ວິດ	The DtD Operators of S	02 79		
	5.0	The DtD Operators of \mathcal{S}_{wg}	12		
4	Hal	fspace Matching	79		
	4.1	Introduction and References	79		
	4.2	The Problem of Interest and Notations	79		
	4.3	Fredholmness for the Exterior of a Triangle	83		
5	Implementation for the Full Space				
0	5.1	Introduction	89		
	5.2	The Sequilinear Form of the Problem	90		
	5.2	The Non-absorptive Case Incident Waves	93		
	5.0	Discretisation	90		
	0.1		55		
6	Opt	imisation of Waveguide Junctions	109		
	6.1	Introduction	109		
	6.2	Incident Fields in the Absorptive Case	109		
	6.3	The Fréchet Derivative	112		
	6.4	Energy Conservation	116		
	6.5	Optimisation	121		
	6.6	Examples	123		
		-			

CONTENTS

1 Introduction

1.1 Introduction

Open waveguides have become a rather ubiquitous element in many technical applications, in particular in the context of telecommunication technology. Formerly being used mostly for long-distance data communication in the form of optical fibres, their use has been extended to smaller scales. In the last few years, optical chip-to-chip interconnects have been demonstrated to work[54], and the field of integrated nanophotonics is steadily growing.

The dissertation on hand is motivated by these developments, but the situation considered is a much simpler, reduced model distantly related to this topic. Why this?

The critical feature of an *open optical waveguide* is the existence of so-called *guided* modes. These are *time-harmonic* solutions to Maxwell's equations with *unboundedly* supported perturbations of the material coefficients (see [70]). The question of interest is the following: what happens if waveguides of different shape and orientation are linked by some structure?

Within the previous sentences, the mathematical modelling has already been determined: one should study *time-harmonic* Maxwell's equation in \mathbb{R}^3 with *unboundedly supported* and *non-separable* perturbations of the homogeneous background material. The key problem is the following. Time-harmonic Maxwell equations need to be completed by a *radiation condition* to ensure uniqueness: in the case of an homogeneous free space, the appropriate one is the Silver–Müller radiation condition (see, for example, [62]).

For open waveguides, this radiation condition will fail, that is, it will not ensure existence and uniqueness any more. How to replace it? The answer to this question is not entirely clear. A number of radiation conditions have been proposed, but so far, no complete analytical study is available (compare also Subsection 1.1.2 below).

So we withdraw from Maxwell's equations before even writing them down, and turn our attention to a simpler problem, which retains many of the problematic features of the Maxwell problem, but gives us more tools to treat it in a tractable fashion: we will consider the *time-harmonic* wave equation in \mathbb{R}^2 with *unboundedly supported*, *nonseparable* perturbations.

For a given incident field $u_{\rm inc}$, we want to consider the problem of finding a solution to

 $\begin{cases} \Delta u(x) + p(x)u(x) = 0 & \text{for } x \in \mathbb{R}^2, \\ u - u_{\text{inc}} & \text{fulfils some radiation condition,} \end{cases}$

where the function p is strictly larger than 0, and takes only 2 values $p_1 > p_0 > 0$, on the domains as sketched in Figure 1.1. Now again arises the question: what is the proper radiation condition?



 $p(x) = p_0$



To avoid creating false hope amongst our readers: we do not give an answer to this question.

We will define a formal radiation condition in Chapter 5, but are far from actually giving a well-posedness result for this problem. It is the problem we wanted to study, but were not able to. However, we will give an overview over results for special cases of this problem in the following section, to convince the reader that it is not as easy at it may seem.

We will instead study a radically simpler problem, namely: find a solution $u \in H^1(\mathbb{R}^2)$ to

$$\Delta u_{\epsilon}(x) + (p(x) + i\epsilon)u_{\epsilon}(x) = f(x) \quad \text{for } x \in \mathbb{R}^2, \tag{1.1.1}$$

where $f \in L^2(\mathbb{R}^2)$ is some given, compactly supported source, and $\epsilon > 0$ is some (small) absorption parameter. Note that the radiation condition vanished, since it can be replaced by the condition $u \in H^1(\mathbb{R}^2)$. This problem is radically simpler, since we switched into the resolvent set of $\Delta + p$. It can be seen to be well-posed by different arguments: we will prove the well-posedness in a variational setting with the help of the Lax-Milgram theorem in Section 1.4.

Is this a suitable replacement for the scattering problem? It is by definition. Let us elaborate a bit on this point: in many cases (in particular the physical literature), radiation conditions are vaguely introduced by claiming that they "select outgoing waves" or prohibit "sources at infinity". This explanation still sounds very imprecise and vague to the author, and luckily, there are well established alternatives. A way of rigorously justifying radiation conditions is the limit-absorption principle: a solution u to the Helmholtz equation $\Delta u + pu = f$ is called outgoing, if the solution u_{ϵ} of (1.1.1) converges locally to u as $\epsilon \to 0$, $\epsilon > 0$.

The limit-absorption principle is the de-facto standard to justify radiation conditions in a multitude of geometries, since it is applicable in very general settings. For this work, it serves the important task of justifying that we treat the absorptive problem. We will give a short introduction into this principle below in Section 1.3.

For the absorptive problem we will derive an alternative formulation, which admits a numerical implementation. This alternative formulation is called *halfspace matching*, the title of our work. To the authors knowledge, it has been first used by Fliss and Joly in [30] for periodic media, and later by Besse et al [8] for hexagonal media. Later it



Figure 1.2: The decomposition of the domain \mathbb{R}^2 into four, overlapping subdomains: the bounded triangle, and the three (hatched) halfspaces $\Omega_0, \Omega_1, \Omega_2$.

has gotten more analytical justification by Tonnoir in [74] (a further publication being prepared [75]). Our main goal is to extend this method to open waveguides. While our study is mostly analytical, we will also show that it works on the numerical side, but only in the form of "experimental numerics".

1.1.1 Overview. Before we outline the remainder of this thesis, let us shortly describe the method of halfspace matching: consider the solution u_{ϵ} of (1.1.1) for some $\epsilon > 0$. We start by decomposing the full plane into four overlapping sub-domains $\mathbb{R}^2 = \Omega \cup \Omega_0 \cup$ $\Omega_1 \cup \Omega_2$, where Ω is a triangle, and the three domains $\Omega_0, \Omega_1, \Omega_2$ are halfspaces with the boundaries $\Gamma_n = \partial \Omega_n, n \in \{0, 1, 2\}$. Figure 1.2 shows how this decomposition of \mathbb{R}^2 is done. We gain the following: for halfspaces containing waveguides, there is an explicit solution formula, which maps the Dirichlet trace on the boundary Γ_n to the solution on Ω_n . Accordingly, if this formula is available, one only needs to determine the Dirichlet traces of the three halfspace boundaries Γ_0 , Γ_1 and Γ_2 and the solution in the triangle Ω . With the help of the solution formula, one also obtains a set of integral equations called *compatibility equations*— which the traces have to fulfil.

This allows to reduce the problem for the solution on the exterior $\mathbb{R}^2 \setminus \overline{\Omega}$ to a set of integral equations for the traces on the halfspace boundaries, while the problem inside the bounded triangle remains the same. Here, two main questions arise. Firstly: are the resulting compatibility equations well posed? Secondly: can these equations be discretised for a numerical computation of solutions?

The goal of this dissertation is to introduce the method and answer the two questions we just posed. Let us quickly describe how we aim to proceed in this matter.

Chapter 1 Here we will introduce elementary notation and confine the area of topics we expect the reader to be familiar with. Furthermore, a section on the limit-absorption principle is included, were we show that the Sommerfeld radiation condition is insufficient for open waveguides. We will also apply the lemma of Lax–Milgram to show well-posedness for absorptive problems.

Chapter 2 To give the representation formula for a halfspace containing a waveguide, we need a tool of utmost importance for the study of open waveguides: the spectral family associated with the reduced Helmholtz operator. We have decided to include this spectral family together with a complete proof in Chapter 2, since the literature offers a number of representations, which still require some work to obtain the representation as given here. At first reading, most of this chapter can be skipped (in particular the long Section 2.4 containing the proof of the expansion theorem).

- **Chapter 3** Having the spectral family at hand, it is rather easy to derive the solution formula for a halfspace. To make this discussion rigorous, a careful analysis is required, which is the topic of Chapter 3. Here we will encounter the Dirichlet-to-Dirichlet operators, which will be the operators involved in the compatibility equations, and prove a number of analytical auxiliary theorems.
- **Chapter 4** At this point, we will finally consider the compatibility equations for the exterior of a triangle. The analytical main result is contained here: the compatibility equations for this case are well-posed in a particular Sobolev space, and consequently, are equivalent to the Helmholtz equation in the exterior of a triangle.
- **Chapter 5** Here we will describe the numerical discretisation we use, and show a number of convergence examples. Furthermore, we will discuss the non-absorptive case briefly here, and speculate shortly about radiation conditions.
- **Chapter 6** To give a few colourful pictures, we will give a few ingredients to treat topology/material optimisation problems of waveguide junctions. A few optimisation examples will be shown, and we perform a few design studies.

1.1.2 On scattering in open waveguides. Let us quickly give an overview on works related to scattering in stratified media/open waveguides. In many cases, the two terms are synonymous, since a certain stack of layered media always acts as an open waveguide. Let us start by the monographs of Wilcox [78] and Weder [76], who treat time-dependent scattering in stratified media. The analysis has been refined later on, for example by Debiévre and Pravica [23, 24, 25] and Christiansen [17]. These works already make heavy use of spectral families as the one we introduce in Chapter 2, but deal mostly with time-dependent scattering.

There are a number of works on time-harmonic scattering in the same geometry. There is some study of open waveguides in the Russian literature available (see for example [48, 64] and references therein), which, however, offers no complete analysis of existence and well-posedness.

Rather recently, a few different ansätze have been developed, which allow a proper well-posedness analysis, the first one being contained in works by Xu [79, 80], which has later been extended to 3D geometries [57].

Another strand of work started with the paper by Magnanini and Santosa [58], where they study the Green's function of an open waveguide, which was later refined and turned into a radiation condition for locally perturbed open waveguides by Ciraolo and Magnanini [19, 20], which was extended later on by Ciraolo [18].

Most influential on this dissertation is another group of works: employing the spectral family, Bonnet Ben-Dhia, Hazard and others define a modal radiation condition, which has been employed for defects of open waveguides in 2D [9], as well as for a junction problem of two different parallel waveguides [10]. Similar arguments have also been applied to 3D waveguides by Hazard [40].

We ought to mention the paper by Hanckes and Nédélec [46], where asymptotics of solutions in stratified media are considered.

We close this section by remarking one important point, which complicates this thesis and forces us to consider the absorptive case: none of the well-posedness analysis of the previous references is easily generalisable to *non-parallel* waveguides as shown in Figure 1.1, despite rising interest in this topic.

1.2. NOTATION AND A FEW BASICS

1.1.3 Numerical methods for open waveguides. To approximate solutions of scattering problems numerically, a common strategy is the following: one truncates the domain and uses some kind of transparent boundary condition to emulate the free space, while the truncated domain is treated, for example, by finite elements. The most commonly used method, also in the context of open waveguides, is the perfectly matched layer (PML), introduced by Berenger [5], which has been later shown by Chew and Weedon [16] to be equivalent to a complex scaling of the exterior solution. For open waveguides, there are some hybrid formulations, which truncate the open waveguide by a PML to obtain a closed waveguide, for which one can then apply mode matching methods (see [35] and references therein).

A newer method are Hardy-space infinite elements (HSIE). Based on the Pole condition [42, 43] by Schmidt, the Hardy space method was already shown to work with closed waveguides [38] and exhibits good convergence for open waveguides [63].

Both PML and HSIE admit much freedom for waveguides in the exterior, and in principle allow the numerical implementation of situations as shown in Figure 1.1.

Also available are boundary integral equation methods in layered media to solve scattering problems, such as the works by Michalski and Zheng [60, 61]. They are, however, restricted to the case of locally perturbed stratified media.

For the readers interested how our method compares to those mentioned, we refer to Subsection 5.4.8 at the end of Chapter 5.

1.2 Notation and a few Basics

We start by listing a number of notations we will frequently use.

Number related notation		
\mathbb{N}	The natural numbers	
\mathbb{Z}	The set of integer numbers	
$\mathbb{Z}/m\mathbb{Z}$	The ring of integers modulo $m, m \in \mathbb{N}$	
\mathbb{R}	The real numbers	
\mathbb{C}	The complex numbers	
\overline{z}	The conjugate of a complex number or function $z \in \mathbb{C}$	
z	The Euclidean norm of a complex number or vector $z\in \mathbb{C}^d$	
$z_1 \cdot z_2$	Dot product $z_1 \cdot z_2 = \sum_{k=1}^d z_1 z_2$, for $z_1, z_1 \in C^d$	
Tenclorical notation		
Topological notation		
\overline{D}	Closure of a set (if the topology is clear)	

D	Closure of a set (if the topology is clear)
$\operatorname{cl}_{\ \cdot\ _X} D$	Closure of D with respect to the indicated norm
\simeq	Equivalence of norms

Derivatives

$\partial_x f$	Derivative of f with respect to x
Δf	Laplacian of f
∇f	Gradient of f
$d\mathcal{F}$	Frechét-derivative of the operator \mathcal{F}

1.2.1 Definition. Throughout this thesis, the square root $\sqrt{\cdot} : \mathbb{C} \to \mathbb{C}$ denotes the square root with branch cut along the positive real axis $\{z \in \mathbb{C} : \operatorname{Re}(z) \ge 0, \operatorname{Im}(z) = 0\}$. For $z = re^{i\theta}$ with $r \ge 0$ and $\theta \in [0, 2\pi)$, the square root is given by

$$\sqrt{z} = \sqrt{r}e^{i\theta/2}.$$

 $\sqrt{\cdot}$ is holomorphic on $\mathbb{C} \setminus [0, \infty)$, and for $y \ge 0$ the limit coming from the upper halfspace exists and fulfils

$$\lim_{\substack{\epsilon \to 0 \\ \epsilon > 0}} \sqrt{y + i\epsilon} = \sqrt{y}.$$

This implies in particular

$$\operatorname{Re}(i\sqrt{\lambda}) \leq 0 \text{ for all } \lambda \in \mathbb{C},$$

and $\operatorname{Re}(i\sqrt{\lambda}) = 0$ if and only if $\lambda \ge 0$.

1.2.2 Derivatives, function spaces. Let us start by introducing the L^2 spaces: if $\Gamma \subset \mathbb{R}^N$ is some (not necessarily smooth) $M \leq N$ -dimensional manifold, we denote by

$$L^{2}(\Gamma) := \{ f : \Gamma \to \mathbb{C} : \|f\|_{\Gamma} < \infty \},\$$

the space of (Lebesgue-)measurable functions, whose $L^2(\Gamma)$ scalar product is defined by

$$\langle f,g \rangle_{\Gamma} = \int_{\Gamma} f(x)\overline{g}(x) \,\mathrm{d}s(x),$$

with the corresponding norm

$$||f||_{\Gamma} := \sqrt{\langle f, f \rangle_{\Gamma}}.$$

Here we denoted by ds the *M*-dimensional surface measure of Γ . Occasionally, we will also encounter weighted spaces: let $\mu : \Gamma \to (0, \infty)$ be some weight, then

$$||f||_{L^{2}(\Gamma,\mu ds)}^{2} := \int_{\Gamma} |f(x)|^{2} \mu(x) ds(x),$$

and denote by $L^2(\Gamma, \mu \, \mathrm{d}s) = \{f : \Gamma \to \mathbb{C} : \|f\|_{L^2(\Gamma, \mu \, \mathrm{d}s)} < \infty\}$ the corresponding L^2 space. In the case that N = M (so that Γ is actually a domain), we denote $\mathrm{d}x = \mathrm{d}s(x)$.

For the remainder of function spaces defined here, we will give the corresponding notations as used in McLean's book [59], if the reader wants to recheck the precise definition and construction. Let $\Omega \subset \mathbb{R}^N$ be some Lipschitz domain. If $f : \Omega \to \mathbb{C}$ is some function, we denote for any multi-index $\alpha \in \mathbb{N}^N$ the derivative with respect to α by

$$\partial_{\alpha}^{|\alpha|} f(x) := \partial_{x_1}^{\alpha_1} \dots \partial_{x_N}^{\alpha_N} f(x) := \frac{\partial^{\alpha_1 + \dots + \alpha_N} f}{\partial x_1^{\alpha_1} \dots \partial x_N^{\alpha_N}} (x)$$

Note that in general, we use the notion of distributional derivatives. Here we denote by $|\alpha| = \alpha_1 + \ldots + \alpha_N$ the order of a multi-index. We then denote by

$$C_0^{\infty}(\Omega) = \{ f : \Omega \to \mathbb{C} : \text{ supp}(f) \subset \Omega, \ \partial_{\alpha}^{|\alpha|} f \text{ is continuous for all } \alpha \in \mathbb{N}^N \},$$

the space of smooth functions with compact support in Ω , which corresponds to the space $C^{\infty}_{\text{comp}}(\Omega)$ or $\mathcal{D}(\Omega)$ from [59]. For $s \geq 0$, the Sobolev-spaces $H^{s}(\Omega)$ and $H^{s}_{0}(\Omega)$ are

defined as in [59]. For the construction of the norms, we refer to the reference, and note that the canonical norm for $s = n \in \mathbb{N} \cup \{0\}$ for those spaces is given by

$$\|f\|_{H^n(\Omega)}^2 = \sum_{\substack{\alpha \in \mathbb{N}^N \\ |\alpha| \le n}} \|\partial_{\alpha}^{|\alpha|} f\|_{\Omega}^2.$$

Since we will work intensively with trace spaces, we point to a certain peculiarity which we will need to deal with: if $I = (a, b) \subset \mathbb{R}$ is some (possibly unbounded) interval, we need to consider the spaces

$$H^{1/2}(I)$$
 and $H^{1/2}_{00}(I)$,

where the latter corresponds to the space $\tilde{H}^{1/2}(I)$ in [59]. The introduction of the space $H_{00}^{1/2}(I)$ is necessary, since functions from the space $H_0^{1/2}(I)$ cannot necessarily be extended by 0 to obtain a function¹ in $H^{1/2}(\mathbb{R})$. For more information on this issue, we refer to Subsection 3.5.11. Finally, if $\Gamma \subset \mathbb{R}^N$ is a (N-1)-dimensional, sufficiently smooth sub-manifold of \mathbb{R}^N , we denote by

$$H^{s}(\Gamma)$$

the corresponding Sobolev space on the boundary. Sufficiently smooth means that Γ is at least $C^{k,1}$ -smooth, where $0 \leq s \leq k-1$ and $k \in \mathbb{N} \cup \{0\}$.

1.2.3 Trace operators. Let Ω be some (sufficiently smooth) domain and let $u \in H^s(\Omega)$ be some function. If $\Gamma \subset \overline{\Omega}$ is again some *M*-dimensional, sufficiently smooth submanifold, we denote by

 $u|_{\Gamma}$

the trace of u on Γ , which is well defined as a function in $H^{s_0}(\Gamma)$, provided that $0 \leq s_0 \leq s - (N - M)/2$ (see for example [59, Chapter 3]).

1.3 The Limit-Absorption Principle

Since the limit-absorption principle is of great importance for the remainder of this thesis, this section gives an introduction on a few examples.

1.3.1 The limit-absorption principle. Consider the example of the Helmholtz equation in free space

$$\Delta u(x) + \kappa_0 u(x) = f(x) \qquad \text{for } x \in \mathbb{R}^N, \tag{1.3.1}$$

where $\kappa_0 > 0$ is a fixed parameter and $f \in C_0^{\infty}(\mathbb{R}^N)$ is some given source with compact support. How can one assure uniqueness for (1.3.1)? The answer is classically given by the Sommerfeld radiation condition [71], which states the following: if one looks for solutions of (1.3.1), which additionally fulfil the condition

$$\lim_{r \to \infty} r^{\frac{N-1}{2}} \left(\partial_r - i\sqrt{\kappa_0}\right) u(r\hat{x}) = 0 \qquad \text{for all } |\hat{x}| = 1, \ \hat{x} \in \mathbb{R}^N, \tag{1.3.2}$$

one can show that exactly one solution exists.

¹ Note that one has

$$H_0^{1/2}(I) := \operatorname{cl}_{H^{1/2}(I)} C_0^{\infty}(I) = H^{1/2}(I),$$

see for example [59, Theorem 3.40].

However, as we will see later, the Sommerfeld radiation condition is restricted to the free \mathbb{R}^N with a constant κ_0 , so the following question arises: how can one construct the outgoing solution without the radiation condition?

Or more precisely: if the Sommerfeld radiation condition is not "the right one", what does "outgoing" actually mean?

The answer lies in the limit-absorption principle. It exploits the fact that for any $\epsilon \neq 0$ and given $f \in L^2(\mathbb{R}^N)$, the equation

$$\Delta u_{\epsilon}(x) + (\kappa_0 + i\epsilon)u_{\epsilon}(x) = f(x) \quad \text{for } x \in \mathbb{R}^N,$$

possesses a unique solution in $H^2(\mathbb{R}^2)$ (which we will prove in a more general context in Theorem 1.4.3). The unique outgoing solution can now be constructed by taking the limit $\epsilon \to 0$, coming from above, that is for $\epsilon > 0$. This way, one defines the outgoing solution in the sense of the limit-absorption principle by

$$u(x) = \lim_{\substack{\epsilon \to 0\\\epsilon > 0}} u_{\epsilon}(x) \quad \text{for } x \in \mathbb{R}^{N}.$$

It turns out that this definition is equivalent to the Sommerfeld radiation condition in the free space. This argument can be generalised for many operators and geometries, and is practically the standard tool to justify radiation conditions.

For the work at hand, it mainly allows us a trick which enables us to simplify the analytical situation significantly: while being interested in the case that $\epsilon = 0$, we choose $\epsilon > 0$ and obtain very easily the existence and uniqueness of solutions.

1.3.2 The failure of the Sommerfeld radiation condition for an open waveguide. It seems appropriate to illustrate that the Sommerfeld radiation condition fails for open waveguides. Let us construct a counter example in a very particular case of the equation

$$\Delta u_{\epsilon}(x) + (\kappa_0 + i\epsilon - q(x_1))u(x) = f(x) \quad \text{for } x \in \mathbb{R}^2, \tag{1.3.3}$$

where $\kappa_0 > 0$, and the function q is given by

$$q(x_1) = \begin{cases} -2 & \text{for } |x_1| < \pi/2, \\ 0 & \text{for } |x_1| > \pi/2. \end{cases}$$

We will construct our solution with a separation of variables ansatz: one easily finds that $u(x) = e^{i\sqrt{\kappa_0+1+i\epsilon}x_2}\psi(x_1)$ is a solution to the homogeneous version of (1.3.3), where ψ is given by

$$\psi(x_1) = \begin{cases} \cos(x_1) & \text{for } |x_1| < \pi/2, \\ \frac{e^{\pi/2}}{\sqrt{2}} e^{-|x_1|} & \text{for } |x_1| > \pi/2. \end{cases}$$

Let now $\tilde{f} \in C_0^{\infty}(\mathbb{R})$, and let us consider the source $f(x) = \psi(x_1)\tilde{f}(x_2)$. We try to solve (1.3.3) by the ansatz

$$u_{\epsilon}(x) = v_{\epsilon}(x_2)\psi(x_1).$$

Using that

$$(\partial_{x_1}^2 - q(x_1))\psi(x_1) = \psi(x_1) \quad \text{for } x_1 \in \mathbb{R},$$

we obtain by inserting our ansatz for u_{ϵ} into (1.3.3) that v_{ϵ} must fulfil

$$v_{\epsilon}''(x_2) + (\kappa_0 + 1 + i\epsilon)v(x_2) = \tilde{f}(x_2) \qquad \text{for } x_2 \in \mathbb{R}.$$

This equation can now easily be solved with the Green's function

$$G_{\epsilon}(x_2, y_2) = \frac{1}{-2i\sqrt{\kappa_0 + 1 + i\epsilon}} e^{i\sqrt{\kappa_0 + 1 + i\epsilon}|x_2 - y_2|}.$$

At this point, it is important to recall our choice of the square root in 1.2.1. It is chosen so that $\operatorname{Re}(i\sqrt{\kappa_0 + 1 + i\epsilon}) < 0$, which implies that the exponential in the Green's function is decreasing for $|x| \to \infty$ or $|y| \to \infty$, which indicates that we have chosen the correct root of $\kappa_0 + 1 + i\epsilon$.

Note that this choice is not unique for $\epsilon = 0$: in this case, the only two square roots of $\kappa_0 + 1$ are real, and it is left open which root is the correct one, since both corresponding exponentials do not decay.

Let us get back to finding the function v_{ϵ} . It is given by

$$v_{\epsilon}(x_2) = \int_{\mathbb{R}} G_{\epsilon}(x_2, y_2) \tilde{f}(y_2) \mathrm{d}y_2.$$

One now easily verifies that the solution we obtained

$$u_{\epsilon}(x) = \psi(x_1) \int_{\mathbb{R}} \frac{e^{i\sqrt{\kappa_0 + 1 + i\epsilon} |x_2 - y_2|}}{-2i\sqrt{\kappa_0 + 1 + i\epsilon}} \tilde{f}(y_2) \,\mathrm{d}y_2$$

is in $H^2(\mathbb{R}^2)$, and fulfils (1.3.3), so we have found the solution in the case with absorption $\epsilon > 0$.

Let us now establish the outgoing solution in the sense of the limit-absorption principle: since \tilde{f} is compactly supported by assumption, we can simply take the limit by the dominated convergence theorem in

$$u_0(x) = \lim_{\substack{\epsilon \to 0\\\epsilon > 0}} u_\epsilon(x) = \psi(x_1) \int_{\mathbb{R}} \lim_{\substack{\epsilon \to 0\\\epsilon > 0}} \left(\frac{e^{i\sqrt{\kappa_0 + 1 + i\epsilon} |x_2 - y_2|}}{-2i\sqrt{\kappa_0 + 1 + i\epsilon}} \tilde{f}(y_2) \right) \, \mathrm{d}y_2$$
$$= \psi(x_1) \int_{\mathbb{R}} \frac{e^{i\sqrt{\kappa_0 + 1} |x_2 - y_2|}}{-2i\sqrt{\kappa_0 + 1}} \tilde{f}(y_2) \, \mathrm{d}y_2.$$

Note that this procedure uniquely determines which square root we have to take for $\sqrt{\kappa_0 + 1}$: it must be the positive square root, since it is the limit of $\sqrt{\kappa_0 + 1 + i\epsilon}$ as $\epsilon \to 0, \epsilon > 0$.

We now show that it does not fulfil the Sommerfeld radiation condition: consider the limit $x = (0, x_2)$ as $x_2 \to +\infty$. For x_2 sufficiently large the interval $(-\infty, x_2)$ contains the whole support of \tilde{f} , and we have that

$$u_0(x) = \psi(0) \int_{\supp(\tilde{f})} \frac{e^{i\sqrt{\kappa_0 + 1} (x_2 - y_2)}}{-2i\sqrt{\kappa_0 + 1}} \tilde{f}(y_2) \, \mathrm{d}y_2$$

= $e^{i\sqrt{\kappa_0 + 1} x_2} \int_{\supp(\tilde{f})} \frac{e^{-i\sqrt{\kappa_0 + 1} y_2}}{-2i\sqrt{\kappa_0 + 1}} \tilde{f}(y_2) \, \mathrm{d}y_2$
= $C e^{i\sqrt{\kappa_0 + 1} x_2}.$

where we used that $x_2 > y_2$ for any $y_2 \in \text{supp}(f)$. Let us now check whether or not the Sommerfeld radiation condition (1.3.2) applies. Note that for $x = (x_1, 0)$ the coefficient in the Helmholtz equation (1.3.3) is given by $\kappa_0 + 2$; accordingly, it seems reasonable to choose $\kappa_0 + 2$ or κ_0 instead of κ_0 in the Sommerfeld radiation condition. However, none of the two choices yields the correct result, since we have that

$$|x_2|^{1/2} \left(\partial_{x_2} u_0(0, x_2) - i\sqrt{\kappa_0 + 2} \, u_0(x_2) \right) = |x_2|^{1/2} \left(i\sqrt{\kappa_0 + 1} - i\sqrt{\kappa_0 + 2} \right) C e^{i\sqrt{\kappa_0 + 1} \, x_2},$$

which clearly does not converge to 0, even if we replace $\sqrt{\kappa_0 + 2}$ by $\sqrt{\kappa_0}$.

1.3.3 The limit-absorption principle for open waveguides. Let us review quickly a few works, were the limit-absorption principle has been shown to hold for open waveguides. Weder's monograph [76] deals with this issue for stratified media. Similarly, a paper by Boutet de Monvel-Berthier and Dragos [11] deals with this topic. Rather recently, Kirsch and Lechleiter have shown that the limiting absorption principle holds for stratified periodic open waveguides [51].

Let us remark again that all those works are restricted to local perturbations of stratified layers, and do not admit non-parallel waveguides or junctions of differently shaped waveguides.

Correspondingly, there is no result available which can be applied to the situation we are interested in (as in the Situation of Figure 1.1). Later on, in the part concerning numerical examples, we will assume it to hold nonetheless to have a somewhat formal justification for our method.

1.4 Well-Posedness for Absorptive Media

1.4.1 The problem of interest. Let $\Omega \subset \mathbb{R}^N$ be a (possibly unbounded) Lipschitz domain and $q \in L^{\infty}(\Omega)$ be real-valued. Furthermore, let

$$\kappa \in \mathbb{C} \setminus \mathbb{R}$$

be some arbitrary, non-real complex number.

For given $g \in H^{1/2}(\partial \Omega)$ and $f \in L^2(\Omega)$ we consider the problem of finding a solution $u \in H^1(\Omega)$ of

$$\begin{cases} \Delta u(x) + (\kappa - q(x))u(x) = f(x) & \text{for } x \in \Omega, \\ u(x) = g(x) & \text{for } x \in \partial\Omega. \end{cases}$$
(1.4.1)

This problem has to be understood in the variational sense, that is, we say that u fulfils

$$\Delta u(x) + (\kappa - q(x))u(x) = f(x) \quad \text{for } x \in \Omega,$$

if for any $\psi \in H_0^1(\Omega)$ we have

$$\int_{\Omega} \nabla u(x) \cdot \nabla \overline{\psi}(x) - (\kappa - q(x))u(x)\overline{\psi}(x) \,\mathrm{d}x = -\int_{\Omega} f(x)\overline{\psi}(x) \,\mathrm{d}x.$$
(1.4.2)

The equality u(x) = g(x) for $x \in \partial \Omega$ has to be understood in the sense of trace operators. The proof of the well-posedness relies on the Lax–Milgram theorem, which reads as follows. **1.4.2 Theorem.** Let $b : \mathcal{H} \times \mathcal{H} \to \mathbb{C}$ be a sesquilinear form, that is, linear in the first argument and conjugate linear in the second. Let b be bounded, that is, there exist a constant $c_1 > 0$ such that

$$|b(u,w)| \le c_1 ||u||_{\mathcal{H}} ||v||_{\mathcal{H}} \quad \text{for all } u, v \in \mathcal{H},$$

and let b be coercive, i.e. there exists $c_2 > 0$ such that

$$\operatorname{Re} b(u, u) \ge c_2 \|u\|_{\mathcal{H}}^2$$
 for all $u \in \mathcal{H}$.

Then for any conjugate linear, continuous functional $F : \mathcal{H} \to \mathbb{C}$, there exists exactly one solution of

$$b(u, \psi) = F(\psi)$$
 for all $\psi \in \mathcal{H}$.

Proof. See [59].

1.4.3 Theorem. Under the assumption of Subsection 1.4.1, there exists exactly one solution $u \in H^1(\Omega)$ of (1.4.1), and the solution operator

$$\mathcal{S}: L^2(\Omega) \times H^{1/2}(\partial \Omega) \to H^1(\Omega), \qquad (f,g) \mapsto u,$$

is continuous.

Proof. We start by assuming that $\operatorname{Im}(\kappa) < 0$, since the case $\operatorname{Im}(\kappa) > 0$ can be dealt with analogously. Let us first study a sesquilinear form related to the weak formulation of (1.4.1). Denote $p := -\|q\|_{L^{\infty}(\Omega)}$. We easily obtain that $\arg(q(x)-\kappa) \leq \arg(p-\kappa) \in [0,\pi)$. In particular, it follows that $\alpha := \arg(p-\kappa)$ cannot be equal to π , since $\operatorname{Im}(\kappa) < 0$. This now implies by elementary calculation that

$$\operatorname{Re}\left(e^{-i\alpha/2}(q(x)-\kappa)\right) \ge \sin(\alpha/2)|\operatorname{Im}(\kappa)| \quad \text{for all } x \in \mathbb{R},$$
 (1.4.3)

(note that $\arg\left(e^{-i\alpha/2}(q(x)-\kappa)\right) < \alpha/2$), as well as

$$\operatorname{Re}\left(e^{-i\alpha/2}\right) \ge \sin(\alpha/2).$$
 (1.4.4)

We now define the sesquilinear form $b: H_0^1(\Omega) \times H_0^1(\Omega) \to \mathbb{C}$ by

$$b(v,w) := e^{-i\alpha/2} \int_{\Omega} \nabla v \cdot \nabla \overline{w} + (q-\kappa)u\overline{w} \, \mathrm{d}x.$$

We claim that b is bounded and coercive: by Hölder's and the Cauchy–Schwarz inequality we have

 $|b(v,w)| \le \max\{1, \|\kappa + q\|_{L^{\infty}(\Omega)}\} \|v\|_{H^{1}(\Omega)} \|w\|_{H^{1}(\Omega)}.$

Furthermore, to see that b is coercive, consider that for $v \in H_0^1(\Omega)$ we have

$$\begin{aligned} \operatorname{Re}(b(v,v)) &= \int_{\Omega} \operatorname{Re}\left(e^{-i\alpha/2}|\nabla v|^{2}\right) + \operatorname{Re}\left(e^{-i\alpha/2}(q(x)-\kappa)|v|^{2}\right) \,\mathrm{d}x\\ &\geq \sin\left(\frac{\alpha}{2}\right) \int_{\Omega} |\nabla v|^{2} \,\mathrm{d}x + \sin\left(\frac{\alpha}{2}\right) |\operatorname{Im}(\kappa)| \int_{\Omega} |v|^{2} \,\mathrm{d}x\\ &\geq \sin\left(\frac{\alpha}{2}\right) \min(|\operatorname{Im}(\kappa)|,1) \|v\|_{H^{1}(\Omega)}^{2}, \end{aligned}$$

where we exploited (1.4.3) and (1.4.4).

Let us now consider the problem (1.4.1). We first show uniqueness: let $u \in H^1(\Omega)$ be a solution to (1.4.1) with homogeneous data, that is g = 0, f = 0. From g = 0 we deduce that $u \in H^1_0(\Omega)$, and if we multiply the variational form (1.4.2) by $e^{-\alpha/2}$, we obtain that

$$b(u, \psi) = 0$$
 for all $\psi \in H_0^1(\Omega)$

Since b is bounded and coercive, this implies by the Lax-Milgram Lemma that u = 0.

Let us now show existence. Fix $g \in \dot{H}^{1/2}(\partial \Omega)$ and $f \in L^2(\Omega)$. Let $w \in H^1(\Omega)$ be an extension of g, that is, some arbitrary function $w \in H^1(\Omega)$ such that $w|_{\partial\Omega} = g$. We now define the conjugate linear functional $F : H^1_0(\Omega) \to \mathbb{C}$ by

$$F(\psi) = -e^{-i\alpha/2} \int_{\Omega} \nabla w \cdot \nabla \overline{\psi} + (q-\kappa)w\overline{\psi} + f\overline{\psi} \,\mathrm{d}x$$

Again by the Lax-Milgram Lemma, we have exactly one solution $v \in H_0^1(\Omega)$ of

$$b(v, \psi) = F(\psi)$$
 for all $\psi \in H_0^1(\Omega)$.

It is now easy to show that $u := v + w \in H^1(\Omega)$ is a variational solution of (1.4.1). To see the continuity of S, we note that by the Cauchy–Schwarz inequality we have

$$\begin{aligned} \|u\|_{H^{1}(\Omega)}^{2} &\leq \frac{1}{c_{2}} |\operatorname{Re} b(u, u)| \\ &\leq \frac{1}{c_{2}} |F(u)| \\ &\leq \frac{1}{c_{2}} (\|w\|_{H^{1}(\Omega)} + \|f\|_{L^{2}(\Omega)}) \|u\|_{H^{1}(\Omega)} \\ &\leq \frac{1}{c_{2}} (\tilde{C} \|g\|_{H^{1/2}(\partial\Omega)} + \|f\|_{L^{2}(\Omega)}) \|u\|_{H^{1}(\Omega)}, \end{aligned}$$

where we used that there exists a constant \tilde{C} independent of f such that $||w||_{H^1(\Omega)} \leq \tilde{C}||g||_{H^{1/2}(\partial\Omega)}$, see [59, Theorem 3.37].

2 The Expansion Theorem

2.1 Introduction and References

2.1.1 Introduction. This chapter deals with a technical, albeit essential tool for this thesis: the spectral family of a certain perturbation of the Laplacian.

Spectral families are an important instrument in the study of many differential equations. From the most prominent example one derives a second nomenclature: spectral families are sometimes called *generalised Fourier transforms*. From this wording we derive our notation: we denote by \mathcal{F}_A the transform which diagonalises the self-adjoint operator A.

This diagonalisation of A — specified below — will allow us to give explicit solution formulae in later chapters. Contained in this solution formulae will be the guided modes, which therefore occur naturally as part of the solution. Note that we already stumbled upon those modes in the introductory Section 1.3.

The outline of this chapter is as follows: after a very brief reminder on the *spectral* theorem, we will define the operator A and give its spectral family in Section 2.2, where the main result, Theorem 2.2.12, will be given. The statement of the result will take some room, since a number of definitions and notations have to be clarified first. In particular, certain generalised eigenfunctions have to be defined and analysed. We will illustrate our main theorem by deriving the conventional 1D Fourier transform from it.

Afterwards, we will consider the so-called Pekeris profile in Section 2.3. For this case we will give explicit formulae for the generalised eigenfunctions and show how to determine the ordinary eigenfunctions, which will be heavily employed in the remainder of the thesis, in particular in the numerical examples.

The last Section 2.4 will be devoted to the proof of Theorem 2.2.12. Since we have chosen a rather elementary proof, this will take some pages: much of it will consist of technical lemmata, which can be skipped at first reading. We refer to the remarks at the beginning of Section 2.4.

2.1.2 References and historical remarks. The representation of the spectral family with the help of generalised eigenfunctions has received vast amounts of attention due to its outstanding importance for the field of mathematical physics. It is an underlying, basic notion in classical quantum mechanics, and of critical importance for many different partial differential equations.

Particularly well studied is the theory of Sturm–Liouville problems. It is concerned with differential operators on sub-intervals of \mathbb{R} and has received much attention during the last century, with one of the first important works being Weyl's paper [77]. Those eigenfunction expansions have been further developed and studied for example by Titchmarch [73] and Kodaira [52]. Also worth mentioning is the book of Coddington [22] as a standard reference on this type of expansion. A critical point of interest is the choice of the generalised eigenfunctions: a second order ordinary differential equation of the form

$$u''(x) + q(x)u(x) = 0$$
 for $x \in \mathbb{R}$

always has two linearly independent solutions, which serve as a kind of basis functions in the associated spectral family. Depending on the choice of these eigenfunctions, the obtained formulae can be rather complicated involving (matrix–valued) weights, which leads to the question: how can one choose suitable solutions to obtain simple expansions?

The answer turns out to lie in scattering theory: if we choose the two solutions as scattered waves corresponding to particular incident fields, the weights involved in the spectral family become very simple. This definition of the generalised eigenfunctions is well known, in particular for (in some sense compact) perturbations of the Laplacian on \mathbb{R}^n : the first work in this direction probably coming from Ikebe [45], which was refined later on, where we note in particular the paper by Agmon [1].

The differential operator considered in this chapter has been thoroughly studied in the context of layered media scattering. In fact, a much more general version is commonly considered there. We will restrict ourselves to mention only a few works, which served as a basis for this chapter. We have to start with Wilcox' monograph [78], from where our representation can be deduced, his generalised eigenfunctions being similarly defined as in our case. Secondly, we want to mention the paper by Magnanini and Santosa [58], where the classical Titchmarch theory of eigenfunction expansions has been applied under the additional assumption that the potential q is axially symmetric. Their work uses a more classical choice of generalised eigenfunctions (as for example in [73, 22]), which gives a more complicated representation involving weights.

2.2 The Operator A and Spectral Family \mathcal{F}_A

We will start by recalling the spectral theorem.

2.2.1 Theorem. Let H be a Hilbert space, and let $A : H \supset D(A) \to H$ be a selfadjoint operator. Then there exists a measure space $(\Lambda, d\mu)$, an unitary operator $\mathcal{F}_A : H \to L^2(\Lambda, d\mu)$ and a real valued function $\hat{\lambda} : \Lambda \to \mathbb{R}$ such that

$$Au = \mathcal{F}_A^{-1}(\lambda \mathcal{F}_A u), \quad \text{for all } u \in D(A),$$

where $\hat{\lambda} \mathcal{F}_A u$ denotes the (point-wise) product of the two functions $\hat{\lambda}$, $\mathcal{F}_A u : \Lambda \to \mathbb{C}$, that is $(\hat{\lambda} \mathcal{F}_A u)(m) = \hat{\lambda}(m) \mathcal{F}_A u(m)$ for all $m \in \Lambda$. Furthermore, it holds that

$$u \in D(A)$$
 if and only if $\lambda \mathcal{F}_A u \in L^2(\Lambda, d\mu)$.

In this form, Theorem 2.2.12 can be found in [67, Chapter VIII.3]. The goal of this chapter is to give and prove an explicit representation of the *generalised Fourier transform* for a particular operator A. Let us define this operator.

2.2.2 Definition and Assumption. Let $A : L^2(\mathbb{R}) \supset H^2(\mathbb{R}) \to L^2(\mathbb{R})$ be defined for $f \in C_0^{\infty}(\mathbb{R})$ by

$$Af(x) := -\Delta f(x) + q(x)f(x)$$
 for $x \in \mathbb{R}$

with the function $q : \mathbb{R} \to \mathbb{R}$ given by

$$q(x) = \begin{cases} q_{-} & \text{if } x < a, \\ q_{0}(x) & \text{if } a < x < b, \\ q_{+} & \text{if } b < x, \end{cases}$$

where $a, b \in \mathbb{R}$, a < b, $q_0 : (a, b) \to \mathbb{R}$ is an $L^{\infty}(a, b)$ function and $q_-, q_+ \in \mathbb{R}$ are constants. We also define $q_m := \operatorname{essinf}(q_0)$ and assume $q_m \leq q_- \leq q_+$.



Figure 2.1: Example profile for q.

Note that the assumption $q_{-} \leq q_{+}$ can actually be dropped: by mirroring the *x*-coordinate, q_{-} and q_{+} can be exchanged. We make it nonetheless to simplify a few notations later on. Let us quickly state the spectral properties of A, and let us recall that the spectrum of A is defined by

 $\sigma(A) = \mathbb{C} \setminus \{\lambda \in \mathbb{C} : \text{The operator } \lambda I - A \text{ is boundedly invertible} \}.$

The point spectrum is defined by

$$\sigma_p(A) := \{ \lambda \in \mathbb{C} : \text{There exists } u \in H^2(\mathbb{R}) \text{ such that } Au = \lambda u \},\$$

while the continuous spectrum is given by

$$\sigma_c(A) := \{ \lambda \in \mathbb{C} : (\lambda I - A) : H^2(\mathbb{R}) \to L^2(\mathbb{R}) \text{ is not surjective,} \\ \text{but } \mathcal{R}(\lambda I - A) \text{ is dense in } L^2(\mathbb{R}) \}.$$

The residual spectrum will be empty for our case, so we have no need to define it. Also note that we are dealing with unbounded self-adjoint operators here.

2.2.3 Theorem. A is self-adjoint. We have $\sigma(A) = \sigma_p(A) \cup \sigma_c(A)$, where

(a) The point spectrum is finite (and possibly empty) $\sigma_p(A) = \{\lambda_1, \ldots, \lambda_N\} \subset (q_m, q_-)$, where $N \in \mathbb{N} \cup \{0\}$. For any $\lambda_n \in \sigma_p(A)$, the corresponding eigenspace has dimension one.

(b)
$$\sigma_c(A) = [q_-, +\infty).$$

Proof. The self-adjointness follows by the Rellich–Kato theorem [68, Theorem X.12] since $-\Delta : L^2(\mathbb{R}) \supset H^2(\mathbb{R}) \rightarrow L^2(\mathbb{R})$ is selfadjoint and since the operator $B : L^2(\mathbb{R}) \rightarrow L^2(\mathbb{R})$, $u \mapsto qu$, is bounded in $L^2(\mathbb{R})$ (*B* is $-\Delta$ -bounded with relative bound 0). (a) can be found in Lemma 2.4.4, while (b) is given by Lemma 2.4.6 below.

We continue by defining the measure space $(\Lambda, d\mu)$ from Theorem 2.2.1 for our operator A. For this, we will need the notion of a disjoint union of sets.

2.2.4 Definition. Let A_1, \ldots, A_n be arbitrary sets. Then the *disjoint union* of A_1, \ldots, A_n is defined by

$$A_1 \cup \ldots \cup A_n := \{(k, x) : k \in \{1, \ldots, n\}, x \in A_k\}.$$

Let us quickly illustrate this notion for readers not familiar with it. For example, the disjoint union $(0, \infty) \dot{\cup} (0, \infty)$ can be considered as a two-fold copy of $(0, \infty)$: if we take some $m \in (0, \infty) \dot{\cup} (0, \infty)$, then it can be considered as a positive number, coming either from the first copy of $(0, \infty)$, i.e. m = (1, x), or the second, that is m = (2, x), where $x \in (0, \infty)$. To give a bit more illustration, consider some function $f \in (L^2(0, \infty))^2 \cong L^2(0, \infty) \times L^2(0, \infty)$. It can be considered as a vector valued function $f : (0, \infty) \to \mathbb{C}^2$ by denoting $f(x) = (f_1(x), f_2(x))^{\top}$. However, it can also be canonically identified with a function $\hat{f} : (0, \infty) \dot{\cup} (0, \infty) \to \mathbb{C}$. If $(k, x) \in (0, \infty) \dot{\cup} (0, \infty)$ is some element of the disjoint union (that is, $k \in \{1, 2\}, x \in (0, \infty)$), we can set $\hat{f}(k, x) := f_k(x)$, to obtain a canonical identification between vector valued functions on $(0, \infty)$ and scalar valued functions on $(0, \infty) \dot{\cup} (0, \infty)$.

Similarly, one can consider $f \in L^2(A_1) \times \ldots \times L^2(A_n)$ as a scalar valued function on the disjoint union: if $f = (f_k)_{k=1}^n$, where $f_k \in L^2(A_k)$, then we define $\hat{f} : A_1 \cup \ldots \cup A_n \to \mathbb{C}$ by $\hat{f}(k, x) := f_k(x)$, where our notation stems from.

2.2.5 Definition. We define

$$\Lambda := \sigma_p(A) \,\dot\cup \, (q_+, \infty) \,\dot\cup \, (q_-, \infty).$$

For a function $f : \Lambda \to \mathbb{C}$ defined on Λ , we denote by $f^p : \sigma_p(A) \to \mathbb{C}$ the part defined on $\sigma_p(A) \subset \Lambda$, by $f^+ : (q_+, \infty) \to \mathbb{C}$ the part defined on $(q_+, \infty) \subset \Lambda$ and by $f^- : (q_-, \infty)$ the part defined on $(q_-, \infty) \subset \Lambda$. For abbreviation we write $f = (f^p, f^+, f^-)$. Furthermore, we define the measure $d\mu$ by

$$\int_{\Lambda} f(m) \,\mathrm{d}\mu(m) := \sum_{n=1}^{N} f^p(\lambda_n) + \frac{1}{2\pi} \sum_{\sigma \in \{\pm\}} \int_{q_{\sigma}}^{\infty} f^{\sigma}(\lambda) \frac{\mathrm{d}\lambda}{2\sqrt{\lambda - q_{\sigma}}},$$

where $f : \Lambda \to \mathbb{C}$ is an arbitrary (Lebesgue-)measurable function. Lastly we define the spectral identity $\hat{\lambda} : \Lambda \to \mathbb{R}$ by $\hat{\lambda} = (\hat{\lambda}^p, \hat{\lambda}^+, \hat{\lambda}^-)$, where $\hat{\lambda}^p(\lambda_n) = \lambda_n$, $\hat{\lambda}^+(\lambda) = \lambda$ and $\hat{\lambda}^-(\lambda) = \lambda$.

How can we now understand Λ ? It is a "version" of the spectrum, which takes into account the "multiplicity" of its elements. Each eigenvalue λ_n only appears once, since its multiplicity is 1 by Theorem 2.2.3. If $\lambda \in \sigma_c(A)$ is some value in the continuous spectrum, there are two possible cases: if $\lambda \in (q_-, q_+]$, then there exists only one point in Λ associated to λ : that is, there exists one linearly independent "generalised eigenfunction" associated to λ . If $\lambda \in (q_+, \infty)$, there exist two, that is, λ has in a sense of "multiplicity" 2. Note that we do not give a precise definition of "generalised eigenfunction" here, and emphasise that the last remark is more of a motivational explanation.

We can now define a few function spaces on Λ .

Figure 2.2: Sketch of the Λ . It consists of three parts: the point spectrum $\sigma_p(A) = \{\lambda_1, \ldots, \lambda_N\}$ (the dots in the figure) and the two intervalls $(q_-, +\infty)$ and (q_+, ∞) (the two thick lines).

2.2.6 Definition. Set

$$C_0^{\infty}(\Lambda) := \left\{ f : \Lambda \to \mathbb{C} : f^+ \in C_0^{\infty}(q_+, \infty), \ f^- \in C_0^{\infty}(q_-, \infty) \right\}.$$

For $f, g \in C_0^{\infty}(\Lambda)$ we define the scalar product

$$\langle f,g \rangle_{\Lambda} := \int_{\Lambda} f(m)\overline{g}(m) \,\mathrm{d}\mu(m),$$

with the corresponding norm

$$||f||^2_{L^2(\Lambda,\mathrm{d}\mu)} := \langle f, f \rangle_{\Lambda},$$

and the corresponding space

 $L^{2}(\Lambda, \mathrm{d}\mu) := \{ f : \Lambda \to \mathbb{C} \text{ measurable} : \|f\|_{L^{2}(\Lambda, \mathrm{d}\mu)} < \infty \} = \overline{C_{0}^{\infty}(\Lambda)}.$

With these definitions, we have the spectral space $L^2(\Lambda, d\mu)$ available. The generalised Fourier transform \mathcal{F}_A will appear as an integral operator with a particular kernel Ψ : $\Lambda \times \mathbb{R} \to \mathbb{C}$. It will be defined with the help of solutions to scattering problems on the real line, for which we introduce a few additional preliminaries in the following lemmata.

2.2.7 Lemma. Let $\lambda > q_{-}$, $\lambda \neq q_{+}$, and consider a solution $u \in H^{2}_{loc}(\mathbb{R})$ of

$$\Delta u(x) + (\lambda - q(x))u(x) = 0 \quad \text{for } x \in \mathbb{R}.$$
(2.2.1)

Then u must have the form

$$u(x) = \begin{cases} \alpha_1 e^{i\sqrt{\lambda - q_-}} x + \alpha_2 e^{-i\sqrt{\lambda - q_-}} x & \text{for } x < a, \\ \alpha_3 e^{i\sqrt{\lambda - q_+}} x + \alpha_4 e^{-i\sqrt{\lambda - q_+}} x & \text{for } x > b, \end{cases}$$
(2.2.2)

with some coefficients $\alpha_1, \ldots, \alpha_4 \in \mathbb{C}$. They fulfil

$$\operatorname{Re}(\sqrt{\lambda - q_{-}})|\alpha_{1}|^{2} + \operatorname{Re}(\sqrt{\lambda - q_{+}})|\alpha_{4}|^{2}$$

=
$$\operatorname{Re}(\sqrt{\lambda - q_{-}})|\alpha_{2}|^{2} + \operatorname{Re}(\sqrt{\lambda - q_{+}})|\alpha_{3}|^{2}.$$
 (2.2.3)

This relation will be called the *energy conservation equation*.

Proof. The representation of the solution follows immediately, considering that $\lambda - q(x) = \lambda - q_{-}$ for any x < a (and $\lambda - q(x) = \lambda - q_{+}$ for x > b). If we multiply (2.2.1) by $\overline{u}(x)$, integrate from a to b and use partial integration, we obtain

$$0 = u'(b)\overline{u}(b) - u'(a)\overline{u}(a) + \int_{a}^{b} -|u'(x)|^{2} + (\lambda - q(x))|u(x)|^{2} dx.$$

If we take the imaginary part of the last equation, the integral vanishes. The remainder

$$0 = \operatorname{Im} \left[u'(b)\overline{u}(b) - u'(a)\overline{u}(a) \right]$$

can be easily recomputed with the help of the representations for x < a and x > b to obtain the energy conservation (2.2.3).

2.2.8 Definition. Let $\lambda > q_-$, $\lambda \neq q_+$, and let $u \in H^2_{loc}(\mathbb{R})$ be a solution of (2.2.1) on $\mathbb{R} \setminus (a, b)$. We say that u is *outgoing* if it has the form

$$u(x) = \begin{cases} \alpha_2 e^{-i\sqrt{\lambda - q_-} x} & \text{for } x < a, \\ \alpha_3 e^{i\sqrt{\lambda - q_+} x} & \text{for } x > b, \end{cases}$$

where $\alpha_2, \alpha_3 \in \mathbb{C}$ are some complex coefficients.

2.2.9 Lemma. (Unique continuation principle) Let $u \in H^2_{loc}(\mathbb{R})$ be a solution to (2.2.1), and let there exist a non-empty open interval $I \subset \mathbb{R}$ such that u(x) = 0 for $x \in I$. Then u(x) = 0 for all $x \in \mathbb{R}$.

Proof. Let I = (c, d). Then $u|_{(-\infty,c)}$ fulfils

$$\begin{cases} \Delta u + (\lambda - q)u = 0 & \text{ in } (-\infty, c), \\ u(c) = 0 = u'(c), \end{cases}$$

since $u \in H^2_{loc}(\mathbb{R})$, and since u vanishes to the right of c on (c, d). This problem has a unique solution by the Picard-Lindelöf theorem (for further references regarding L^{∞} coefficients, we refer to Section 2.4), which is obviously given by u = 0 on $(-\infty, c)$. The same way one can confirm that u vanishes also on (d, ∞) .

2.2.10 Lemma. Let $\lambda > q_{-}$, $\lambda \neq q_{+}$ and let $u_{i} \in H^{2}_{loc}(\mathbb{R} \setminus (a, b))$ be given by

$$u_{\mathbf{i}}(x) = \begin{cases} c_1 e^{i\sqrt{\lambda - q_-} x} & \text{for } x < a, \\ c_4 e^{-i\sqrt{\lambda - q_+} x} & \text{for } x > b, \end{cases}$$

where $c_1, c_4 \in \mathbb{C}$ are arbitrary coefficients. Then there exists exactly one solution $u \in H^2_{loc}(\mathbb{R})$ of (2.2.1) such that

$$u - u_i$$
 is outgoing.

Proof. Let us show uniqueness: let u_1 and u_2 be two solutions fulfilling (2.2.1) such that $u_k - u_i$ is outgoing, $k \in \{1, 2\}$. Then $w = u_1 - u_2$ is also outgoing, implying that $\alpha_1 = \alpha_4 = 0$, where $\alpha_1, \ldots, \alpha_4$ are the coefficients of w in the expansion (2.2.2). We have by the energy conservation (2.2.3)

$$0 = \operatorname{Re}(\sqrt{\lambda - q_{-}})|\alpha_{2}|^{2} + \operatorname{Re}(\sqrt{\lambda - q_{+}})|\alpha_{3}|^{2}.$$

Note that both terms are non-negative, that $\operatorname{Re}\left(\sqrt{\lambda-q_{-}}\right) > 0$ and that $\operatorname{Re}\left(\sqrt{\lambda-q_{+}}\right) \geq 0$. Note that $\sqrt{\lambda-q_{+}}$ might be imaginary, but cannot be real and negative. This implies that at least $\alpha_{3} = 0$, which in turn yields that w(x) = 0 for x > b. The unique continuation principle now implies w = 0.

To see the existence of the solution, we now restrict ourselves to the incident field $c_1 = 1$, $c_4 = 0$ and to the case $\lambda \in (q_-, q_+)$. The other cases can be treated analogously. Let us denote by $w \in H^2_{loc}(\mathbb{R})$ the unique solution of (2.2.1) such that

$$w'(b) = i\sqrt{\lambda - q_+}, \quad w(b) = 1,$$

and let again $\alpha_1, \ldots, \alpha_4$ be its coefficients. By the Picard–Lindelöf theorem, w exists and is unique. Clearly, $\alpha_3 = 1$, $\alpha_4 = 0$. We obtain by the energy conservation equation that

$$\operatorname{Re}(\sqrt{\lambda - q_{-}})|\alpha_{1}|^{2} = \operatorname{Re}(\sqrt{\lambda - q_{-}})|\alpha_{2}|^{2},$$

so that $\alpha_1 \neq 0 \neq \alpha_2$, since otherwise the unique continuation principle would imply w = 0. One can now easily verify that $u(x) := \alpha_1^{-1} w(x)$ fulfils the Helmholtz equation, while $u - u_i$ is outgoing.

With these preliminaries, we can now define the kernel of the generalised Fourier transform.

2.2.11 Definition. We define the function $\Psi : \Lambda \times \mathbb{R} \to \mathbb{C}, \Psi = (\Psi^p, \Psi^+, \Psi^-)$ as follows: for $\lambda_n \in \sigma_p(A)$, let $\Psi^p : \sigma_p(A) \times \mathbb{R} \to \mathbb{C}$ be defined by

$$\Psi^p(\lambda_n, x) = \frac{\phi_n(x)}{\|\phi_n\|_{L^2(\mathbb{R})}},$$

where ϕ_n is a nontrivial eigenfunction for the eigenvalue λ_n (recall that λ_n has multiplicity 1 by Theorem 2.2.3).

Let $\sigma \in \{\pm\}$. Then we define $\Psi^{\sigma} : (q_{\sigma}, \infty) \times \mathbb{R} \to \mathbb{C}$ as follows: for $\lambda > q_{\sigma}$, let $\Psi^{\sigma}(\lambda, \cdot)$ be the unique solution to

$$\begin{cases} \Delta_x \Psi^{\sigma}(\lambda, x) + (\lambda - q(x))\Psi^{\sigma}(\lambda, x) = 0 & \text{for } x \in \mathbb{R}, \\ \Psi^{\sigma}(\lambda, \cdot) - u_i^{\sigma} & \text{is outgoing,} \end{cases}$$

where the incident field is given by either

$$u_i^-(x) = \begin{cases} e^{i\sqrt{\lambda - q_-}x} & \text{if } x < a, \\ 0 & \text{if } b < x, \end{cases}$$

or

$$u_i^+(x) = \begin{cases} 0 & \text{if } x < a, \\ e^{-i\sqrt{\lambda - q_+}x} & \text{if } b < x, \end{cases}$$

depending on σ . For any $\lambda_n \in \sigma_p(A)$, we call $\Psi^p(\lambda_n, \cdot)$ a normalised eigenfunction, while for $\lambda \in \sigma_c(A)$, we call $\Psi^+(\lambda, \cdot)$ and $\Psi^-(\lambda, \cdot)$ generalised eigenfunctions.

2.2.12 Theorem. For $\Psi : \Lambda \times \mathbb{R} \to \mathbb{C}$ as defined in Definition 2.2.11, we define $\mathcal{F}_A : L^2(\mathbb{R}) \to L^2(\Lambda, d\mu)$ for $f \in C_0^{\infty}(\mathbb{R})$ and $m \in \Lambda$ by

$$\mathcal{F}_A f(m) = \int_{\mathbb{R}} f(x) \overline{\Psi}(m, x) \, \mathrm{d}x.$$

It is unitary, i.e. we have the Parseval's relation for $f, g \in L^2(\mathbb{R})$,

$$\langle f, g \rangle_{\mathbb{R}} = \langle \mathcal{F}_A f, \mathcal{F}_A g \rangle_{\Lambda}$$
 (2.2.4)

and its inverse $\mathcal{F}_A^{-1}: L^2(\Lambda, \mathrm{d}\mu) \to L^2(\mathbb{R})$ is defined for $\tilde{f} \in C_0^\infty(\Lambda)$ by

$$\mathcal{F}_A^{-1}\tilde{f}(x) = \int_{\Lambda} \Psi(m, x)\tilde{f}(m) \,\mathrm{d}\mu(m).$$

It diagonalizes A, i.e.

$$Af = \mathcal{F}_A^{-1}\hat{\lambda}\mathcal{F}_A f$$

for any $f \in H^2(\mathbb{R})$. Here $\hat{\lambda} \mathcal{F}_A f$ denotes the point-wise product of $\hat{\lambda}$ with $\mathcal{F}_A f$. **Proof.** The proof is (almost) contained in Section 2.4 below. We point out that the different parts of the Theorem are proved at the end of this chapter in Lemmata 2.4.12 and 2.4.13, and point also to the remarks of Subsection 2.4.14.

2.2.13 Example. To illustrate the expansion theorem, let us use it to derive the Fourier inversion formula: when q(x) = 0 for all $x \in \mathbb{R}$, we obtain $Af = -\Delta f$, so we will end up diagonalizing the Laplacian. We set a = b = 0 and $q_- = q_+ = q_m = 0$, so in this case, Theorem 2.2.3 yields $\sigma_p(A) = \emptyset$, and $\sigma_c(A) = [0, \infty)$. To determine $\Psi^{\pm}(\lambda, \cdot)$ we have to find for $\lambda > 0$ the solution of $\Delta_x \Psi^+(\lambda, \cdot) + \lambda \Psi^+(\lambda, \cdot) = 0$ such that

$$\Psi^{+}(\lambda, x) = \begin{cases} \alpha e^{i\sqrt{\lambda}x} & \text{if } x < 0, \\ e^{i\sqrt{\lambda}x} + \beta e^{-i\sqrt{\lambda}x} & \text{if } x > 0. \end{cases}$$

Clearly, this implies $\Psi^+(\lambda, x) = e^{i\sqrt{\lambda}x}$ for all $x \in \mathbb{R}$. Likewise we obtain $\Psi^-(\lambda, x) = e^{-i\sqrt{\lambda}x}$. From Theorem 2.2.12 we obtain for any $f \in L^2(\mathbb{R})$

$$f(x) = (\mathcal{F}_A^{-1}\mathcal{F}_A f)(x) = \int_{\Lambda} \Psi(m, x) \int_{\mathbb{R}} \overline{\Psi}(m, y) f(y) \, \mathrm{d}y \, \mathrm{d}\mu(m).$$

Applying the definition of the spectral space $(\Lambda, d\mu)$, we can write the right hand side integral by

$$\begin{split} f(x) &= \frac{1}{2\pi} \int_{0}^{\infty} \Psi^{+}(\lambda, x) \int_{\mathbb{R}} \overline{\Psi}^{+}(\lambda, y) f(y) \, \mathrm{d}y \frac{\mathrm{d}\lambda}{2\sqrt{\lambda}} \\ &+ \frac{1}{2\pi} \int_{0}^{\infty} \Psi^{-}(\lambda, x) \int_{\mathbb{R}} \overline{\Psi}^{-}(\lambda, y) f(y) \, \mathrm{d}y \frac{\mathrm{d}\lambda}{2\sqrt{\lambda}} \\ &= \frac{1}{2\pi} \int_{0}^{\infty} e^{i\sqrt{\lambda}x} \int_{\mathbb{R}} e^{-i\sqrt{\lambda}x} f(y) \, \mathrm{d}y \frac{\mathrm{d}\lambda}{2\sqrt{\lambda}} \\ &+ \frac{1}{2\pi} \int_{0}^{\infty} e^{-i\sqrt{\lambda}x} \int_{\mathbb{R}} e^{i\sqrt{\lambda}x} f(y) \, \mathrm{d}y \frac{\mathrm{d}\lambda}{2\sqrt{\lambda}} \end{split}$$

We now substitute $\xi = \sqrt{\lambda}$ in the first and $\xi = -\sqrt{\lambda}$ in the second integral to obtain

$$f(x) = \frac{1}{2\pi} \int_{0}^{\infty} e^{i\xi x} \int_{\mathbb{R}} e^{-i\xi x} f(y) \,\mathrm{d}y \,\mathrm{d}\xi + \frac{1}{2\pi} \int_{-\infty}^{0} e^{i\xi x} \int_{\mathbb{R}} e^{-i\xi x} f(y) \,\mathrm{d}y \,\mathrm{d}\xi,$$

which, not so surprisingly, turns out to be the Fourier inversion formula.

2.3 The Pekeris Profile

In this section we consider a special profile q, which admits a (mostly) explicit representation of the eigenfunctions and generalised eigenfunctions. Its name originates from Pekeris' paper [65], where it has been studied for the first time. Due to its explicit form, it will be the foundation of the numerical examples we will show in later chapters, where we will need to implement \mathcal{F}_A . Let a < b and consider for (arbitrary) constants $q_m < q_- \leq q_+$ the profile

$$q(x) = \begin{cases} q_{-} & \text{for } x < a, \\ q_{m} & \text{for } a < x < b, \\ q_{+} & \text{for } b < x. \end{cases}$$

We only demand $q_m < q_-$ to obtain a point spectrum by Theorem 2.2.3, which will be shown to be non-empty under the given assumptions. If $q_m \ge q_-$, the calculations below can be carried out as shown, too. However, in this case the point spectrum $\sigma_p(A)$ is empty, and the corresponding part can be skipped. To determine the spectral family of $A = -\Delta + q$, we fix some $\lambda \in \mathbb{R} \setminus \{q_+, q_-, q_m\}$ and study solutions $u \in H^2_{\text{loc}}(\mathbb{R})$ of the equation $(-\Delta + q - \lambda)u = 0$; due to the piecewise constant coefficients they have the form

$$u(x) = \begin{cases} \alpha_1 e^{i\sqrt{\lambda - q_-}(x-a)} + \alpha_2 e^{-i\sqrt{\lambda - q_-}(x-a)} & \text{if } x < a, \\ \alpha_3 e^{i\sqrt{\lambda - q_m}x} + \alpha_4 e^{-i\sqrt{\lambda - q_m}x} & \text{if } a < x < b, \\ \alpha_5 e^{i\sqrt{\lambda - q_+}(x-b)} + \alpha_6 e^{-i\sqrt{\lambda - q_+}(x-b)} & \text{if } b < x, \end{cases}$$
(2.3.1)

where $\alpha_1, \ldots, \alpha_6 \in \mathbb{C}$ are some complex coefficients. To proceed, we need a few abbreviations, and define the following quantities for $\sigma \in \{-, +, m\}$

$$\mu_{\sigma}(\lambda) := \sqrt{\lambda - q_{\sigma}}, \ p_a(\lambda) := e^{i\sqrt{\lambda - q_m}a}, \ p_b(\lambda) := e^{i\sqrt{\lambda - q_m}b}.$$

To shorten notation, we omit the argument λ in the following for the five functions defined in the last line. Since we demand that $u \in H^2_{loc}(\mathbb{R})$, u and u' are continuous (by Sobolev-embedding). Setting x = a and x = b in the representations of u, we obtain that the coefficients $\alpha_1, \ldots, \alpha_6$ must fulfil the following set of equations:

$$\alpha_{1} + \alpha_{2} = p_{a}\alpha_{3} + p_{a}^{-1}\alpha_{4},$$

$$\alpha_{5} + \alpha_{6} = p_{b}\alpha_{3} + p_{b}^{-1}\alpha_{4},$$

$$\mu_{-}\alpha_{1} - \mu_{-}\alpha_{2} = \mu_{m}p_{a}\alpha_{3} + \mu_{m}p_{a}^{-1}\alpha_{4},$$

$$\mu_{+}\alpha_{5} - \mu_{+}\alpha_{6} = \mu_{m}p_{b}\alpha_{3} + \mu_{m}p_{b}^{-1}\alpha_{4}.$$

(2.3.2)

Solving the first and second equation for α_2 and α_5 , and inserting the result into the third and fourth equations, we obtain the following system of equations for $\alpha_3, \alpha_4, \alpha_1, \alpha_6$, which we write in matrix-vector form

$$\begin{pmatrix} (\mu_{-} + \mu_{m})p_{a} & (\mu_{-} - \mu_{m})p_{a}^{-1} \\ (\mu_{+} - \mu_{m})p_{b} & (\mu_{+} + \mu_{m})p_{b}^{-1} \end{pmatrix} \begin{pmatrix} \alpha_{3} \\ \alpha_{4} \end{pmatrix} = \begin{pmatrix} 2\mu_{-}\alpha_{1} \\ 2\mu_{+}\alpha_{6} \end{pmatrix}.$$
 (2.3.3)

Let us denote the left hand side matrix by $B = B(\lambda)$, and let

$$d(\lambda) = \det(B(\lambda))$$

= $(\mu_{-} + \mu_{m})(\mu_{+} + \mu_{m})p_{a}p_{b}^{-1} - (\mu_{-} - \mu_{m})(\mu_{+} - \mu_{m})p_{b}p_{a}^{-1}$ (2.3.4)

denote its determinant. We have to determine (a) the point spectrum, and the corresponding normalised eigenfunctions and (b) the generalised eigenfunctions (see Definition 2.2.11). Let us do both.

2.3.1 Eigenfunctions. We have to search for eigenvalue/eigenvector pairs $\lambda \in \mathbb{R}$, $u \in H^2(\mathbb{R})$. By Theorem 2.2.3, we only have to consider $\lambda \in (q_m, q_-)$. For such λ the terms corresponding to α_1 and α_6 in (2.3.1) increase exponentially, and hence $\alpha_1 = \alpha_6 = 0$ must vanish for u to be in $H^2(\mathbb{R})$. This implies by (2.3.3) that

$$B(\lambda) \begin{pmatrix} \alpha_3\\ \alpha_4 \end{pmatrix} = 0,$$

and this may happen only if $d(\lambda) = \det(B(\lambda)) = 0$. One easily sees that for any $\lambda \in (q_m, q_-)$ such that $d(\lambda) = 0$, any element of the Kernel $(\hat{\alpha}_3, \hat{\alpha}_4) \in \mathcal{N}(B(\lambda))$ generates an $H^2(\mathbb{R})$ eigenfunction, so determining all zeros of d is equivalent to determining the point spectrum (for more details, we refer to the proof of Lemma 2.4.4). Let us rewrite $d(\lambda) = 0$ in a fashion that allows numerical computation of *all* zeros: by easy manipulation of equation (2.3.4), we obtain that $d(\lambda) = 0$ is equivalent to

$$\frac{\mu_m + \mu_-}{\mu_m - \mu_-} \frac{\mu_m + \mu_+}{\mu_m - \mu_+} \frac{p_a^2}{p_b^2} = 1.$$
(2.3.5)

Note that for $\lambda \in (q_m, q_-)$ we have $\mu_m = \sqrt{\lambda - q_m} > 0$, while $\mu_\sigma = \sqrt{\lambda - q_\sigma} = i\sqrt{q_\sigma - \lambda} \in i(0, \infty)$, so that $\overline{\mu_m + \mu_\sigma} = \mu_m - \mu_\sigma$. Since $|p_a| = |p_b| = 1$, the left hand side of (2.3.4) is always a complex number of modulus one, and we only have to check whether the arguments of the two sides of the equality are equal. Elementary calculation now yields

$$\theta(\lambda) := \arg\left(\frac{\mu_m + \mu_-}{\mu_m - \mu_-} \frac{\mu_m + \mu_+}{\mu_m - \mu_+} \frac{p_a^2}{p_b^2}\right)$$
$$= 2 \operatorname{atan}\left(\sqrt{\frac{q_+ - \lambda}{\lambda - q_m}}\right) + 2 \operatorname{atan}\left(\sqrt{\frac{q_- - \lambda}{\lambda - q_m}}\right) - 2(b - a)\sqrt{\lambda - q_m}$$

Hence (2.3.5) is fulfilled if and only if

$$\theta(\lambda) = 2\pi n \tag{2.3.6}$$

for some $n \in \mathbb{Z}$. Note that θ is bounded on $[q_m, q_-]$ and continuous, and one easily sees that it is also strictly decreasing. Hence, for any

$$n \in (\theta(q_{-})(2\pi)^{-1}, \ \theta(q_{m})(2\pi)^{-1}) \cap \mathbb{Z}$$

there exists exactly one solution of $\theta(\lambda) = 2\pi n$. Numerically, one can find those by solving (2.3.6) for all suitable $n \in \mathbb{Z}$ by Newton's method. Denote this finite set of solutions (that is, the eigenvalues of A) by $\{\lambda_1, \ldots, \lambda_N\}$, ordered ascendingly. For each $n \in \{1, \ldots, N\}$, the Kernel $\mathcal{N}(B(\lambda_n))$ has dimension 1, since $B(\lambda_n)$ is a 2 × 2 matrix and easily seen to be non-zero. Let $(\alpha_3, \alpha_4)^{\top} \in \mathcal{N}(B(\lambda_n)) \setminus \{0\}$ be some vector. After setting $\alpha_1 = \alpha_6 = 0$, we can retrieve α_2 and α_5 from the first two equations (2.3.2), and arrive at a full coefficient vector $\alpha_1, \ldots, \alpha_6$. (2.3.1) then yields an eigenfunction u.

2.3. THE PEKERIS PROFILE

To obtain the normalised eigenfunction $\Psi^p(\lambda_n, \cdot)$ corresponding to λ_n , one calculates straightforwardly

$$\begin{aligned} \|u\|_{L^{2}(\mathbb{R})}^{2} = &(b-a)\left(|\alpha_{3}|^{2} + |\alpha_{4}|^{2}\right) + \frac{1}{\mu_{m}}\operatorname{Im}\left(\alpha_{3}\overline{\alpha}_{4}(p_{b}^{2} - p_{a}^{2})\right) \\ &+ \frac{1}{2|\mu_{-}|}\left|\alpha_{3}p_{a} + \alpha_{4}p_{a}^{-1}\right|^{2} + \frac{1}{2|\mu_{+}|}\left|\alpha_{3}p_{b} + \alpha p_{b}^{-1}\right|^{2}.\end{aligned}$$

Hence, the corresponding real eigenfunction is given by

$$\Psi^{p}(\lambda_{n}, x) = \sqrt{\frac{\overline{\alpha}_{4}|\alpha_{3}|}{\alpha_{3}|\alpha_{4}|}} \|u\|_{L^{2}(\mathbb{R})}^{-1} u(x) \quad \text{for } x \in \mathbb{R},$$

where the factor under the square root ensures that $\Psi^p(\lambda_n, \cdot)$ is real.



Figure 2.3: Sketch of all proper eigenfunctions (in total 3) of A, for a given Pekeris– profile, also sketched (a = -5, b = 5, $q_{-} = 0$, $q_{m} = -2$, $q_{+} = -1$). The functions are sinusoidal for $x \in (-5, 5)$, and exponential elsewhere.

2.3.2 Generalised eigenfunctions. One easily checks that for $\lambda > q_{-}$ the determinant $d(\lambda)$ cannot vanish, since the absolute value of the numerator in (2.3.5) is strictly larger than the denominator. Accordingly, $B = B(\lambda)$ is invertible, and for fixed known α_1, α_6 , we can retrieve from (2.3.3) the coefficients α_3, α_4 by

$$\begin{pmatrix} \alpha_3 \\ \alpha_4 \end{pmatrix} = B^{-1} \begin{pmatrix} 2\mu_- \alpha_1 \\ 2\mu_+ \alpha_6 \end{pmatrix}$$

$$= \frac{1}{d(\lambda)} \begin{pmatrix} 2\mu_- (\mu_+ + \mu_m) p_b^{-1} & 2\mu_+ (\mu_m - \mu_-) p_a^{-1} \\ 2\mu_- (\mu_m - \mu_+) p_b & 2\mu_+ (\mu_- + \mu_m) p_a \end{pmatrix} \begin{pmatrix} \alpha_1 \\ \alpha_6 \end{pmatrix},$$

$$(2.3.7)$$

while we can retrieve α_2, α_5 from the remaining coefficients from the first two equations of (2.3.3), namely by

$$\begin{aligned}
\alpha_2 &= p_a \alpha_3 + p_a^{-1} \alpha_4 - \alpha_1, \\
\alpha_5 &= p_b \alpha_3 + p_b^{-1} \alpha_4 - \alpha_6.
\end{aligned}$$
(2.3.8)

Let $\lambda > q_{-}$, and consider the generalised eigenfunction $\Psi^{-}(\lambda, \cdot)$ from Definition 2.2.11. It solves $(-\Delta - \lambda + q)\Psi^{-}(\lambda, \cdot) = 0$ and it must have the form

$$\Psi^{-}(\lambda, x) = \begin{cases} c_{-}e^{-i\sqrt{\lambda - q_{-}} x} + e^{i\sqrt{\lambda - q_{-}} x}, & \text{if } x < a, \\ c_{+}e^{i\sqrt{\lambda - q_{+}} x} & \text{if } x > b. \end{cases}$$

By comparing coefficients, we obtain that $\Psi^{-}(\lambda, \cdot)$ corresponds to u with $\alpha_{1} = e^{i\sqrt{\lambda-q_{-}}a}$ and $\alpha_{6} = 0$. The remaining coefficients $\alpha_{2}, \ldots, \alpha_{5}$ then follow from (2.3.7) and (2.3.8). Similarly, one easily sees that for $\lambda > q_{+}$, the second eigenfunction $\Psi^{+}(\lambda, \cdot)$ is obtained in the same fashion by comparing coefficients and finding that $\alpha_{1} = 0$ and $\alpha_{6} = e^{i\sqrt{\lambda-q_{+}}b}$.

2.3.3 Definition. Let us introduce the function

$$\alpha: \Lambda \to \mathbb{C}^6.$$

For a given $m \in \Lambda$, it gives the coefficients $\alpha_1, \ldots, \alpha_6 \in \mathbb{C}$ of the corresponding eigenfunction $\Psi(m, \cdot)$, expanded as in (2.3.1).

Note that all formulae we calculated are explicit in terms of elementary function, provided, one has determined the point spectrum. This will allow us to implement the transformation of the operator \mathcal{F}_A .

2.4 Proof of the Expansion Theorem

2.4.1 Overview. Our goal is to prove Theorem 2.2.12. This, however, will take some space, and we want to start by giving an overview over the proof and by introducing some elementary notation from spectral theory (for an introduction into those topics, we will generally refer to [67, Chapter VII and VIII]). Let us denote the resolvent of the operator A for $\lambda \in \mathbb{C} \setminus \mathbb{R}$ by

$$R_{\lambda} := (A - \lambda)^{-1}$$

Since A is self-adjoint by Theorem 2.2.3, the resolvent $R_{\lambda} : L^2(\mathbb{R}) \to H^2(\mathbb{R})$ is bounded for any non-real λ . For some bounded interval $I \subset \mathbb{R}$, denote by $P_I : L^2(\mathbb{R}) \to L^2(\mathbb{R})$ the spectral projection associated with A. The core of our proof now rests in two steps: firstly, we can compute $P_{(a,b)}$ for a < b with the help of Stone's formula (see for example [67, Theorem VII.13 and Remark after Theorem VIII.6])

$$\frac{1}{2}(P_{[a,b]}f + P_{(a,b)}f) = \frac{1}{2\pi i} \lim_{\epsilon \to 0} \int_{a}^{b} R_{\lambda+i\epsilon}f - R_{\lambda-i\epsilon}f \,\mathrm{d}\lambda.$$
(2.4.1)

Once a representation for the spectral projection is found, we will use that for any $f \in L^2(\mathbb{R})$ we have that (see [67, Remark after Theorem VIII.5])

$$f = \lim_{\rho \to \infty} P_{(-\rho,\rho)} f.$$

This allows us to obtain a rather complicated representation of the identity. It will turn out, however, that this representation is nothing but the inversion formula $f = \mathcal{F}_A^{-1} \mathcal{F}_A f$ from Theorem 2.2.12, which will afterwards yield the remaining properties of the theorem. We can now structure the remainder of this section as follows.

2.4. PROOF OF THE EXPANSION THEOREM

an

(1) We need to study the resolvent R_{λ} very precisely. To do this, we will derive a representation of the resolvent with a Green's function, that is, we will find $G_{\lambda} : \mathbb{R} \times \mathbb{R} \to \mathbb{C}$ such that

$$R_{\lambda}f(x) = \int_{\mathbb{R}} G_{\lambda}(x, y)f(y) \, \mathrm{d}y \qquad \text{for } x \in \mathbb{R}.$$

 G_{λ} will be expressed with the help of two particular solutions v_{λ}^{\pm} of the Helmholtz equation, which are introduced in Subsection 2.4.2 and are studied in Lemmata 2.4.3 and 2.4.4. The Green's function will be given in Lemma 2.4.5.

- (2) To take the limit $\epsilon \to 0$ in Stone's formula (2.4.1), we will prepare a number of auxiliary results: this will be the topic of Lemmata 2.4.6 to 2.4.9. We can then give the core argument of this section, which is contained in the proof of Lemma 2.4.10, where we apply Stone's formula and take the limit $\epsilon \to 0$.
- (3) The remainder of the section will be concerned with proving the claims of Theorem 2.2.12.
- **2.4.2 Two special solutions.** For $\lambda \in \mathbb{C}$, we define v_{λ}^+ and v_{λ}^- as the unique solutions of

$$-\Delta v_{\lambda}^{\pm}(x) - (\lambda - q(x))v_{\lambda}^{\pm}(x) = 0 \quad \text{for } x \in \mathbb{R}$$
(2.4.2)

such that

$$v_{\lambda}^{+}(a) = 1, \ \partial_{x}v_{\lambda}^{+}(a) = -i\sqrt{\lambda} - q_{-},$$

d
$$v_{\lambda}^{-}(b) = 1, \ \partial_{x}v_{\lambda}^{-}(b) = i\sqrt{\lambda} - q_{+}.$$
 (2.4.3)

Those functions are both well-defined by Lemma 2.4.3 below. For $\lambda \in \mathbb{C} \setminus \{q_+\}$ we have that

$$v_{\lambda}^{+}(x) = \begin{cases} e^{-i\sqrt{\lambda - q_{-}}(x-a)} & \text{if } x < a, \\ \beta_{\lambda}^{+}e^{i\sqrt{\lambda - q_{+}}(x-b)} + \alpha_{\lambda}^{+}e^{-i\sqrt{\lambda - q_{+}}(x-b)} & \text{if } x > b, \end{cases}$$
(2.4.4)

while for $\lambda \in \mathbb{C} \setminus \{q_{-}\}$

$$v_{\lambda}^{-}(x) = \begin{cases} \alpha_{\lambda}^{-} e^{i\sqrt{\lambda - q_{-}}(x-a)} + \beta_{\lambda}^{-} e^{-i\sqrt{\lambda - q_{-}}(x-a)} & \text{if } x < a, \\ e^{i\sqrt{\lambda - q_{+}}(x-b)} & \text{if } x > b, \end{cases}$$
(2.4.5)

where $\alpha_{\lambda}^{+}, \alpha_{\lambda}^{-}, \beta_{\lambda}^{+}, \beta_{\lambda}^{-} \in \mathbb{C}$ are some complex coefficients. Let us point at the most important feature of these two functions: if $\lambda \in \mathbb{C} \setminus [q_{-}, \infty), v_{\lambda}^{+}(x)$ decays exponentially as $x \to -\infty$, while $v_{\lambda}^{-}(x)$ decays exponentially as $x \to +\infty$. We furthermore define the Wronskian of the two functions as

$$w_{\lambda} := W(v_{\lambda}^+, v_{\lambda}^-)(x) = v_{\lambda}^+(x)\partial_x v_{\lambda}^-(x) - (\partial_x v_{\lambda}^+)(x)v_{\lambda}^-(x), \qquad (2.4.6)$$

Note that the Wronskian w_{λ} does not depend on $x \in \mathbb{R}$. This can be easily seen by differentiating the right hand side of (2.4.6) with respect to x and utilising that v_{λ}^{+} and v_{λ}^{-} both fulfil (2.4.2). The Wronskian can be easily expressed with the coefficients of (2.4.4) and (2.4.5) as follows

$$w_{\lambda} = -2i\sqrt{\lambda - q_{-}} \ \alpha_{\lambda}^{-} = -2i\sqrt{\lambda - q_{+}} \ \alpha_{\lambda}^{+}, \qquad (2.4.7)$$

whenever the quantities on the right are defined; this follows by applying the definition of the Wronskian (2.4.6) to the representation for x < a (or x > b) after a bit of straightforward calculation.

The existence and uniqueness of the two solutions would follow immediately if we had assumed that the coefficient q is continuous, since then the well-know Picard–Lindelöf theorem applies. We will give the corresponding result for a more general situation in the following.

2.4.3 Definition and Lemma. Let $q \in L^{\infty}(\mathbb{R})$. For $\lambda \in \mathbb{C}$, let $\tilde{\psi}_{\lambda}, \hat{\psi}_{\lambda} : \mathbb{R} \to \mathbb{C}$ be the solutions to

$$-\Delta\psi_{\lambda}(x) + q(x)\psi_{\lambda}(x) = \lambda\psi_{\lambda}(x) \quad \text{for } x \in \mathbb{R}$$

such that

$$\tilde{\psi}_{\lambda}(a) = 1, \ \partial_x \tilde{\psi}_{\lambda}(a) = 0, \quad \text{and} \quad \hat{\psi}_{\lambda}(a) = 0, \ \partial_x \hat{\psi}_{\lambda}(a) = 1.$$

 $\tilde{\psi}_{\lambda}, \hat{\psi}_{\lambda}$ are well defined and for any bounded interval $I \subset \mathbb{R}$, the mappings $\mathbb{C} \to H^2(I), \ \lambda \mapsto \tilde{\psi}_{\lambda}$ and $\mathbb{C} \to H^2(I), \ \lambda \mapsto \hat{\psi}_{\lambda}$ are entire¹. In particular, the matrix

$$T_{\lambda} = \begin{pmatrix} t_{11}(\lambda) & t_{12}(\lambda) \\ t_{21}(\lambda) & t_{22}(\lambda) \end{pmatrix} := \begin{pmatrix} \tilde{\psi}_{\lambda}(b) & \hat{\psi}_{\lambda}(b) \\ \partial_x \tilde{\psi}_{\lambda}(b) & \partial_x \hat{\psi}_{\lambda}(b) \end{pmatrix}$$

is an entire function of λ , that is, all entries t_{nm} are entire functions $(n, m \in \{1, 2\})$. For any $\lambda \in \mathbb{C}$ we have that $\det(T_{\lambda}) \neq 0$.

Furthermore, the mappings $\mathbb{C} \setminus [q_-, +\infty) \to H^2(I)$, $\lambda \mapsto v_{\lambda}^{\pm}$, are holomorphic on their domain and can be continuously extended such that for any $\lambda > 0$ the limits

$$\lim_{\substack{\epsilon \to 0\\\epsilon > 0}} v_{\lambda+i\epsilon}^{\pm} = v_{\lambda}^{\pm},$$

exist, where the convergence takes place in $H^2(I)$.

Proof. For the fact that $\lambda \mapsto \tilde{\psi}_{\lambda}$, $\lambda \mapsto \hat{\psi}_{\lambda}$ are entire functions and well-defined, we refer to [50, Section 4.2], where those statements are proved for $q \in L^2(I)$, which is clearly the case if $q \in L^{\infty}(\mathbb{R})$. To see that the determinant cannot vanish, let us note the meaning of the matrix T_{λ} : if we have some solution U of (2.4.2) with $U(a) = c_1 \in \mathbb{C}$, $\partial_x U(a) = c_2 \in \mathbb{C}$, we obtain that $U = c_1 \tilde{\psi}_{\lambda} + c_2 \hat{\psi}_{\lambda}$, so that

$$\begin{pmatrix} U(b)\\ \partial_x U(b) \end{pmatrix} = \begin{pmatrix} c_1 \tilde{\psi}_{\lambda}(b) + c_2 \hat{\psi}_{\lambda}(b)\\ c_1 \partial_x \tilde{\psi}_{\lambda}(b) + c_2 \partial_x \hat{\psi}_{\lambda}(b) \end{pmatrix} = T_{\lambda} \begin{pmatrix} c_1\\ c_2 \end{pmatrix}.$$

Thus the matrix T_{λ} maps the Dirichlet and Neumann boundary data at a to the Dirichlet and Neumann boundary data at b. Now assume $\det(T_{\lambda}) = 0$, so that we have some $c \in \mathbb{C}^2 \setminus \{0\}$ such that $T_{\lambda}c = 0$. Then the function $U = c_1\tilde{\psi}_{\lambda} + c_2\hat{\psi}_{\lambda}$ solves (2.4.2) and

$$\begin{pmatrix} U(b)\\ \partial_x U(b) \end{pmatrix} = T_\lambda c = 0.$$

This, however, implies that U = 0 everywhere by the unique continuation principle, so that we obtain c = 0. This gives a contradiction.

¹Recall that an entire function is a function $\mathbb{C} \to \mathbb{C}$, which is holomorphic on the entirety of \mathbb{C} .

2.4. PROOF OF THE EXPANSION THEOREM

To see that the statements for v_{λ}^+ hold, note that

$$v_{\lambda}^{+} = \psi_{\lambda} - i\sqrt{\lambda - q_{-}} \,\,\hat{\psi}_{\lambda},$$

so that the differentiability properties of the square root (Definition 1.2.1) are inherited.

The following lemma characterises the point spectrum of A with the help of the Wronskian w_{λ} and gives a number of technical properties. The proof is rather long and technical and not very rich in instructive details, so the reader can safely skip it at first reading.

2.4.4 Lemma. $w_{\lambda} : \mathbb{C} \setminus [q_{-}, +\infty) \to \mathbb{C}$ is holomorphic, and can be continuously extended so that the one sided limits

$$\lim_{\substack{\lambda \to \mu \\ \operatorname{Im}(\lambda) > 0}} w_{\lambda} = w_{\mu} \quad \text{for } \mu \in [q_{-}, \infty)$$

exist. We furthermore have:

- (a) $\sigma_p(A) = \{\lambda \in \mathbb{C} : w_\lambda = 0\} \setminus \{q_-\} \subset (q_m, q_-).$
- (b) $\sigma_p(A) = \{\lambda_1, \dots, \lambda_N\}$ is a finite (but possibly empty) set (i.e. $N \in \mathbb{N} \cup \{0\}$).
- (c) If $w_{q_-} = 0$, then we have for λ in some neighbourhood of q_-

$$w_{\lambda} = C\sqrt{\lambda} - q_{-} + \mathcal{O}(|\lambda - q_{-}|),$$

with $C \in \mathbb{C} \setminus \{0\}$.

Proof. Let us rewrite w_{λ} with the help of T_{λ} : We have that

$$w_{\lambda} = \det \begin{pmatrix} v_{\lambda}^{+}(b) & v_{\lambda}^{-}(b) \\ \partial_{x}v_{\lambda}^{+}(b) & \partial_{x}v_{\lambda}^{-}(b) \end{pmatrix} = \det \begin{bmatrix} T_{\lambda} \begin{pmatrix} v_{\lambda}^{+}(a) \\ \partial_{x}v_{\lambda}^{+}(a) \end{pmatrix} \middle| \begin{pmatrix} v_{\lambda}^{-}(b) \\ \partial_{x}v_{\lambda}^{-}(b) \end{pmatrix} \end{bmatrix}$$

Exploiting the boundary conditions for v_{λ}^{\pm} in (2.4.3), we can more explicitly calculate

$$w_{\lambda} = \det \left[\left. T_{\lambda} \begin{pmatrix} 1 \\ -i\sqrt{\lambda - q_{-}} \end{pmatrix} \right| \begin{pmatrix} 1 \\ i\sqrt{\lambda - q_{+}} \end{pmatrix} \right]$$

$$= -t_{21}(\lambda) + i\sqrt{\lambda - q_{+}} t_{22}(\lambda)$$

$$+ i\sqrt{\lambda - q_{-}} t_{11}(\lambda) + \sqrt{\lambda - q_{+}} \sqrt{\lambda - q_{-}} t_{12}(\lambda).$$

(2.4.8)

Since t_{nm} are entire functions of λ , we obtain that $\lambda \mapsto w_{\lambda}$ is holomorphic whenever the two square roots $\sqrt{\cdot - q_{\pm}}$ are holomorphic. This yields the continuity as well as the continuous extension of the one sided limits for real $\lambda \geq q_{-}$ (compare also the choice of the square root in Definition 1.2.1). Let us now show the three assertions (a)-(c).

- (a) To show the equivalence of the sets, we split the proof further into two parts.
 - (i) We firstly show that $\sigma_p(A)$ is contained in (q_m, q_-) and subset of $\{\lambda \in \mathbb{C} : w_\lambda = 0\}$. Let $\lambda \in \sigma_p(A)$ be an eigenvalue, which implies that $\lambda \in \mathbb{R}$, since A is self-adjoint.

Let $U \in H^2(\mathbb{R})$ be the corresponding eigenfunction, which can be expanded in the form (if $\lambda \notin \{q_+, q_-\}$):

$$U(x) = \begin{cases} c_1 e^{i\sqrt{\lambda - q_-}(x-a)} + c_2 e^{-i\sqrt{\lambda - q_-}(x-a)} & \text{if } x < a, \\ c_3 e^{i\sqrt{\lambda - q_+}(x-b)} + c_4 e^{-i\sqrt{\lambda - q_+}(x-b)} & \text{if } x > b. \end{cases}$$

If $\lambda > q_-$, then both terms for x < a do not decay, and hence U = 0 on $(-\infty, a)$, which implies that U = 0 everywhere by the unique continuation principle, a contradiction. By the same argument we can rule out $\lambda \in \{q_+, q_-\}$, with the terms for x < a (or x > b) replaced by $c_1 + c_2 x$. Hence this is not possible, and there are no eigenvalues larger than q_- .

Let $\lambda < q_-$. The terms after the coefficients c_1 and c_4 increase exponentially on their domains, and since $U \in H^2(\mathbb{R})$, we obtain $c_1 = c_4 = 0$, so that

$$U(x) = \begin{cases} c_2 e^{-i\sqrt{\lambda - q_-}(x-a)} & \text{if } x < a, \\ c_3 e^{i\sqrt{\lambda - q_+}(x-b)} & \text{if } x > b. \end{cases}$$
(2.4.9)

Comparing coefficients for x < a, we obtain $U(x) = c_2 v_{\lambda}^+(x)$, while by comparing coefficients for x > b, we obtain $U(x) = c_3 v_{\lambda}^-(x)$. Since $c_2 \neq 0 \neq c_3$, this implies that v_{λ}^+ and v_{λ}^- are linearly dependent, and hence $w_{\lambda} = 0$. This shows that $\sigma_p(A) \subset \{\lambda \in \mathbb{C} : w_{\lambda} = 0\}$.

Let us show that $\lambda \leq q_m$ cannot be an eigenvalue: multiplying $AU(x) = \lambda U(x)$ by $\overline{U}(x)$, integrating over \mathbb{R} and using integration by parts we obtain

$$\int_{\mathbb{R}} |\partial_x U|^2 + (q(x) - \lambda)|U|^2 \,\mathrm{d}x = 0.$$

Since $q(x) - \lambda \geq 0$ for almost all $x \in \mathbb{R}$, this implies that $\int_{\mathbb{R}} |\partial_x U|^2 dx = 0$, so U must be constant, which implies U = 0 since $U \in H^2(\mathbb{R})$. Thus, we have shown that $\sigma_p(A) \subset (q_m, q_-)$.

(ii) Let us now prove that $\sigma_p(A) \supset \{\lambda \in \mathbb{C} : w_\lambda = 0\} \setminus \{q_-\}$, i.e. that any zero of w_λ (except q_-) is also an eigenvalue: let $\lambda \in \mathbb{C}$ be such that $w_\lambda = 0$. So v_λ^+ and v_λ^- are linearly dependent, and we have $c \in \mathbb{C}$ such that $v_\lambda^+ = cv_\lambda^-$, from which we obtain by comparing coefficients for x < a and x > b that v_λ^+ must be of the form (2.4.9).

If $\lambda \in \mathbb{C} \setminus [q_-, +\infty)$, any solution U of the form (2.4.9) is an eigenfunction, since it decays exponentially as $x \to \pm \infty$. Hence v_{λ}^{\pm} are eigenfunctions, implying $\lambda \in \sigma_p(A)$. If $\lambda > q_-$, $\lambda \neq q_+$, we can apply the energy conservation relation (2.2.3) to $v_{\lambda}^+ = cv_{\lambda}^-$ to obtain

$$\operatorname{Re}(\sqrt{\lambda - q_{-}}) + \operatorname{Re}(\sqrt{\lambda - q_{+}})|\beta_{\lambda}^{+}|^{2} = 0,$$

where β_{λ}^{+} is the coefficient in the expansion (2.4.4). This gives a contradiction, since the first term is strictly larger than zero, while the second larger or equal zero.

Note that we have not considered the behaviour of the Wronskian at q_+ : it holds that $w_{q_+} \neq 0$, which will be shown at the end of the proof.

(b) Let us prove that there exist only finitely many eigenvalues, that is, solutions to $w_{\lambda} = 0$. Since all zeros are contained in the bounded interval $(q_m, q_-]$, it is sufficient
to show that there are no accumulation points of zeros in $[q_m, q_-]$. Let us rewrite w_{λ} as

$$w_{\lambda} = i\sqrt{\lambda - q_{-}} \left(\frac{-t_{21}(\lambda) + it_{11}(\lambda)\sqrt{\lambda - q_{+}}}{i\sqrt{\lambda - q_{-}}} - \left(t_{21}(\lambda) - it_{12}(\lambda)\sqrt{\lambda - q_{+}} \right) \right),$$

so that $\lambda \in (-\infty, q_{-})$ solves $w_{\lambda} = 0$ if and only if

$$0 = \frac{g(\lambda)}{i\sqrt{\lambda - q_{-}}} - h(\lambda), \qquad (2.4.10)$$

where the functions g and h are defined by

$$g(\lambda) := -t_{21}(\lambda) + it_{11}(\lambda)\sqrt{\lambda - q_+},$$

$$h(\lambda) := -t_{22}(\lambda) + it_{12}(\lambda)\sqrt{\lambda - q_+}.$$

Now assume there exists an infinite sequence $(\lambda_j)_{j\in\mathbb{N}}$ of solutions to (2.4.10). Since all zeros of (2.4.10) are contained in $(q_m, q_-]$ and since the right hand side of (2.4.10) is holomorphic on $\{\lambda \in \mathbb{C} : \operatorname{Re}(\lambda) < q_-\}$, the sequence $(\lambda_j)_{j\in\mathbb{N}}$ must converge to q_- . Here we used that holomorphic functions cannot have accumulation points in the interior of their domain.

Let us now consider the right hand side of (2.4.10) around q_{-} : since g and h are holomorphic in a neighbourhood of q_{-} , there exist holomorphic functions \tilde{g} and \tilde{h} as well as $p_1, p_2 \in \mathbb{N} \cup \{0\}$ such that

$$g(\lambda) = (\lambda - q_{-})^{p_{1}} \tilde{g}(\lambda),$$

$$h(\lambda) = (\lambda - q_{-})^{p_{2}} \tilde{h}(\lambda),$$

for λ in a neighbourhood of q_{-} . Inserting those into (2.4.10) for $\lambda = \lambda_j$, we obtain

$$0 = \frac{g(\lambda_j)}{i\sqrt{\lambda_j - q_-}} - h(\lambda_j) = \frac{(\lambda_j - q_-)^{p_1}\tilde{g}(\lambda_j)}{i\sqrt{\lambda_j - q_-}} - (\lambda_j - q_-)^{p_2}\tilde{h}(\lambda_j),$$

which can be reformulated after a straightforward manipulation to

$$\tilde{h}(\lambda_j) = (\lambda_j - q_-)^{p_1 - p_2 - \frac{1}{2}} \tilde{g}(\lambda_j).$$

Let us take the limit $j \to +\infty$ in this equations. The left hand side converges to $\tilde{h}(q_{-}) \neq 0$, while the right hand side converges either to 0 (if $p_1 - p_2 - 1/2 > 0$) or to $\pm \infty$ (if $p_1 - p_2 - 1/2 < 0$), depending on the sign of $g(q_{-}) \neq 0$. In any case, this gives a contradiction.

(c) Let now $w_{q_-} = 0$. We will again only consider the case $q_- < q_+$, since the case $q_+ = q_-$ is simpler. Rewriting (2.4.8), we obtain

$$w_{\lambda} = -t_{21}(\lambda) + it_{11}(\lambda)\sqrt{\lambda - q_+} + i\sqrt{\lambda - q_-} \left(t_{22}(\lambda) - it_{12}(\lambda)\sqrt{\lambda - q_+}\right).$$

$$(2.4.11)$$

Setting $\lambda = q_{-}$, we obtain from $w_{q_{-}} = 0$

$$t_{21}(q_{-}) = it_{11}(q_{-})\sqrt{q_{-}-q_{+}}$$

This implies that $t_{22}(q_{-}) - it_{12}(q_{-})\sqrt{q_{-}-q_{+}} \neq 0$: assume on the contrary, that also $t_{22}(q_{-}) - it_{12}(q_{-})\sqrt{q_{-}-q_{+}} = 0$. Then

$$\det T_{q_{-}} = t_{11}(q_{-})t_{22}(q_{-}) - t_{12}(q_{-})t_{21}(q_{-})$$

= $t_{11}(q_{-})it_{12}(q_{-})\sqrt{q_{-} - q_{+}} - t_{12}(q_{-})it_{11}(q_{-})\sqrt{q_{-} - q_{+}}$
= 0,

which cannot happen by Definition and Lemma 2.4.3. Hence we have for λ in a neighbourhood of q_{-}

$$\begin{aligned} -t_{21}(\lambda) + it_{11}(\lambda)\sqrt{\lambda - q_+} &= \mathcal{O}(|\lambda - q_-|), \\ t_{22}(\lambda) - it_{12}(\lambda)\sqrt{\lambda - q_+} &= C + \mathcal{O}(|\lambda - q_-|). \end{aligned}$$

where $C = t_{22}(q_-) - it_{12}(q_-)\sqrt{q_- - q_+} \neq 0$. Note that both functions on the left are holomorphic in a neighbourhood of q_- , and allow a zeroth order Taylor expansion with an error in $\mathcal{O}(|\lambda - q_-|)$. Accordingly

$$w_{\lambda} = \mathcal{O}(|\lambda - q_{-}|) + \sqrt{\lambda - q_{-}}(C + \mathcal{O}(|\lambda - q_{-}|))$$
$$= C\sqrt{\lambda - q_{-}} + \mathcal{O}(|\lambda - q_{-}|).$$

Note also that this approximation holds despite w_{λ} being discontinuous around q_{-} , since the square root $\sqrt{\lambda - q_{-}}$ is discontinuous as a complex function there. As a function on \mathbb{R} , however, w_{λ} is continuous.

Let us finish the proof by picking up the last morsel, namely that $w_{q_+} \neq 0$. Assume that $w_{q_+} = 0$, and consider $\lambda \in (q_-, q_+)$: note that for these λ , the first two summands in (2.4.11) are real, while the third summand is strictly imaginary. Thus from the continuity on the positive real axis we have $w_{\lambda} \to 0$ as $\lambda \to q_+$ from the left, and obtain

$$-t_{21}(\lambda) + it_{11}(\lambda)\sqrt{\lambda - q_+} \to 0 \quad \text{and} \quad t_{22}(\lambda) - it_{12}(\lambda)\sqrt{\lambda - q_+} \to 0$$

as $\lambda \to q_+$, $\lambda < q_+$. This, however, implies that $t_{21}(q_+) = 0$ and $t_{22}(q_+) = 0$, which again implies $\det(T_{q_+}) = 0$, which is forbidden by Definition and Lemma 2.4.3.

We can now give the Green's function of the operator A for the whole resolvent set.

2.4.5 Lemma. For any $\lambda \in \mathbb{C} \setminus (\sigma_p(A) \cup [q_-, \infty))$ and $f \in C_0^{\infty}(\mathbb{R})$, we have

$$R_{\lambda}f(x) = \int_{\mathbb{R}} G_{\lambda}(x, y)f(y) \, \mathrm{d}y,$$

where

$$G_{\lambda}(x,y) = \frac{1}{w_{\lambda}} \begin{cases} v_{\lambda}^{+}(x)v_{\lambda}^{-}(y) & \text{if } x < y, \\ v_{\lambda}^{-}(x)v_{\lambda}^{+}(y) & \text{if } y < x. \end{cases}$$

Proof. We will only sketch the proof here, since the construction is a standard tool in the theory of ordinary differential equations (see for example [73, Section 1.4]). Note that for $\lambda \in \mathbb{C} \setminus (\sigma_p(A) \cup [q_-, +\infty)), G_{\lambda}$ is well defined. Define

$$u(x) := \int_{\mathbb{R}} G_{\lambda}(x, y) f(y) \, \mathrm{d}y$$
$$= \frac{1}{w_{\lambda}} v_{\lambda}^{-}(x) \int_{-\infty}^{x} v_{\lambda}^{+}(y) f(y) \, \mathrm{d}y + \frac{1}{w_{\lambda}} v_{\lambda}^{+}(x) \int_{x}^{+\infty} v_{\lambda}^{-}(y) f(y) \, \mathrm{d}y.$$

Firstly, note that u fulfils

$$-\Delta u(x) + (q(x) - \lambda)u(x) = f(x) \quad \text{for } x \in \mathbb{R},$$

as can be verified via a straightforward calculation: apply Δ to the right hand side of the definition of u, and exploit the fundamental theorem of calculus as well as the fact that v_{λ}^{\pm} solve the homogeneous equation (2.4.2), and keep in mind the definition of the Wronskian (2.4.6). It remains to show that $u \in L^2(\mathbb{R})$. Since clearly $u \in L^2_{loc}(\mathbb{R})$ (in fact, u is easily seen to be continuous), we only check the asymptotics of u, exemplarily for the first term on the right hand side. Since $\lambda \in \mathbb{C} \setminus [q_-, \infty)$, we have that $v_{\lambda}^+(x) \to 0$ as $x \to -\infty$, and $v_{\lambda}^-(x) \to 0$ as $x \to +\infty$ with exponential decay. Now, let us take the limit $x \to +\infty$ in the first term

$$I_1(x) = v_{\lambda}^-(x) \int_{-\infty}^x v_{\lambda}^+(y) f(y) \,\mathrm{d}y.$$

If x is large enough, $(-\infty, x)$ will contain $\operatorname{supp}(f)$, and the integral will stay constant. Thus we obtain that $I_1(x) = Cv_{\lambda}^-(x) \to 0$ exponentially as $x \to \infty$ for some constant $C \in \mathbb{C}$. On the other hand, if we take the limit $x \to -\infty$, we will obtain $(-\infty, x) \cap \operatorname{supp}(f) = \emptyset$ for sufficiently small x, and hence $I_1(x) = 0$. The analogous argument for the second term gives that u(x) converges exponentially to zero as $x \to \pm\infty$, and hence $u \in L^2(\mathbb{R})$, which finishes the proof.

We proceed by showing that the continuous spectrum is the interval $[q_{-}, \infty)$.

2.4.6 Lemma. It holds that

$$\sigma_c(A) = [q_-, +\infty)$$

Proof. By the proof of Lemma 2.4.5, we obtain that the resolvent set $\rho(A) = \{\lambda \in \mathbb{C} : R_{\lambda} : L^2(\mathbb{R}) \to H^2(\mathbb{R}), \text{ exists, is continuous and bijective}\}$ must contain the set $\mathbb{C}\setminus(\sigma_p(A)\cup[q_-,+\infty))$. We are done if we can show that $[q_-,\infty)\subset\sigma(A)$. Let $\lambda\in(q_-,\infty)$, and for $n\in\mathbb{N}$ let $u_n\in H^2(\mathbb{R})$ be a cut-off of v_{λ}^+ such that

$$u_n(x) := \begin{cases} 0 & \text{if } x < -n - 1, \\ v_{\lambda}^-(x) & \text{if } -n < x < n, \\ 0 & \text{if } n + 1 < x, \end{cases}$$

with extensions on the two intervals (-n-1, -n) and (n, n+1) such that $u_n \in C^2(\mathbb{R})$. Note that $Au_n(x) = \lambda u_n(x)$ for $x \in (-n, n)$, and the extensions can be constructed in

a fashion so that $||Au_n - \lambda u_n||_{L^2(\mathbb{R})}$ stays bounded, since v_{λ}^- , $\partial_x v_{\lambda}^-$ are continuous and bounded on \mathbb{R} . Thus

$$\frac{\|Au_n - \lambda u_n\|_{L^2(\mathbb{R})}}{\|u_n\|_{L^2(\mathbb{R})}} \to 0 \quad \text{as } n \to \infty,$$

which implies by Weyl's criterion [67, Theorem VII.12] that $\lambda \in \sigma(A)$. The Lemma then follows by taking the characterisation of $\sigma_p(A)$ of Lemma 2.4.4 and the closedness of the spectrum into account.

The following Lemma gives another technical property for the Wronskian, which will be needed for the integrals related to the singularities of the Green's function.

2.4.7 Lemma. Let $\lambda_n \in \sigma_p(A)$, so that $cv_{\lambda_n}^+ = v_{\lambda_n}^-$ for some $c \neq 0$. Then

$$[\partial_{\lambda} w_{\lambda}]|_{\lambda = \lambda_n} = -c \|v_{\lambda_n}^+\|_{L^2(\mathbb{R})}^2.$$

Proof. For some function f depending on λ , we denote by an overdot the derivative with respect to λ , that is $\dot{f} := \partial_{\lambda} f$. We aim to compute for some $y \in \mathbb{R}$

$$\dot{w}_{\lambda} = \partial_{\lambda} W(v_{\lambda}^+, v_{\lambda}^-)(y) = W(\dot{v}_{\lambda}^+, v_{\lambda}^-)(y) + W(v_{\lambda}^+, \dot{v}_{\lambda}^-)(y).$$

We explicitly denoted the dependence on the point $y \in \mathbb{R}$, since the terms on the right hand side depend on y, while the left hand side is constant (as a function of y). First note that from differentiating (2.4.2) with respect to λ we obtain that

$$-\Delta \dot{v}_{\lambda}^{+}(x) - (\lambda - q(x))\dot{v}_{\lambda}^{+}(x) = v_{\lambda}^{+}(x) \quad \text{for } x \in \mathbb{R},$$
(2.4.12)

while from differentiating (2.4.4) we obtain that

$$\dot{v}_{\lambda}^{+}(x) = -\frac{i(x-a)}{2\sqrt{\lambda - q_{-}}} e^{-i\sqrt{\lambda - q_{-}}(x-a)} \quad \text{for } x < a.$$
(2.4.13)

With the help of (2.4.2) and (2.4.12), one easily verifies by straightforward calculation that for any $x \in \mathbb{R}$

$$\partial_x W(\dot{v}^+_{\lambda}, v^-_{\lambda})(x) = -v^+_{\lambda}(x)v^-_{\lambda}(x).$$

Setting $\lambda = \lambda_n$ in this equation and integrating with respect to x form $y_0 < y$ to y we obtain by the fundamental theorem of calculus

$$W(\dot{v}_{\lambda_n}^+, v_{\lambda_n}^-)(y) - W(\dot{v}_{\lambda_n}^+, v_{\lambda_n}^-)(y_0) = -\int_{y_0}^y v_{\lambda_n}^+(x)v_{\lambda_n}^-(x)\,\mathrm{d}x.$$

Since $v_{\lambda_n}^-(x) = cv_{\lambda_n}^+(x)$ and $\dot{v}_{\lambda_n}^-(x)$ as well as their derivatives converge to 0 as $x \to -\infty$, as can be seen from (2.4.4) and (2.4.13), we can take the limit $y_0 \to -\infty$ in the last equation to obtain

$$W(\dot{v}_{\lambda_n}^+, v_{\lambda_n}^-)(y) = -\int_{-\infty}^y v_{\lambda_n}^+(x)v_{\lambda_n}^-(x)\,\mathrm{d}x = -c\int_{-\infty}^y |v_{\lambda_n}^+(x)|^2\,\mathrm{d}x.$$

Note that we used the fact that $v_{\lambda_n}^+(x) \in \mathbb{R}$ for any $x \in \mathbb{R}$, since the PDE as well as the boundary conditions for $v_{\lambda_n}^+$ are real (due to $\lambda_n \in (q_m, q_-)$ by Lemma 2.4.4). Similarly, one can prove that

$$W(v_{\lambda_n}^+, \dot{v}_{\lambda_n}^-)(y) = -c \int_y^\infty |v_{\lambda_n}^+(x)|^2 \,\mathrm{d}x,$$

so that we obtain by adding the last two equations

$$\dot{w}_{\lambda} = -c \int_{\mathbb{R}} |v_{\lambda_n}^+(x)|^2 \,\mathrm{d}x,$$

which ends the proof.

We have to clarify the relation between the expansion basis $\Psi^{\pm}(\lambda, \cdot)$ and our solutions v_{λ}^{\pm} .

2.4.8 Lemma. For any $\sigma \in \{\pm\}$ and $\lambda \in (q_{\sigma}, \infty)$ it holds that

$$\frac{v_{\lambda}^{\sigma}(x)}{|a_{\lambda}^{\sigma}|} = c_{\lambda}^{\sigma} \Psi^{\sigma}(\lambda, x) \qquad \text{for } x \in \mathbb{R},$$

with $|c_{\lambda}^{\sigma}| = 1$. Furthermore, there exist constants C_1, C_2 such that for all $m \in \Lambda$ and $x \in \mathbb{R}$

$$|\Psi(m,x)| < C_1, \ |\partial_x \Psi(m,x)| < C_2 \left(1 + \sqrt{|\hat{\lambda}(m)|}\right).$$
 (2.4.14)

Proof. Note that

$$\frac{v_{\lambda}^{\sigma}(x)}{|a_{\lambda}^{\sigma}|} = c_{\lambda}^{\sigma} \Psi^{\sigma}(\lambda, x) \qquad \text{for } x \in \mathbb{R},$$

with $|c_{\lambda}^{\sigma}| = 1$ follows easily by comparing the coefficients of Ψ^- (see Definition 2.2.11) and v_{λ}^- (see (2.4.4)), and similarly for Ψ^+ and v_{λ}^+ . For the proper eigenfunctions, the statement simply holds by definition.

It remains to show that the bounds for the (generalised) eigenfunctions as well as their derivatives hold. Let us first recall that the mappings $\mathbb{R} \times \mathbb{R} \to \mathbb{C}$, $(\lambda, x) \mapsto v_{\lambda}^{\sigma}(x)$ are continuous. Also, note that $\alpha_{\lambda}^{\sigma}$ cannot vanish for $\lambda \in [q_{\sigma}, \infty)$: considering v_{λ}^{-} , one obtains for $\lambda \in (q_{-}, q_{+})$ that there holds the energy conservation property (2.2.3)

$$\sqrt{\lambda - q_{-}} |\alpha_{\lambda}^{-}|^{2} = \sqrt{\lambda - q_{-}} |\beta_{\lambda}^{-}|^{2},$$

so that $\alpha_{\lambda}^{-} = 0$ would yield $\beta_{\lambda}^{-} = 0$, giving $v_{\lambda}^{-} = 0$, and by the unique continuation principle, a contradiction. This argument can be continued to obtain that $\lim_{\lambda \to q_{-}} |\alpha_{\lambda}^{-}| \neq 0$, since this would imply that $v_{q_{-}}^{-}$ would vanish. The same can be shown for $\lambda \in [q_{+}, \infty)$. Here we have by (2.2.3) that

$$\sqrt{\lambda - q_{-}} |\alpha_{\lambda}^{-}|^{2} = \sqrt{\lambda - q_{+}} + \sqrt{\lambda - q_{-}} |\beta_{\lambda}^{-}|^{2},$$

which implies by the same argument as before that α_{λ}^{-} stays away from zero.

Hence the continuity and boundedness of the coefficients of v_{λ}^- is inherited by the coefficients of $\Psi^-(\lambda, \cdot)$. Furthermore, considering the asymptotics of $\Psi^-(\lambda, x)$ as $x \to \pm \infty$ with help of (2.4.5), one obtains that the bounds (2.4.14) hold for all $x \in \mathbb{R}$ and for

 λ on any bounded subset. The critical point is to estimate the terms as $\lambda \to +\infty$. We refer to [50, Theorem 4.5]: from this theorem it easily follows that for any $x \in (a, b)$ one has

$$\begin{aligned} |v_{\lambda}^{-}(x) - e^{-\sqrt{\lambda - q_{-}}(x-a)}| &\leq \frac{C}{1 + \sqrt{|\lambda|}}, \\ \text{and } |\partial_{x}v_{\lambda}^{-}(x) - \partial_{x}e^{-\sqrt{\lambda - q_{-}}(x-a)}| &\leq \frac{C}{1 + \sqrt{|\lambda|}}, \end{aligned}$$

so that one obtains the bounds (2.4.14) immediately for $x \in (a, b)$, while they follow for x > b and x < a directly by (2.4.5) (Here we used again that α_{λ} cannot vanish). For $\Psi^+(\lambda, \cdot)$, the proof is the same. Note that the estimates for the proper eigenfunctions $\Psi^p(\lambda_n, \cdot)$ are also easily seen to hold.

The definition of the generalised eigenfunctions will play a crucial role in then following lemma. In fact, this is a critical step of the whole process: we rewrite the imaginary part of the Green's function for $\lambda \in (q_-, \infty)$ with help of the generalised eigenfunctions.

2.4.9 Lemma. For any $x, y \in \mathbb{R}$ and real λ , we have the following representations

$$G_{\lambda}(x,y) - \overline{G}_{\lambda}(x,y) = \begin{cases} 0 & \text{if } \lambda \in (-\infty, q_{-}) \setminus \sigma_{p}(A), \\ \frac{\Psi^{-}(\lambda, x)\overline{\Psi}^{-}(\lambda, y)}{-2i\sqrt{\lambda - q_{-}}} & \text{if } \lambda \in (q_{-}, q_{+}), \\ \sum_{\sigma \in \{\pm\}} \frac{\Psi^{\sigma}(\lambda, x)\overline{\Psi}^{\sigma}(\lambda, y)}{-2i\sqrt{\lambda - q_{\sigma}}} & \text{if } \lambda \in (q_{+}, +\infty). \end{cases}$$

Proof. We only consider the case x > y, since the case x < y then follows by symmetry. We have to recompute

$$G_{\lambda}(x,y) - \overline{G}_{\lambda}(x,y) = \frac{1}{w_{\lambda}}v_{\lambda}^{-}(x)v_{\lambda}^{+}(y) - \frac{1}{\overline{w}_{\lambda}}\overline{v}_{\lambda}^{-}(x)\overline{v}_{\lambda}^{+}(y).$$

We will consider the three cases for λ :

(a) If $\lambda \in (-\infty, q_-) \setminus \sigma_p(A)$, then $w_{\lambda}, v_{\lambda}^{\pm}$ are all strictly real, hence it follows that $G_{\lambda}(x, y) - \overline{G}_{\lambda}(x, y) = 0.$

(b) Let $\lambda \in (q_-, q_+)$. Then v_{λ}^- is real, as can be seen from (2.4.4) for x < a, and the fact that (2.4.2) has strictly real coefficients. Accordingly, it follows from (2.4.4) for x > b that $\alpha_{\lambda}^- = \overline{\beta_{\lambda}}$. In other words, $v_{\lambda}^- = \overline{v_{\lambda}}$.

Since \overline{v}^+_{λ} solves the eigenvalue equation (2.4.2), it must be a linear combination of v^-_{λ} and v^+_{λ} . Here we obtain by comparing coefficients for x < a

$$\begin{aligned} \overline{v}_{\lambda}^{+}(x) &= e^{i\sqrt{\lambda-q_{-}} (x-a)} \\ &= \frac{1}{\alpha_{\lambda}^{-}} \left(\alpha_{\lambda}^{-} e^{i\sqrt{\lambda-q_{-}} (x-a)} + \beta_{\lambda}^{-} e^{-i\sqrt{\lambda-q_{-}} (x-a)} \right) - \frac{\beta_{\lambda}^{-}}{\alpha_{\lambda}^{-}} e^{-i\sqrt{\lambda-q_{a}} (x-a)} \\ &= \frac{1}{\alpha_{\lambda}^{-}} v_{\lambda}^{-}(x) - \frac{\beta_{\lambda}^{-}}{\alpha_{\lambda}^{-}} v_{\lambda}^{+}(x). \end{aligned}$$

2.4. PROOF OF THE EXPANSION THEOREM

So with help of the representation for w_{λ} from equation (2.4.7) it follows

$$\begin{aligned} G_{\lambda}(x,y) &- \overline{G}_{\lambda}(x,y) = \frac{1}{w_{\lambda}} v_{\lambda}^{-}(x) v_{\lambda}^{+}(y) - \frac{1}{\overline{w}_{\lambda}} \overline{v}_{\lambda}^{-}(x) \overline{v}_{\lambda}^{+}(y) \\ &= v_{\lambda}^{-}(x) \left(\frac{1}{w_{\lambda}} v_{\lambda}^{+}(x) - \frac{1}{\overline{w}_{\lambda}} \left(\frac{1}{\alpha_{\lambda}^{-}} v_{\lambda}^{-}(x) - \frac{\beta_{\lambda}^{-}}{\alpha_{\lambda}^{-}} v_{\lambda}^{+}(x) \right) \right) \\ &= v_{\lambda}^{-}(x) \frac{1}{2i\sqrt{\lambda - q_{-}}} \left(\frac{1}{-\alpha_{\lambda}^{-}} v_{\lambda}^{+}(x) - \frac{1}{\overline{\alpha_{\lambda}^{-}}} \frac{1}{\alpha_{\lambda}^{-}} v_{\lambda}^{-}(x) + \frac{1}{\overline{\alpha_{\lambda}^{-}}} \frac{\beta_{\lambda}^{-}}{\alpha_{\lambda}^{-}} v_{\lambda}^{+}(x) \right). \end{aligned}$$

Since $\alpha_{\lambda}^{-} = \overline{\beta}_{\lambda}^{-}$ the terms inside the bracket with $v_{\lambda}^{+}(x)$ cancel out, and we obtain

$$G_{\lambda}(x,y) - \overline{G}_{\lambda}(x,y) = \frac{1}{-2i\sqrt{\lambda - q_{-}}|\alpha_{\lambda}^{i}|^{2}}v_{\lambda}^{-}(x)\overline{v}_{\lambda}^{-}(y),$$

where we used again that v_{λ}^{-} is real. Applying Lemma 2.4.8 yields the representation.

(c) Now let $\lambda > q_+$. In this case, neither v_{λ}^+ nor v_{λ}^- are real, so we have to calculate the representation of both conjugates \overline{v}_{λ}^+ and \overline{v}_{λ}^- . By comparing coefficients for x < a, we obtain

$$\overline{v}_{\lambda}^{+}(x) = e^{i\sqrt{\lambda - q_{-}}(x - a)} = \frac{1}{\alpha_{\lambda}^{-}}v_{\lambda}^{-}(x) - \frac{\beta_{\lambda}^{-}}{\alpha_{\lambda}^{-}}v_{\lambda}^{+}(x).$$

Similarly, we can obtain another representation by comparing the coefficients for x > b, which yields (we omit the details of the calculations here):

$$\overline{v}_{\lambda}^{+}(x) = \frac{|\alpha_{\lambda}^{+}|^{2} - |\beta_{\lambda}^{+}|^{2}}{\alpha_{\lambda}^{+}}v_{\lambda}^{-}(x) + \frac{\overline{\beta}_{\lambda}^{+}}{\alpha_{\lambda}^{+}}v_{\lambda}^{+}(x).$$

If we compare the two different representations for \overline{v}_{λ}^+ , we obtain by considering the coefficients in front of v_{λ}^+ on the right hand side that

$$\frac{\overline{\beta}_{\lambda}^{+}}{\alpha_{\lambda}^{+}} = -\frac{\beta_{\lambda}^{-}}{\alpha_{\lambda}^{-}}.$$
(2.4.15)

•

Repeating the same procedure for $\overline{v}_{\lambda}^{-}$, we end up with the following "conjugation matrix":

$$\begin{pmatrix} \overline{v}_{\lambda}^{-}(x) \\ \overline{v}_{\lambda}^{+}(x) \end{pmatrix} = \begin{pmatrix} -\frac{\beta_{\lambda}}{\alpha_{\lambda}^{+}} & \frac{1}{\alpha_{\lambda}^{+}} \\ \frac{1}{\alpha_{\lambda}^{-}} & -\frac{\beta_{\lambda}^{-}}{\alpha_{\lambda}^{-}} \end{pmatrix} \begin{pmatrix} v_{\lambda}^{-}(x) \\ v_{\lambda}^{+}(x) \end{pmatrix} =: K \begin{pmatrix} v_{\lambda}^{-}(x) \\ v_{\lambda}^{+}(x) \end{pmatrix}$$

Note that taking the conjugate of the last equation one obtains $K^{-1} = \overline{K}$. Let us use the conjugation matrix K to rewrite the Greens function

$$\begin{aligned} G_{\lambda}(x,y) &- \overline{G}_{\lambda}(x,y) = \frac{1}{w_{\lambda}} v_{\lambda}^{-}(x) v_{\lambda}^{+}(y) - \frac{1}{\overline{w}_{\lambda}} \overline{v}_{\lambda}^{-}(x) \overline{v}_{\lambda}^{+}(y) \\ &= \begin{pmatrix} v_{\lambda}^{-}(x) \\ v_{\lambda}^{+}(x) \end{pmatrix}^{\top} \begin{pmatrix} 0 & 1/w_{\lambda} \\ 0 & 0 \end{pmatrix} \begin{pmatrix} v_{\lambda}^{-}(y) \\ v_{\lambda}^{+}(y) \end{pmatrix} + \begin{pmatrix} \overline{v}_{\lambda}^{-}(x) \\ \overline{v}_{\lambda}^{+}(x) \end{pmatrix}^{\top} \begin{pmatrix} 0 & 1/\overline{w}_{\lambda} \\ 0 & 0 \end{pmatrix} \begin{pmatrix} \overline{v}_{\lambda}^{-}(y) \\ \overline{v}_{\lambda}^{+}(y) \end{pmatrix} \\ &= \begin{pmatrix} v_{\lambda}^{-}(x) \\ v_{\lambda}^{+}(x) \end{pmatrix}^{\top} \begin{bmatrix} \begin{pmatrix} 0 & 1/w_{\lambda} \\ 0 & 0 \end{pmatrix} \overline{K} + K^{\top} \begin{pmatrix} 0 & 1/\overline{w}_{\lambda} \\ 0 & 0 \end{pmatrix} \end{bmatrix} \begin{pmatrix} \overline{v}_{\lambda}^{-}(y) \\ \overline{v}_{\lambda}^{+}(y) \end{pmatrix}, \end{aligned}$$

where we used K to transform the v_{λ}^{\pm} vector on the right in the first summand, and on the left for the second summand. Explicitly calculating the matrix in the middle we obtain

$$G_{\lambda}(x,y) - \overline{G}_{\lambda}(x,y) = \begin{pmatrix} v_{\lambda}^{-}(x) \\ v_{\lambda}^{+}(x) \end{pmatrix}^{\top} \begin{pmatrix} \frac{1}{w_{\lambda}\overline{\alpha_{\lambda}^{-}}} & \frac{\beta_{\lambda}^{+}}{\overline{w}_{\lambda}\alpha_{\lambda}^{+}} - \frac{\overline{\beta_{\lambda}^{-}}}{\overline{w}_{\lambda}\overline{\alpha_{\lambda}^{-}}} \\ 0 & -\frac{1}{\overline{w}_{\lambda}\alpha_{\lambda}^{+}} \end{pmatrix} \begin{pmatrix} \overline{v}_{\lambda}^{-}(y) \\ \overline{v}_{\lambda}^{+}(y) \end{pmatrix}.$$
(2.4.16)

Let us recalculate each of the entries. We have by (2.4.7)

$$\frac{1}{w_{\lambda}\overline{\alpha}_{\lambda}^{-}} = \frac{1}{-2i\sqrt{\lambda - q_{-}}|\alpha_{\lambda}^{-}|^{2}} \quad \text{and} \quad -\frac{1}{\overline{w}_{\lambda}\alpha_{\lambda}^{+}} = \frac{1}{-2i\sqrt{\lambda - q_{+}}|\alpha_{\lambda}^{+}|^{2}}$$

For the third coefficient we can calculate with the help of (2.4.15) and (2.4.7):

$$\frac{\beta_{\lambda}^{+}}{\overline{w}\alpha_{\lambda}^{+}} - \frac{\overline{\beta}_{\lambda}^{-}}{w\overline{\alpha}_{\lambda}^{-}} = \frac{\beta_{\lambda}^{+}}{\overline{w}\alpha_{\lambda}^{+}} + \frac{\beta_{\lambda}^{+}}{w\overline{\alpha}_{\lambda}^{+}}$$
$$= \beta_{\lambda}^{+} \left(\frac{1}{2i\sqrt{\lambda - q_{+}}|\alpha_{\lambda}^{+}|^{2}} + \frac{1}{-2i\sqrt{\lambda - q_{+}}|\alpha_{\lambda}^{+}|^{2}} \right)$$
$$= 0.$$

Plugging this into (2.4.16) and applying Lemma 2.4.8 ends the proof.

We are finally done with technical preliminaries, and can apply Stone's formula. 2.4.10 Lemma. For any $f \in C_0^{\infty}(\mathbb{R})$ and any $h_1 < h_2$ with $h_1, h_2 \notin \sigma_p(A)$ we have

$$\frac{1}{2} \left(P_{(h_1,h_2)} f(x) + P_{[h_1,h_2]} f(x) \right) \\
= \sum_{\lambda_n \in (h_1,h_2)} \Psi^p(\lambda_n, x) \langle f, \Psi^p(\lambda_n, \cdot) \rangle_{\mathbb{R}} \\
+ \frac{1}{2\pi} \sum_{\sigma \in \{\pm\}} \int_{\max\{h_1,q_\sigma\}}^{\max\{h_2,q_\sigma\}} \Psi^\sigma(\lambda, x) \langle f, \Psi^\sigma(\lambda, \cdot) \rangle_{\mathbb{R}} \frac{\mathrm{d}\lambda}{2\sqrt{\lambda - q_\sigma}}$$

Proof. We assume that $h_1 < q_-$ and $q_+ < h_2$ for simplicity, to have all relevant terms appear in the calculation. We rewrite Stone's formula into two parts, which we will



Figure 2.4: The paths $\gamma_{\epsilon,r}^+, \gamma_{\epsilon,r}^-, \gamma_{0,r}^+, \gamma_{0,r}^-$ from the proof of Lemma 2.4.10.

consider separately

$$\frac{1}{2} \left(P_{(h_1,h_2)} f(x) + P_{[h_1,h_2]} f(x) \right) = \frac{1}{2\pi i} \lim_{\epsilon \to 0} \int_{h_1}^{h_2} R_{\lambda+i\epsilon} f(x) - R_{\lambda-i\epsilon} f(x) \, \mathrm{d}\lambda$$
$$= \frac{1}{2\pi i} \left(\lim_{\epsilon \to 0} \int_{h_1}^{q_--r} R_{\lambda+i\epsilon} f(x) - R_{\lambda-i\epsilon} f(x) \, \mathrm{d}\lambda + \lim_{\epsilon \to 0} \int_{q_--r}^{h_2} R_{\lambda+i\epsilon} f(x) - R_{\lambda-i\epsilon} f(x) \, \mathrm{d}\lambda \right)$$
$$= \frac{1}{2\pi i} (I_1 + I_2),$$

where r > 0 is sufficiently small such that $\sigma_p(A) \subset (-\infty, q_- - r)$. We rewrite I_1 as follows

$$I_1 = \lim_{\epsilon \to 0} \left(\int_{\gamma_{\epsilon,r}^+} R_\lambda f(x) \, \mathrm{d}\lambda + \int_{\gamma_{\epsilon,r}^-} R_\lambda f(x) \, \mathrm{d}\lambda \right),$$

where $\gamma_{\epsilon,r}^+$ and $\gamma_{\epsilon,r}^-$ are the paths as shown in Figure 2.4, which go from $h_1 + i\epsilon$ to $-q_m - r + i\epsilon$ (and from $-q_m - r - i\epsilon$ to $h_1 - i\epsilon$, respectively) and avoid the singularities of $G_{\lambda}(x, y)$ at the elements of the point spectrum. Taking the limit $\epsilon \to 0$, and noting that the parts of $\gamma_{0,r}^+$ and $\gamma_{0,r}^-$ on the real line cancel each other out, we obtain

$$I_1 = \int_{\gamma_{0,r}^+} R_{\lambda} f(x) \,\mathrm{d}\lambda + \int_{\gamma_{0,r}^-} R_{\lambda} f(x) \,\mathrm{d}\lambda = \sum_{\lambda_n \in (h_1, -q_m)} \int_{\lambda_n + rS^1} R_{\lambda} f(x) \,\mathrm{d}\lambda,$$

where $\lambda_n + rS^1 = \{z \in \mathbb{C} : z = \lambda_n + re^{-i\theta}, \theta \in (0, 2\pi)\}$ denotes the clockwise circlic path around λ_n with radius r. In other words, we have to calculate the residuals of $R_{\lambda}f(x)$ at the eigenvalues λ_n . Representing the resolvent (Lemma 2.4.5) by the Green's function and using Fubini's Theorem we obtain

$$\int_{\lambda_n+rS^1} R_{\lambda}f(x) \, \mathrm{d}\lambda = \int_{\lambda_n+rS^1} \int_{\mathbb{R}} G_{\lambda}(x,y)f(y) \, \mathrm{d}y \, \mathrm{d}\lambda$$
$$= \int_{-\infty}^{x} \int_{\lambda_n+rS^1} \frac{1}{w_{\lambda}} v_{\lambda}^-(x) v_{\lambda}^+(y) \, \mathrm{d}\lambda f(y) \, \mathrm{d}y$$
$$+ \int_{x}^{+\infty} \int_{\lambda_n+rS^1} \frac{1}{w_{\lambda}} v_{\lambda}^+(x) v_{\lambda}^-(y) \, \mathrm{d}\lambda f(y) \, \mathrm{d}y.$$
(2.4.17)

Let us take the limit $r \to 0$ of the first integral: the mapping $\lambda \mapsto v_{\lambda}^{\pm}(x)$ converges (uniformly with respect to x in some compact subset) as $\lambda \to \lambda_n$ by Lemma 2.4.3. However, $1/w_{\lambda}$ is singular, but we have $w_{\lambda} = (\partial_{\lambda}w_{\lambda_n}) \cdot (\lambda - \lambda_n) + \mathcal{O}(|\lambda - \lambda_n|^2)$ by Taylor expansion around λ_n , where $\partial_{\lambda}w_{\lambda_n} = -c ||v_{\lambda_n}^+||_{L^2(\mathbb{R})}^2 \neq 0$ by Lemma 2.4.7 ($c \in \mathbb{C} \setminus \{0\}$ being the constant from the very same lemma), so that

$$\int_{\lambda_n+rS^1} \frac{1}{w_{\lambda}} v_{\lambda}^-(x) v_{\lambda}^+(y) \, \mathrm{d}\lambda = \int_0^{2\pi} \frac{1}{(\partial_{\lambda} w_{\lambda_n}) r e^{-i\theta} + \mathcal{O}(r^2)} v_{\lambda}^-(x) v_{\lambda}^+(y) (-ire^{-i\theta}) \, \mathrm{d}\theta$$
$$= -\frac{2\pi i}{\partial_{\lambda} w_{\lambda_n}} v_{\lambda_n}^-(x) v_{\lambda_n}^+(y), \quad \text{by letting } r \to 0.$$

In view of $v_{\lambda_n}^- = c v_{\lambda_n}^+$ (see again Lemma 2.4.7), we obtain that

$$-\frac{2\pi i}{\partial_{\lambda}w_{\lambda_n}}v_{\lambda_n}^+(x)v_{\lambda_n}^-(y) = -\frac{2\pi i}{-c\|v_{\lambda_n}^+\|_{L^2(\mathbb{R})}^2} \left(cv_{\lambda_n}^+(x)\right)v_{\lambda_n}^+(y)$$
$$= 2\pi i\Psi^p(\lambda_n, x)\Psi^p(\lambda_n, y),$$

since $\Psi^p(\lambda_n, x) = \|v_{\lambda_n}^+\|_{L^2(\mathbb{R})}^{-1} v_{\lambda_n}^+(x)$ is the normalised eigenfunction corresponding to λ_n (see Theorem 2.2.3). Performing the same calculation for the second integral of (2.4.17), we obtain that we can rewrite (2.4.17) as

$$\int_{\lambda_n+rS^1} R_{\lambda}f(x) \, \mathrm{d}\lambda$$

= $2\pi i \int_{-\infty}^x \Psi^p(\lambda_n, x) \Psi^p(\lambda_n, y) f(y) \, \mathrm{d}y + 2\pi i \int_x^\infty \Psi^p(\lambda_n, x) \Psi^p(\lambda_n, y) f(y) \, \mathrm{d}y$
= $2\pi i \Psi^p(\lambda_n, x) \langle f, \Psi^p(\lambda_n, \cdot) \rangle_{\mathbb{R}}$

where we used that $\Psi^p(\lambda_n, \cdot)$ is strictly real, that is $\Psi^p(\lambda_n, \cdot) = \overline{\Psi}^p(\lambda_n, \cdot)$. So we arrive at our final result for I_1 , namely that

$$I_1 = 2\pi i \sum_{\substack{n \in \{1, \dots, N\}\\\lambda_n \in (h_1, -q_m)}} \Psi^p(\lambda_n, x) \langle f, \Psi^p(\lambda_n, \cdot) \rangle_{\mathbb{R}}.$$

Now let us consider I_2 , given by

$$I_2 = \lim_{\epsilon \to 0} \left(\int_{q_--r}^{h_2} R_{\lambda+i\epsilon} f(x) \, \mathrm{d}\lambda - \int_{q_--r}^{h_2} R_{\lambda-i\epsilon} f(x) \, \mathrm{d}\lambda \right).$$

Let us consider the first integral, and recall we still keep $x \in \mathbb{R}$ fixed. By Lemma 2.4.5 we have

$$\lim_{\epsilon \to 0} \int_{q_--r}^{h_2} R_{\lambda+i\epsilon} f(x) \, \mathrm{d}\lambda = \lim_{\epsilon \to 0} \int_{q_--r}^{h_2} \int_{\mathbb{R}} G_{\lambda+i\epsilon}(x,y) f(y) \, \mathrm{d}y \, \mathrm{d}\lambda.$$

We want to exchange integration and limit. The dominated convergence theorem tells us that this is possible, provided we find some integrable $g: [q_- - r, h_2] \times \operatorname{supp}(f) \to \mathbb{R}$ such that

$$|G_{\lambda+i\epsilon}(x,y)f(y)| \le g(\lambda,y)$$
 for all $\lambda \in [q_- - r, h_2], y \in \operatorname{supp}(f)$ and $\epsilon \in [0, \epsilon_0].$

Since the two functions $(\lambda, x) \mapsto v_{\lambda}^{\pm}(x)$ are continuous on the compact set

 $M = \{ (\lambda_0 + i\epsilon, x) \in \mathbb{C} \times \mathbb{R} : \lambda_0 \in [q_- - r, h_2], \ \epsilon \in [0, \epsilon_0], \ x \in \mathrm{supp}(f) \}$

by Lemma 2.4.3, we obtain a constant B > 0 such that

 $|v_{\lambda+i\epsilon}^{\pm}(y)| \leq B$ for all $\lambda \in [q_{-} - r, h_2], y \in \operatorname{supp}(f)$ and $\epsilon \in [0, \epsilon_0],$

and it follows that

$$|G_{\lambda+i\epsilon}(x,y)f(y)| \le \frac{B^2 ||f||_{L^{\infty}(\mathbb{R})}}{|w_{\lambda+i\epsilon}|},$$

so we only have to find a majorant for $1/|w_{\lambda+i\epsilon}|$. The map $\lambda \mapsto w_{\lambda}$ is continuous for $\lambda \in \hat{M} = \{\lambda \in \mathbb{C} : \operatorname{Re}(\lambda) \in [q_- r, h_2], \operatorname{Im}(\lambda) \in [0, \epsilon_0]\}$, and it can only be zero at q_- . If w_{λ} has no zero at q_- , we are done, since then $1/|w_{\lambda}| \leq C$ for $\lambda \in \hat{M}$, and we have found a majorant. Assume that $w_{q_-} = 0$. Then we have according to Lemma 2.4.4 (c) some constant $C \in \mathbb{C} \setminus \{0\}$ such that for $\lambda + i\epsilon$ in a neighbourhood of q_- :

$$\frac{1}{|w_{\lambda+i\epsilon}|} = \frac{1}{|C\sqrt{\lambda+i\epsilon-q_-} + \mathcal{O}(|\lambda+i\epsilon-q_-|)|} \le \frac{\tilde{C}}{\sqrt{|\lambda-q_-|}},$$

where the last estimate can be seen by elementary calculation, provided that the neighbourhood is small enough. Outside this neighbourhood $\lambda \mapsto w_{\lambda}$ is continuous and has no zeros, hence $1/w_{\lambda} \leq \hat{C}$. So we obtain that

$$\frac{1}{|w_{\lambda+i\epsilon}|} \le \frac{\tilde{C}}{\sqrt{|\lambda-q_-|}} + \hat{C} \quad \text{for any } \lambda \in [q_- - r, h_2], \epsilon \in [0, \epsilon_0].$$

The right hand side is integrable, and hence we have found the majorant

$$g(\lambda, y) = B^2 \|f\|_{L^{\infty}(\mathbb{R})} \left(\frac{\tilde{C}}{\sqrt{|\lambda - q_-|}} + \hat{C} \right),$$

which is integrable on $[q_- - r, h_2] \times \text{supp}(f)$. By the dominated convergence theorem we now obtain

$$\lim_{\epsilon \to 0} \int_{q_--r}^{h_2} R_{\lambda+i\epsilon} f(x) \, \mathrm{d}\lambda = \int_{q_--r}^{h_2} \int_{\mathbb{R}} G_{\lambda}(x,y) f(y) \, \mathrm{d}y \, \mathrm{d}\lambda.$$

For the second integral in I_2 , we cannot do the very same procedure, since $G_{\lambda-i\epsilon}(x,y)$ does not converge to $G_{\lambda}(x,y)$ as $\epsilon \to 0$, since the square root is chosen precisely so that the jump happens at the point $\epsilon = 0$. However, since A has purely real coefficients, one easily verifies that we must have $G_{\lambda}(x,y) = \overline{G_{\lambda}}(x,y)$, so that

$$\begin{split} \lim_{\epsilon \to 0} \int\limits_{q_{-}-r}^{h_{2}} R_{\lambda-i\epsilon}f(x) \, \mathrm{d}\lambda &= \lim_{\epsilon \to 0} \int\limits_{q_{-}-r}^{h_{2}} \int\limits_{\mathbb{R}} \overline{G}_{\lambda+i\epsilon}(x,y)f(y) \, \mathrm{d}y \, \mathrm{d}\lambda \\ &= \int\limits_{q_{-}-r}^{h_{2}} \int\limits_{\mathbb{R}} \overline{G}_{\lambda}(x,y)f(y) \, \mathrm{d}y \, \mathrm{d}\lambda. \end{split}$$

From this now follows

$$I_{2} = \int_{q_{-}}^{h_{2}} \int_{\mathbb{R}} \left(G_{\lambda}(x, y) - \overline{G}_{\lambda}(x, y) \right) f(y) \, \mathrm{d}y \, \mathrm{d}\lambda$$

where we used that $G_{\lambda}(x, y) = \overline{G}_{\lambda}(x, y)$ for $\lambda < q_{-}$. An application of Lemma 2.4.9 to rewrite $G_{\lambda}(x, y) - \overline{G}_{\lambda}(x, y)$ and trivial rearrangements now end the proof.

2.4.11 Corollary. For $\rho > ||q||_{L^{\infty}(\mathbb{R})}$ and $f \in C_0^{\infty}(\mathbb{R})$ we have

$$P_{(-\rho,\rho)}f(x) = \sum_{n=1}^{N} \Psi^{p}(\lambda_{n}, x) \langle f, \Psi^{p}(\lambda_{n}, \cdot) \rangle_{\mathbb{R}} + \frac{1}{2\pi} \sum_{\sigma=\pm} \int_{q_{\sigma}}^{\rho} \Psi^{\sigma}(\lambda, x) \langle f, \Psi^{\sigma}(\lambda, \cdot) \rangle_{\mathbb{R}} \frac{\mathrm{d}\lambda}{2\sqrt{\lambda - q_{\sigma}}}$$

Furthermore, $P_{(-\rho,\rho)}f \to f$ in $L^2(\mathbb{R})$ as $\rho \to \infty$.

Proof. For $\rho > ||q||_{L^{\infty}(\mathbb{R})}$, the interval $(-\rho, \rho)$ contains the whole point spectrum, and hence the sum over the proper eigenfunctions contains all of them. Since the integrals on the right hand side are dominated by the Lebesgue measure $d\lambda$, we have $P_{(-\rho,\rho)} = P_{[-\rho,\rho]}$, and so the first assertion follows by Lemma 2.4.10.

The $L^2(\mathbb{R})$ convergence now follows from general properties of the projection valued measure, since $P_{(-\rho,\rho)}$ converges to I in the strong operator topology (see [67, Remark after Theorem VIII.5]).

The previous corollary will be the germ statement that allows us to derive all required properties of the generalised Fourier transform. Let us start by showing its unitarity.

2.4.12 Lemma. Let $f, g \in C_0^{\infty}(\mathbb{R})$, and denote by $\tilde{f}, \tilde{g} : \Lambda \to \mathbb{C}$ the transforms of f and g, i.e.

$$\tilde{f}(m) = \mathcal{F}_A f(m) = \int_{\mathbb{R}} f(x) \overline{\Psi}(m, x) \, \mathrm{d}x,$$

and similarly $\tilde{g} = \mathcal{F}_A g$. Then we have Parseval's equality

$$\langle f,g\rangle_{\mathbb{R}} = \langle f,\tilde{g}\rangle_{\Lambda}$$

and \mathcal{F}_A can be extended to a unitary isomorphism $\mathcal{F}_A : L^2(\mathbb{R}) \to \mathcal{R}(\mathcal{F}_A)$, where the range $\mathcal{R}(\mathcal{F}_A)$ is understood as a subspace of the Hilbert space $L^2(\Lambda, d\mu)$.

Proof. Let $f \in C_0^{\infty}(\mathbb{R})$, and let f(m) be defined by the integral above: note that the integral is (point-wise) well-defined, since $\langle f, \Psi^p(\lambda_n, \cdot) \rangle_{\mathbb{R}}$ is clearly finite, and that

$$|\tilde{f}^{\sigma}(\lambda)| = \left| \int_{\mathbb{R}} f(x) \overline{\Psi}^{\sigma}(\lambda, x) \, \mathrm{d}x \right| \le \|\Psi^{\sigma}(\lambda, \cdot)\|_{L^{\infty}(\mathbb{R})} \|f\|_{L^{1}(\mathbb{R})} \le C_{1} \|f\|_{L^{1}(\mathbb{R})},$$

by Hölders inequality and since $|\Psi^{\sigma}(x,\lambda)| < C_1$ by Lemma 2.4.8 for any $\sigma \in \{\pm\}, \lambda \in (q_{\sigma}, \infty)$. Now let $f_{\rho} := P_{(-\rho,\rho)}f$, that is

$$f_{\rho}(x) = \sum_{n=1}^{N} \Psi^{p}(\lambda_{n}, x) \tilde{f}^{p}(\lambda_{n}) + \frac{1}{2\pi} \sum_{\sigma \in \{\pm\}} \int_{q_{\sigma}}^{\rho} \Psi^{\sigma}(\lambda, x) \tilde{f}^{\sigma}(\lambda) \frac{\mathrm{d}\lambda}{2\sqrt{\lambda - q_{\sigma}}}$$

Multiplying the last equation by $\overline{f}(x)$, and integrating over \mathbb{R} we obtain

$$\begin{split} \langle f_{\rho}, f \rangle_{\mathbb{R}} &= \sum_{n=1}^{N} \int_{\mathbb{R}} \overline{f}(x) \Psi^{p}(\lambda_{n}, x) \, \mathrm{d}x \ \tilde{f}^{p}(\lambda_{n}) \\ &+ \frac{1}{2\pi} \sum_{\sigma \in \{\pm\}} \int_{q_{\sigma}}^{\rho} \int_{\mathbb{R}} \overline{f}(x) \Psi^{\sigma}(\lambda, x) \, \mathrm{d}x \ \tilde{f}^{\sigma}(\lambda) \frac{\mathrm{d}\lambda}{2\sqrt{\lambda - q_{\sigma}}}, \end{split}$$

where we used that the integrand

$$\overline{f}(x)\Psi_{\sigma}(\lambda,x)\widetilde{f}^{\sigma}(\lambda)\frac{1}{2\sqrt{\lambda-q_{\sigma}}}$$

is absolutely integrable on $\mathbb{R} \times \{m \in \Lambda : \hat{\lambda}(m) < \rho\}$, and hence we were able to apply Fubini's theorem. We obtain

$$\langle f_{\rho}, f \rangle_{\mathbb{R}} = \sum_{n=1}^{N} |\tilde{f}^{p}(\lambda_{n})|^{2} + \frac{1}{2\pi} \sum_{\sigma \in \{\pm\}} \int_{q_{\sigma}}^{\rho} |\tilde{f}^{\sigma}(\lambda)|^{2} \frac{\mathrm{d}\lambda}{2\sqrt{\lambda - q_{\sigma}}}.$$

Taking the limit $\rho \to \infty$ we obtain by Corollary 2.4.11 that

$$\langle f, f \rangle_{\mathbb{R}} = \int_{\Lambda} |\tilde{f}(m)|^2 \,\mathrm{d}\mu(m).$$

Consequently the map $\mathcal{F}_A : L^2(\mathbb{R}) \to L^2(\Lambda, \mu)$ is an isometry. Similarly, one can prove in the same fashion by considering $\langle g_{\rho}, f \rangle_{\mathbb{R}}$ with $g_{\rho} = P_{(-\rho,\rho)}g$ that

$$\langle g, f \rangle_{\mathbb{R}} = \langle \tilde{g}, f \rangle_{\Lambda}.$$

This finishes the proof.

2.4.13 Lemma. Let $f \in C_0^{\infty}(\mathbb{R})$ and set $\tilde{f} := \mathcal{F}_A f$. Then

$$\mathcal{F}_A(Af)(m) = \hat{\lambda}(m)\tilde{f}(m).$$

Proof. For $f \in C_0^{\infty}(\mathbb{R})$, we deduce by integrating by parts two times (the boundary terms vanish, since f has compact support)

$$(\mathcal{F}_A A f)(m) = \int_{\mathbb{R}} A f(x) \overline{\Psi}(m, x) \, \mathrm{d}x = \int_{\mathbb{R}} (-\Delta + q(x)) f(x) \overline{\Psi}(m, x) \, \mathrm{d}x$$
$$= \int_{\mathbb{R}} f(x) (-\Delta + q(x)) \overline{\Psi}(m, x) \, \mathrm{d}x = \int_{\mathbb{R}} f(x) \hat{\lambda}(m) \overline{\Psi}(m, x) \, \mathrm{d}x$$
$$= \hat{\lambda}(m) \tilde{f}(m),$$

which finishes the proof.

2.4.14 Sweeping up. It is easy to verify the remaining statements of Theorem 2.2.12, up to one fact: we have not shown that $\mathcal{F}_A : L^2(\mathbb{R}) \to L^2(\Lambda, \mathbb{R})$ maps onto $L^2(\Lambda, d\mu)$, that is, that \mathcal{F}_A is surjective. This follows from the Weyl-Kodeira-Titchmarch theory, but remains unproved here. We cop out and refer to [27]. In any case, it is not of any importance for the remainder of this thesis: one might replace $L^2(\Lambda, d\mu)$ with the (closed) subspace $\mathcal{R}(\mathcal{F}_A)$ in Theorem 2.2.12, which is sufficient for all following arguments.

2.4.15 On distributions. For the conventional Fourier transform, the set of Schwarz test functions is well-know to allow the extension to arbitrary Schwarz distributions. Can one define a set of distributions adopted to our generalised Fourier transform \mathcal{F}_A , to allow its extension to certain distributions? This possible, but in general, not without problems, since in particular the regularity of $q \in L^{\infty}(\mathbb{R})$ creates some complications. Exactly this is done for a similar problem in Hazard's paper [39].

For more general information on distributions and transforms, we refer to Zayed's book [81], and the functional analysis books by Berezansky, Sheftel and Us [6, 7].

3 Halfspace Problems

3.1 Introduction and References

3.1.1 Introduction. In this chapter we will analyse an explicit solution formula, which will form the core of the method of halfspace matching. Most of the work from Section 3.5 onwards is preparing the central analytical theorems of Chapter 4, and consequently this and the next chapter are closely linked. The reader is invited to check Chapter 4 after reading Sections 3.1 to 3.3, to get an overview over the pupose of this chapter.

We will consider the halfspace

$$\mathbb{R}^2_+ := \{ x = (x_1, x_2) \in \mathbb{R}^2 : x_2 > 0 \}.$$

Let furthermore A and q be the operator and potential from Section 2.2, defined in Definition and Assumption 2.2.2. We study the problem

$$\begin{cases} \Delta u(x) + (\kappa - q(x_1))u(x) = 0 & \text{for } x \in \mathbb{R}^2_+, \\ u(x_1, 0) = g(x_1) & \text{for } x_1 \in \mathbb{R}, \end{cases}$$
(3.1.1)

together with appropriate regularity conditions for u and g. To make this problem well posed, we will generally assume that

$$\kappa \in \mathbb{C} \setminus \mathbb{R},$$

so that Theorem 1.4.3 applies and the problem possesses exactly one solution in $H^1(\mathbb{R}^2_+)$, provided the boundary data fulfils $g \in H^{1/2}(\mathbb{R})$. Note that the assumption for κ can be weakened for example to $\kappa \in \mathbb{C} \setminus [\inf(q), +\infty)$, but we will stay with $\kappa \in \mathbb{C} \setminus \mathbb{R}$ for simplicity.

Our first goal in Section 3.2 is to give an integral representation of the solution u with the help of the generalised Fourier transform associated with A. We will dwell in detail on this representation, and carefully consider its mapping properties.

Afterwards, we will illustrate the representation and the different terms involved in Section 3.3. At this point, we will show a limit-absorption argument, to get closer to the actual (absorption free) problem we are interested in.

Thereafter, we will shortly introduce the *Dirichlet-to-Neumann* operator in Section 3.4. It will play an important role in the numerical formulation of Chapter 5.

The remainder of this chapter will study the *Dirichlet-to-Dirichlet operators* (abbreviated as DtD operators), which will be important in Chapter 4: for $c \in \mathbb{R}$ and $\theta \in (0, \pi)$, the DtD operator D_c^{θ} is the operator which maps the Dirichlet data g to the trace of the solution u on the line $\Gamma_c^{\theta} = \{(c + \cos(\theta)z, \sin(\theta)z) : z > 0\}$. Its analysis will be rather involved, since its properties will be crucial for the analysis of halfspace matching in Chapter 4. We point to one aspect that will be critical: the mapping properties of D_c^{θ}



Figure 3.1: Sketch for the DtD operator D_c^{θ} : it maps the Dirichlet boundary data on $\mathbb{R} \times \{0\}$ to the Dirichlet trace of the solution on the boundary Γ_{θ}^{c} .

are dependent on the support of the source $g \in H^{1/2}(\mathbb{R})$; if the support of g and the boundary Γ_c^{θ} have a contact point, D_c^{θ} will not be compact. If there is some positive distance between the two sets, D_c^{θ} will be compact.

To deal with the non-compact part of D_c^{θ} , we will study an auxiliary problem: the Laplace equation on the halfspace \mathbb{R}^2_+ . This will be the concern of Section 3.5.

Finally, the works will flow together in Section 3.6, where we will prove the mapping properties of D_c^{θ} .

3.1.2 References and historical remarks. Halfspace solutions for waveguides can be found for example in [58], and are heavily employed in [10, 9]. The proof of the mapping properties of the DtD operators can be found in [74], which, however, is restricted to the case $\theta = \pi/2$ in the free space. A further paper on this topic by Fliss, Bonnet-Ben Dhia and Tonnoir is being prepared at the moment [75], dealing in more detail with the free space case.

3.1.3 Acknowledgement. This chapter contains some of the core arguments of this thesis, and the author would like to gratefully acknowledge two important hints and contributions for this work. The core step of the proof of Section 3.5 was provided by Antoine Tonnoir to the author. It was expanded and split mainly between the two Lemmata 3.5.6 and 3.5.9. Secondly, the general idea of the splitting in the proofs of the Theorems 3.6.4 and 3.6.6 is due to Sonia Fliss.

3.2 The Solution Operator \mathcal{S}_{wg}

3.2.1 Formal derivation of the solution formula. Let us formally derive an explicit solution formula for (3.1.1). Fix some boundary data $g \in C_0^{\infty}(\mathbb{R})$ and consider the (unique) solution $u \in H^1(\mathbb{R}^2_+)$ of (3.1.1). The basic idea is to rewrite the Helmholtz operator as

$$\Delta + \kappa - q = \partial_{x_2}^2 - A + \kappa,$$

and to then utilise the diagonalisation of A by \mathcal{F}_A . Here A is understood as acting on the x_1 -coordinate, that is, $A = -\partial_{x_1}^2 + q$. Hence, let us define the (partial) generalised Fourier transforms for $x_2 > 0$ and $m \in \Lambda$ by

$$\tilde{u}(m, x_2) := [\mathcal{F}_A u(\cdot, x_2)](m), \quad \tilde{g}(m) := \mathcal{F}_A g(m).$$

So we obtain by Theorem 2.2.12 that

$$\begin{aligned} \mathcal{F}_A \left[\Delta u(\cdot, x_2) - (q(\cdot) + \kappa) u(\cdot, x_2) \right](m) \\ &= \mathcal{F}_A \left[\partial_{x_2}^2 u(\cdot, x_2) - A u(\cdot, x_2) + \kappa u(\cdot, x_2) \right](m) \\ &= \partial_{x_2}^2 \tilde{u}(m, x_2) - (\hat{\lambda}(m) - \kappa) \tilde{u}(m, x_2). \end{aligned}$$

It follows that (3.1.1) is equivalent to

$$\begin{cases} \partial_{x_2}^2 \tilde{u}(m, x_2) - (\hat{\lambda}(m) - \kappa) \tilde{u}(m, x_2) = 0 & \text{for } x_2 > 0, \ m \in \Lambda, \\ \tilde{u}(m, 0) = \tilde{g}(m) & \text{for } m \in \Lambda, \end{cases}$$

together with some appropriate decay condition $u(m, x_2) \to 0$ as $x_2 \to \infty$ to ensure that $u \in H^1(\mathbb{R}^2_+)$. For any fixed $m \in \Lambda$, the latter problem is an ordinary differential equation in x_2 , and one can easily compute for $\kappa \in \mathbb{C} \setminus \mathbb{R}$ the solution

$$\tilde{u}(m, x_2) = e^{i\sqrt{\kappa - \hat{\lambda}(m)} x_2} \tilde{g}(m).$$

Let us recall the definition of the square root: for $\lambda \in \mathbb{C}$, the square root is chosen such that $\operatorname{Im}(\sqrt{\lambda}) \geq 0$ (see Definition 1.2.1). From $\kappa \in \mathbb{C} \setminus \mathbb{R}$ it follows that $\operatorname{Re}(i(\kappa - \hat{\lambda}(m))^{1/2}) < 0$ for any $m \in \Lambda$, so that $\tilde{u}(m, x_2)$ is exponentially decreasing in x_2 . Of course, a second, exponentially increasing solution exists, but since we aim to obtain $u \in H^1(\mathbb{R}^2_+)$, we must drop it. If we now apply the inverse transform, we obtain the following.

3.2.2 The solution formula. Let $\kappa \in \mathbb{C} \setminus \mathbb{R}$. Formally, for any boundary data $g \in C_0^{\infty}(\mathbb{R})$ the solution u of (3.1.1) is given by

$$u(x) = \mathcal{S}_{wg}g(x) := \int_{\Lambda} e^{i\sqrt{\kappa - \hat{\lambda}(m)} x_2} \Psi(m, x_1)\tilde{g}(m) \,\mathrm{d}\mu(m), \qquad (3.2.1)$$

where $\tilde{g} = \mathcal{F}_A g$ and Λ , Ψ denote the quantities related to the spectral family of A (see Definitions 2.2.5 and 2.2.11 as well as Theorem 2.2.12). Hereby we define the operator S_{wg} .

The remainder of this chapter will be concerned with proving a number of properties related to S_{wg} . Particularly, our first goal is to prove that the operator S_{wg} indeed gives the same solution as Theorem 1.4.3.

Note that one can make sense of (3.2.1) for $g \in L^2(\mathbb{R})$. In this case, one interprets (3.2.1) for some $x_2 > 0$ as

$$\mathcal{S}_{\mathrm{wg}}g(\cdot, x_2) = \mathcal{F}_A^{-1}\left[\exp\left(i\sqrt{\kappa - \hat{\lambda}(\cdot)} x_2\right)\mathcal{F}_A g\right].$$

The latter functions and mappings are all defined in L^2 : for $g \in L^2(\mathbb{R})$, we have $\mathcal{F}_A g \in L^2(\Lambda, d\mu)$. The exponential has modulus less or equal than 1, so that we can apply the inverse transform \mathcal{F}_A^{-1} to the product inside the brackets.

To proceed with the analysis, we need an equivalent norm for $H^s(\mathbb{R})$ for $s \in [0, 2]$, which is adapted to the transform \mathcal{F}_A .

3.2.3 Lemma. Let $s \in [0, 2]$. Then for $f : \mathbb{R} \to \mathbb{C}$ the norm

$$\|f\|_{H^s_A(\mathbb{R})} := \int_{\Lambda} (1+|\lambda(m)|)^s |\mathcal{F}_A f(m)|^2 \,\mathrm{d}\mu(m)$$

is an equivalent norm on $H^{s}(\mathbb{R})$.

Proof. Let us first show that for $f \in H^2(\mathbb{R})$, the norm $(\|f\|_{L^2(\mathbb{R})}^2 + \|Af\|_{L^2(\mathbb{R})}^2)^{1/2}$ is an equivalent norm on $H^2(\mathbb{R})$. It is well known that $(\|f\|_{L^2(\mathbb{R})} + \|\Delta f\|_{L^2(\mathbb{R})})^{1/2}$ is a norm of $H^2(\mathbb{R})$. We have by the triangle and Hölder's inequality

$$\begin{aligned} \|f\|_{\mathbb{R}}^{2} + \|Af\|_{\mathbb{R}}^{2} &= \|f\|_{\mathbb{R}}^{2} + \|(-\Delta + q)f\|_{\mathbb{R}}^{2} \\ &\leq (1 + 2\|q\|_{L^{\infty}(\mathbb{R})}^{2})\|f\|_{\mathbb{R}}^{2} + 2\|\Delta f\|_{\mathbb{R}}^{2} \\ &\leq (2 + \|q\|_{L^{\infty}(\mathbb{R})}^{2})\left(\|f\|_{\mathbb{R}}^{2} + \|\Delta f\|_{\mathbb{R}}^{2}\right). \end{aligned}$$

Recall that we denote $\|\cdot\|_{\mathbb{R}} = \|\cdot\|_{L^2(\mathbb{R})}$ (see Subsection 1.2.2). On the other hand we have

$$\begin{split} \|f\|_{\mathbb{R}}^{2} + \|\Delta f\|_{\mathbb{R}}^{2} &= \|f\|_{\mathbb{R}}^{2} + \|(-\Delta + q)f - qf\|_{\mathbb{R}}^{2} \\ &\leq (1 + 2\|q\|_{L^{\infty}(\mathbb{R})}^{2})\|f\|_{\mathbb{R}}^{2} + 2\|Af\|_{\mathbb{R}}^{2} \\ &\leq (2 + 2\|q\|_{L^{\infty}(\mathbb{R})}^{2})\left(\|f\|_{\mathbb{R}} + \|Af\|_{\mathbb{R}}^{2}\right) \end{split}$$

so that the two norms are equivalent. The remainder now follows by a standard interpolation argument, see for example [59, Theorem B.7].

3.2.4 Lemma. Let $\kappa \in \mathbb{C} \setminus \mathbb{R}$. Then there exist constants $c_1, c_2 > 0$ such that

$$c_1 \leq \frac{|\lambda - \kappa|}{1 + |\lambda|} \leq c_2 \quad \text{for all } \lambda \in \mathbb{R}.$$

Proof. Let $\kappa = \kappa_r + i\kappa_i$ with $\kappa_r \in \mathbb{R}, \kappa_i \in \mathbb{R} \setminus \{0\}$. Then

$$|\kappa - \lambda| = \sqrt{(\kappa_r - \lambda)^2 + \kappa_i^2}.$$

The latter term is bounded from below by $|\kappa_i|$, and one easily sees that

$$\frac{|\lambda-\kappa|}{1+|\lambda|} \longrightarrow 1 \quad \text{as} \quad \lambda \to \pm \infty,$$

so that the boundedness from above and below follows immediately.

We will now show that the formal solution formula yields the correct solution.

3.2.5 Lemma. Let $\kappa \in \mathbb{C} \setminus \mathbb{R}$, and let S_{wg} be defined by (3.2.1). For given $g \in C_0^{\infty}(\mathbb{R})$, the function $u = S_{wg}g \in H^2(\mathbb{R}^2_+)$ is the unique variational solution given by Theorem 1.4.3. Accordingly, we can continuously extend $S_{wg} : H^{1/2}(\mathbb{R}) \to H^1(\mathbb{R}^2_+)$.

Proof. For $f \in H^2(\mathbb{R}^2_+)$ we have the following equivalent norm on $H^2(\mathbb{R}^2_+)$

$$\|f\|_{H^2_A(\mathbb{R}^2_+)}^2 = \int_{0}^{\infty} \int_{\Lambda} (1+|\hat{\lambda}(m)|)^2 |\tilde{f}(m,x_2)|^2 + |\partial_{x_2}^2 \tilde{f}(m,x_2)|^2 \,\mathrm{d}\mu(m) \,\mathrm{d}x_2, \qquad (3.2.2)$$

3.2. THE SOLUTION OPERATOR S_{WG}

where $f(m, x_2) = [\mathcal{F}_A f(\cdot, x_2)](m)$ denotes the partial Fourier transform of f. This equivalence can be seen as follows. The norm

$$||f||_{H^2(\mathbb{R}^2_+)}^2 \cong \int_0^\infty \int_{\mathbb{R}} |f(x)|^2 + |\partial_{x_1}^2 f(x)|^2 + |\partial_{x_2}^2 f(x)|^2 \, \mathrm{d}x_1 \, \mathrm{d}x_2,$$

is easily seen to be an equivalent $H^2(\mathbb{R}^2_+)$ norm¹, where we denoted the equivalence in the sense of norms by " \cong ". We note that

$$\begin{split} \|f\|_{H^{2}(\mathbb{R}^{2}_{+})}^{2} &\cong \int_{0}^{\infty} \int_{\mathbb{R}} |f(x)|^{2} + |\partial_{x_{1}}^{2}f(x)|^{2} + |\partial_{x_{2}}^{2}f(x)|^{2} \,\mathrm{d}x_{1} \,\mathrm{d}x_{2} \\ &= \int_{0}^{\infty} \|f(\cdot, x_{2})\|_{\mathbb{R}}^{2} + \|\partial_{x_{1}}^{2}f(\cdot, x_{2})\|_{\mathbb{R}}^{2} + \|\partial_{x_{2}}^{2}f(\cdot, x_{2})\|_{\mathbb{R}}^{2} \,\mathrm{d}x_{2} \\ &\cong \int_{0}^{\infty} \|f(\cdot, x_{2})\|_{H^{2}(\mathbb{R})}^{2} + \|\partial_{x_{2}}^{2}f(\cdot, x_{2})\|_{\mathbb{R}}^{2} \,\mathrm{d}x_{2} \\ &\cong \int_{0}^{\infty} \|f(\cdot, x_{2})\|_{H^{2}_{A}(\mathbb{R})}^{2} + \|\partial_{x_{2}}^{2}f(\cdot, x_{2})\|_{\mathbb{R}}^{2} \,\mathrm{d}x_{2}, \end{split}$$

where we used Lemma 3.2.3 for the last equivalence. Rewriting the $H^2_A(\mathbb{R})$ -norm by its definition in Lemma 3.2.3 and writing the $L^2(\mathbb{R})$ -norm with the help of Parseval's relation (2.2.4), we obtain that the last line of the previous equation is in fact (3.2.2), and thus $\|\cdot\|_{H^2_4(\mathbb{R}^2_+)}$ is an equivalent norm on $H^2(\mathbb{R}^2_+)$.

Let $\mathcal{S}_2: H^{1/2}(\mathbb{R}) \to H^1(\mathbb{R}^2_+)$ be the (continuous) operator, which maps $g \in H^{1/2}(\mathbb{R})$ to the (unique) solution $v = S_2 g \in H^1(\mathbb{R}^2_+)$ of (3.1.1). The continuity follows from Theorem 1.4.3. We will show the following: for $g \in C_0^{\infty}(\mathbb{R})$, the function $u = S_{wg}g$ coincides with $v = S_2 g$, that is, $u \in H^1(\mathbb{R})$ fulfils (3.1.1). Since $C_0^{\infty}(\mathbb{R})$ is dense in $L^2(\mathbb{R})$, this will imply that the two operators actually agree, i.e. $S_2 = S_{\text{wg}} : H^{1/2}(\mathbb{R}) \to H^1(\mathbb{R}^2_+)$. Let $g \in C_0^{\infty}(\mathbb{R})$, and let $u = S_{\text{wg}}g : \mathbb{R}^2_+ \to \mathbb{C}$. Its partial generalised Fourier transform

is given by

$$\tilde{u}(m, x_2) = e^{i\sqrt{\kappa - \hat{\lambda}(m)} x_2} \tilde{g}(m)$$

and we can compute

$$\begin{split} \|u\|_{H^2_A(\mathbb{R}^2_+)}^2 &= \int_0^\infty \int_{\Lambda} (1+|\hat{\lambda}(m)|)^2 |e^{i\sqrt{\kappa-\hat{\lambda}(m)} \, x_2} \tilde{g}(m)|^2 \\ &+ |\partial^2_{x_2} e^{i\sqrt{\kappa-\hat{\lambda}(m)} \, x_2} \tilde{g}(m)|^2 \mathrm{d}\mu(m) \, \mathrm{d}x_2 \\ &= \int_{\Lambda} \left[(1+|\hat{\lambda}(m)|)^2 + |\hat{\lambda}(m) - \kappa|^2 \right] |\tilde{g}(m)|^2 \int_0^\infty e^{-2\operatorname{Im}\sqrt{\kappa-\hat{\lambda}(m)} \, x_2} \mathrm{d}x_2 \, \mathrm{d}\mu(m) \\ &= \int_{\Lambda} \frac{(1+|\hat{\lambda}(m)|)^2 + |\hat{\kappa} - \lambda(m)|^2}{2\operatorname{Im}\sqrt{\kappa-\hat{\lambda}(m)}} |\tilde{g}(m)|^2 \, \mathrm{d}\mu(\lambda). \end{split}$$

¹This norm lies between the two $H^2(\mathbb{R})$ norms $\|f\|_{L^2(\mathbb{R}^2_+)}^2 + \|\Delta f\|_{L^2(\mathbb{R}^2_+)}^2$ and the classical $H^2(\mathbb{R}^2_+)$ norm, which involves all partial derivatives up to order two.

Since $\kappa \in \mathbb{C} \setminus \mathbb{R}$, there exists some small c > 0 such that $\operatorname{Im}(\kappa - \hat{\lambda}(m))^{1/2} > c > 0$ for any $m \in \Lambda$. This in turn implies that there exists C > 0 such that

$$\frac{(1+|\hat{\lambda}(m)|)^2+|\kappa-\hat{\lambda}(m)|^2}{2\operatorname{Im}\sqrt{\kappa-\hat{\lambda}(m)}} < C(1+|\hat{\lambda}(m)|)^2 \quad \text{for } m \in \Lambda,$$

so that it follows

$$\|u\|_{H^2_A(\mathbb{R}^2_+)}^2 \le C \int_{\Lambda} (1+|\hat{\lambda}(m)|)^2 |\tilde{g}(m)|^2 \,\mathrm{d}\mu(m) = C \|g\|_{H^2_A(\mathbb{R})}^2.$$

Consequently, our solution u is in $H^2(\mathbb{R}^2_+)$. As a result, it is sufficient so show that u fulfils $(\Delta - q + \kappa)u = 0$ in the distributional sense and that u = g on $\mathbb{R} \times \{0\}$. Let $\phi \in C_0^\infty(\mathbb{R}^2_+)$, and consider

$$\begin{split} \langle (\Delta - q + \kappa) u, \phi \rangle_{\mathbb{R}^2_+} &= \langle (-A + \partial_{x_2}^2 + \kappa) u, \phi \rangle_{\mathbb{R}^2_+} \\ &= \int_0^\infty \langle (-A + \partial_{x_2}^2 + \kappa) u(\cdot, x_2), \phi(\cdot, x_2) \rangle_{\mathbb{R}} \, \mathrm{d}x_2 \\ &= \int_0^\infty \langle (-\hat{\lambda} + \partial_{x_2}^2 + \kappa) \tilde{u}(\cdot, x_2), \, \tilde{\phi}(\cdot, x_2) \rangle_{\Lambda} \, \mathrm{d}x_2, \end{split}$$

where ϕ again denotes the partial generalised Fourier transform of ϕ . Note that we used that $\partial_{x_2}^2 \mathcal{F}_A u(\cdot, x_2) = \mathcal{F}_A \partial_{x_2}^2 u(\cdot, x_2)$, as can be seen easily to follow from $u \in H^2(\mathbb{R}^2_+)$. We now have for μ -almost all $m \in \Lambda$

$$(-\hat{\lambda}(m) + \partial_{x_2}^2 + \kappa)\tilde{u}(m, x_2) = (-\hat{\lambda}(m) + \partial_{x_2}^2 + \kappa)e^{i\sqrt{\kappa - \hat{\lambda}(m)}x_2}\tilde{g}(m) = 0,$$

so that $\langle (\Delta - q + \kappa)u, \phi \rangle_{\mathbb{R}^2_+} = 0$, and hence u is a solution in the sense of Section 1.4. It remains to show that u = g on $\mathbb{R} \times \{0\}$ in the sense of the trace operator. Since $u \in H^2(\mathbb{R}^2_+)$, u is continuous by the Sobolev-embedding and we have

$$u(x_1,0) = \int_{\Lambda} e^{i\sqrt{\kappa - \hat{\lambda}(m)} \, 0} \Psi(m,x_1) \tilde{g}(m) \, \mathrm{d}\mu(\lambda) = g(x_1)$$

by the inversion formula of Theorem 2.2.12. Hence, u is the unique solution of Theorem 1.4.3, and thus $S_{\text{wg}}g = S_2g$ for any $g \in C_0^{\infty}(\mathbb{R})$. In other words, S_{wg} agrees with S_2 on a dense subset of $H^{1/2}(\mathbb{R})$, and it can be continuously extended to obtain S_2 , finishing the proof.

3.2.6 Remark. With the help of the representation formula (3.2.1), the operator S_{wg} can be shown to be extendable to different spaces. Let us give a few examples.

(a) Firstly, one can show the continuity of S_{wg} without resorting to the Lax–Milgram theorem: utilising that Lemma 3.2.3 actually holds for any $s \in [0, 2]$, it is a rather straightforward calculation with the help of Lemma 3.2.4 to show that $S_{wg}: H^{1/2}(\mathbb{R}^2_+) \to H^1(\mathbb{R}^2_+)$ and $S_{wg}: H^{3/2}(\mathbb{R}) \to H^2(\mathbb{R}^2_+)$ are continuous.

3.3. THE MODAL REPRESENTATION

(b) It is furthermore possible to extend S_{wg} to spaces of weaker regularity: one can define the weighted Sobolev space

$$H^{1}(\mathbb{R}^{2}_{+}, x_{2} \, \mathrm{d}x) := \left\{ f \in \mathcal{D}'(\mathbb{R}^{2}_{+}) : \int_{\mathbb{R}^{2}_{+}} x_{2} \left[|f|^{2} + |\nabla f|^{2} \right] \, \mathrm{d}x < \infty \right\}.$$

It is then again rather elementary to show that

$$\mathcal{S}_{wg}: L^2(\mathbb{R}) \to H^1(\mathbb{R}^2_+, x_2 \mathrm{d}x)$$

is continuous using the same ingredients and norm representations as in the previous proof. The importance of the space $H^1(\mathbb{R}^2_+, x_2 \, dx)$ lies in the fact that it allows less regularity close to the boundary of \mathbb{R}^2_+ , so that "more general" traces are allowed. However, there is no proper trace-operator from $H^1(\mathbb{R}^2_+, x_2 \, dx)$ into $L^2(\mathbb{R})$, so that the existence of a trace becomes a more delicate issue. Here, it can be understood in the sense that $u(\cdot, x_2)$ converges to g in $L^2(\mathbb{R})$ as $x_2 \to 0$.

For some fixed $g \in L^2(\mathbb{R})$ the solution $u = S_{wg}g$ will generally not belong to $H^1(\mathbb{R}^2_+)$, and hence it will not be necessarily a weak solution of the Helmholtz equation $\Delta u + (\kappa - q)u = 0$ in \mathbb{R}^2_+ . However, it can be shown that it is still a distributional solution, again using the same ingredients as in the proof shown above. For more details on those matters, we refer to [56, 55, 12, 53].

We will not use this framework with $L^2(\mathbb{R})$ boundary data any further, but leave it here as a side note.

3.3 The Modal Representation

Our next goal is to give a physical illustration of the expansion we just obtained. In this section we pass to a "proper" scattering problem, that is, to a real $\kappa_0 \in \mathbb{R}$. Let $\kappa = \kappa_0 + i\epsilon$ with $\kappa_0 \in \mathbb{R}$ and $\epsilon > 0$, and let us consider again problem (3.1.1), namely

$$\begin{cases} \Delta u_{\epsilon}(x) + (\kappa_0 + i\epsilon - q(x_1))u_{\epsilon}(x) = 0 & \text{for } x \in \mathbb{R}^2_+, \\ u_{\epsilon}(x_1, 0) = g(x_1) & \text{for } x_1 \in \mathbb{R}. \end{cases}$$
(3.3.1)

Our goal is to consider the behaviour of the solution as $\epsilon \to 0$. By Section 3.2.2 and Lemma 3.2.5, the solution is given by

$$u_{\epsilon}(x) = \mathcal{S}^{\epsilon}_{\text{wg}}g(x) := \int_{\Lambda} e^{i\sqrt{\kappa_0 + i\epsilon - \hat{\lambda}(m)} \, x_2} \Psi(m, x_1) \tilde{g}(m) \, \mathrm{d}\mu(m), \qquad (3.3.2)$$

where $\tilde{g} = \mathcal{F}_A g$. Consider the kernel $e^{i\sqrt{\kappa_0 + i\epsilon - \hat{\lambda}(m)} x_2} \Psi(m, x_1)$ as $\epsilon \to 0+$; it converges pointwise by our choice of the square root. However, its limit is not decreasing any more, implying that the solution u_{ϵ} will not converge in $H^1(\mathbb{R}^2_+)$. In which sense can one now define a solution to the limit problem? There are several available options, and we will show that u_{ϵ} converges in a weighted Sobolev space. **3.3.1 Lemma.** Let \mathcal{S}^0_{wg} be defined by (3.3.2) with $\kappa = \kappa_0 + i0 \in \mathbb{R}$, where $\kappa_0 \in \mathbb{R}$. Then for any s > 1, the operator $\mathcal{S}^0_{wg} : H^{1/2}(\mathbb{R}) \to H^1(\mathbb{R}^2_+, (1+x_2)^{-s} dx)$ is continuous, and for any $g \in H^{1/2}(\mathbb{R})$

$$\mathcal{S}_{wg}^{\epsilon}g \to \mathcal{S}_{wg}^{0}g \qquad \text{as } \epsilon \to 0, \ \epsilon > 0,$$

with convergence in $H^1\left(\mathbb{R}^2_+, (1+x_2)^{-s} \mathrm{d}x\right)$.

Proof. Let $g \in H^{1/2}(\mathbb{R})$. By Parseval's relation (2.2.4) we have

$$\| (\mathcal{S}_{wg}^{\epsilon} - \mathcal{S}_{wg}^{0})g \|_{L^{2}(\mathbb{R}^{2}_{+}, (1+x_{2})^{-s} dx)}^{2} = \int_{0}^{\infty} \int_{\Lambda} \frac{1}{(1+x_{2})^{s}} \left| e^{i\sqrt{\kappa_{0} + i\epsilon - \hat{\lambda}(m)} x_{2}} - e^{i\sqrt{\kappa_{0} - \hat{\lambda}(m)} x_{2}} \right|^{2} |\tilde{g}(m)|^{2} d\mu(m) dx_{2}$$

The integrand is bounded by

$$2\frac{1}{(1+x_2)^s}|\tilde{g}(m)|^2,$$

since the modulus of both exponential terms is less or equal one. This term is in $L^1((0,\infty) \times \Lambda, dx_2 \times d\mu)$ and accordingly, one can apply the dominated convergence theorem to exchange limit and integral to obtain

$$\lim_{\epsilon \to 0} \| (\mathcal{S}_{\rm wg}^{\epsilon} - \mathcal{S}_{\rm wg}^{0}) g \|_{L^{2}(\mathbb{R}^{2}_{+}, (1+x_{2})^{-s} \, \mathrm{d}x)}^{2} = 0,$$

and hence $\mathcal{S}_{wg}^{\epsilon}g \to \mathcal{S}_{wg}^{0}g$ in $L^{2}(\mathbb{R}^{2}_{+},(1+x_{2})^{-s}\mathrm{d}x)$. To show that convergence is actually in $H^{1}(\mathbb{R}^{2}_{+},(1+x_{2})^{-s}\mathrm{d}x)$, one repeats the same procedure for the derivatives, exploiting the norms of Lemma 3.2.3.

For the mapping properties of \mathcal{S}^0_{wg} , one notices for example for $g \in H^{1/2}(\mathbb{R})$ that

$$\begin{split} \|\mathcal{S}_{\text{wg}}^{0}g\|_{L^{2}\left(\mathbb{R}^{2}_{+},(1+x_{2})^{-s}\,\mathrm{d}x\right)}^{2} \\ &= \int_{0}^{\infty} \int_{\Lambda} \frac{1}{(1+x_{2})^{s}} \left| e^{i\sqrt{\kappa_{0}-\hat{\lambda}(m)}\,x_{2}} \right|^{2} |\tilde{g}(m)|^{2}\mathrm{d}\mu(m)\,\mathrm{d}x_{2} \\ &\leq \int_{0}^{\infty} \int_{\Lambda} \frac{1}{(1+x_{2})^{s}} |\tilde{g}(m)|^{2}\,\mathrm{d}\mu(m)\,\mathrm{d}x_{2} \\ &= \int_{0}^{\infty} \frac{1}{(1+x_{2})^{s}}\,\mathrm{d}x_{2} \int_{\Lambda} |\tilde{g}(m)|^{2}\mathrm{d}\mu(m) \\ &\leq C \|g\|_{L^{2}(\mathbb{R})}^{2}. \end{split}$$

And similarly for the norms of the derivatives (again, take note of Lemma 3.2.3).

We have shown here that S_{wg}^{ϵ} converges in the strong operator topology of operators $H^{1/2}(\mathbb{R}) \to H^1(\mathbb{R}^2_+, (1+x_2)^{-s} dx)$. In a more classical framework, one usually shows the convergence of S_{wg}^{ϵ} in the norm topology of (different) weighted spaces. We do not pursue this here since it is non-essential for the remainder of the thesis (for classical scattering, see for example [1], and for the waveguide case [76]).

3.3.2 The modes of the open waveguide. Let us dwell a bit on the previous lemma and explain what we have shown: given any boundary $g \in H^{1/2}(\mathbb{R})$, the *outgoing* solution of (3.3.1) is given by

$$u_0(x) = \int_{\Lambda} e^{i\sqrt{\kappa_0 - \hat{\lambda}(m)} x_2} \Psi(m, x_1) \tilde{g}(m) \,\mathrm{d}\mu(m).$$

Note that we have used the notion of outgoing in the sense of the limit-absorption principle: that is, a solution u to (3.3.1) with $\epsilon = 0$ is called outgoing, if it is the limit of u_{ϵ} as $\epsilon > 0$ converges to 0.

Let us have closer look at the representation we have just obtained for this limitabsorption solution. If $\kappa_0 > q_m$ and $\sigma_p(A) \neq \emptyset$ (recall the notation of Definition and Assumption 2.2.2), this solution will not decay any more, but will contain terms constant in modulus as $x_2 \to +\infty$ inside the finite sum over the eigenfunctions. Let us denote

$$U(m;x) := e^{i\sqrt{\kappa_0 - \hat{\lambda}(m)} x_2} \Psi(m, x_1),$$

where $m \in \Lambda$. We call $U(m; \cdot)$ a *(generalised) mode* of the waveguide. Note that any mode fulfils

$$\Delta_x U(m; x) + (\kappa_0 - q(x_1))U(m; x) = 0.$$

As before, we denote $U = (U^p, U^+, U^-)$ to indicate the part of U living on the different parts of Λ . To illustrate the different types of modes, we will now assume that

$$\kappa_0 > q_+ > q_-,$$

which is a non-essential assumption, since the other cases can be discussed in the same fashion. Also note Figure 3.2, where we have plotted a few of these modes.

(a) Guided modes. For $n \in \{1, ..., N\}$, we can define the propagation constant $k_2 = \sqrt{\kappa_0 - \lambda_n} > 0$, to rewrite

$$U^p(\lambda_n; x) = e^{ik_2x_2}\Psi^p(\lambda_n, x_1).$$

This solution of the Helmholtz equation is usually called a *guided mode*: it propagates along the x_2 -axis without decay, while it is localized to the waveguide core $\{x \in \mathbb{R}^2 : a < x_1 < b\}$ in the sense that it decays exponentially as $x_1 \to \pm \infty$. In a sense, guided modes represent energy *trapped* inside the waveguide.

(b) Totally reflected, propagative modes. Let us now consider $m = (-, \lambda)$, where $\lambda \in (q_-, q_+)$, so that

$$U^{-}(\lambda; x) = e^{i\sqrt{\kappa_0 - \lambda} x_2} \Psi^{-}(\lambda, x_1)$$

decays exponentially as $x_1 \to +\infty$, but does not decay as $x_1 \to -\infty$. Defining $k_2^- := \sqrt{\kappa_0 - \lambda}$, and $k_1^- = \sqrt{\lambda - q_-}$, one obtains with the help of the Definition 2.2.11 that we have the following representation for U_{λ}^- outside of the waveguide core $\{x \in \mathbb{R}^2 : a < x_1 < b\}$:

$$U^{-}(\lambda; x) = \begin{cases} e^{i(k_{1}^{-}x_{1}+k_{2}^{-}x_{2})} + r_{\lambda}^{-}e^{i(-k_{1}^{-}x_{1}+k_{2}^{-}x_{2})} & \text{for } x_{1} < a, \\ t_{\lambda}^{-}e^{ik_{2}x_{2}}e^{-\sqrt{q_{+}-\lambda}x_{1}} & \text{for } x_{1} > b, \end{cases}$$

where r_{λ}^{-} , $t_{\lambda}^{-} \in \mathbb{C}$ are some coefficients. The last representation now allows a neat physical interpretation: for $x_1 < a$, the solution consists of two plane waves. One that is incident to the waveguide, and a reflected wave. It is a total reflection, since on the other side of the waveguide, U_{λ}^{-} decays exponentially.

(c) Partially reflected, propagative modes. Let $m = (-, \lambda)$ with $\lambda \in (q_+, \kappa_0)$. Let us define the two wave vectors for the right and left hand side by $k_2^- = k_2^+ = \sqrt{\kappa_0 - \lambda}$, and set $k_1^+ = \sqrt{\lambda - q_+}$ and $k_1^- = \sqrt{\lambda - q_-}$. In this case, a part of the incident wave is transmitted, and one obtains a plane wave on the right hand side of the waveguide, which reads as follows

$$U^{-}(\lambda, x) = \begin{cases} e^{i(k_{1}^{-}x_{1} + k_{2}^{-}x_{2})} + r_{\lambda}^{-}e^{i(-k_{1}^{-}x_{1} + k_{2}^{-}x_{2})} & \text{for } x_{1} < a_{\lambda} \\ t_{\lambda}^{-}e^{i(k_{1}^{+}x_{1} + k_{2}^{+}x_{2})} & \text{for } x_{1} > b. \end{cases}$$

The same discussion can be done for a plane wave incident from the right, which corresponds to $m = (+, \lambda)$ with $\lambda \in (q_+, \kappa_0)$.

(d) Evanescent modes. For $\lambda > \kappa_0$, the exponential $e^{i\sqrt{\kappa_0-\lambda}x_2}$ becomes exponentially decreasing for $x_2 \to +\infty$. These modes oscillate rather quickly along the x_1 -axis.

3.3.3 The plane wave representation. Let us, as before in Example 2.2.13, illustrate this expansion theorem for the case of a free halfspace, where no waveguide is present: considering again q(x) = 0 and $q_+ = q_- = 0$, we obtain that \mathcal{F}_A is practically the Fourier transform (after some transformation, compare again Example 2.2.13), and the expansion of the solution can be rewritten as follows

$$u_0(x) = \frac{1}{2\pi} \int_{\mathbb{R}} e^{i\sqrt{\kappa_0 - \xi^2} x_2 + i\xi x_1} \int_{\mathbb{R}} e^{-i\xi y} g(y) \,\mathrm{d}y \,\mathrm{d}\xi.$$

This is the so-called *plane wave spectrum representation*, which forms the basis for the field of Fourier optics [36]. In this case the types of modes are somewhat reduced. Since $q_m = q_- = q_+ = 0$, the point spectrum of A is empty, and accordingly, there are no guided modes. Also the totally reflected propagating modes are not present, and the partially reflected modes are plane waves (if $|\xi|^2 < \kappa_0$) of the form

$$e^{i\sqrt{\kappa_0-\xi^2}\,x_2+i\xi x_1}.$$

from which the name of the representation stems from. The exponentially decaying evanescent modes are also present for $|\xi|^2 > \kappa_0$.

3.4 The Dirichlet to Neumann Operator

At a later point in Chapter 5, we will need the so-called *Dirichlet-to-Neumann* operator \mathcal{N} (abbreviated as DtN operator) associated with the halfspace problem (3.1.1). Let us quickly define and show some properties of it.

3.4.1 Definition. For $\kappa \in \mathbb{C}$ and $g \in C_0^{\infty}(\mathbb{R})$, let $u = S_{wg}g$ denote the solution of the halfspace problem (3.1.1). We now define $\mathcal{N}g : \mathbb{R} \to \mathbb{C}$ by

$$\mathcal{N}g(x_1) = \partial_{x_2}u(x_1, 0) \quad \text{for } x_1 \in \mathbb{R}.$$
(3.4.1)

For $\kappa \in \mathbb{R}$, we understand S_{wg} in the sense of Lemma 3.3.1.

In other words: the Dirichlet to Neumann operator maps the Dirichlet boundary data of u to its Neumann boundary data. Let us also remark that that (3.4.1) has to be



Figure 3.2: Sketches of the different types of modes, as well as (exemplary) plots of their real parts $\operatorname{Re} U(m; x)$. The grey background indicates the function q, with the straight black lines indicating the boundaries of the waveguide (i.e. the sets $\{x = a\}$ and $\{x = b\}$). The curly arrows show the direction of propagation of the plane waves, while dotted arrows indicate an exponential decay of the modes. For (c) note that there is a corresponding mode with a plane wave incident from the right side, which is not shown. Also note that the two wave-vectors in (c) must fulfil Snell's Law.

understood as an equality of functions in $L^2(\mathbb{R})$. Since $q \in L^{\infty}(\mathbb{R}^2_+)$, the solution u will generally be only in $H^2(\mathbb{R})$, so that its normal derivative is only in $H^{1/2}(\mathbb{R})$, which does not allow point evaluation, even despite the smooth Dirichlet boundary data g.

In most literature, the DtN operators are considered as operators from $H^{1/2}(\mathbb{R})$ into its dual $H^{-1/2}(\mathbb{R})$. However, since we will use an $H^1(\mathbb{R})$ framework for the traces later in Chapter 5, we will use $H^1(\mathbb{R})$ as the domain of \mathcal{N} . To implement the DtN operator, we will also need a more explicit representation, which can be given with the help of the spectral family \mathcal{F}_A .

3.4.2 Lemma. For $\kappa \in \mathbb{C}$, the DtN operator can be extended to a continuous linear operator $\mathcal{N} : H^1(\mathbb{R}) \to L^2(\mathbb{R})$. For any $g \in H^1(\mathbb{R})$, it holds that

$$\mathcal{N}g = \mathcal{F}_A^{-1}\left(i\sqrt{\kappa - \hat{\lambda}(\cdot)} \mathcal{F}_A g(\cdot)\right).$$

Proof. We will only show a formal calculation, which can be easily made rigorous if one considers weak derivatives. Let $g \in C_0^{\infty}(\mathbb{R})$. For $x \in \mathbb{R}^2_+$, we have (formally)

$$\begin{aligned} \partial_{x_2} u(x) &= \partial_{x_2} (\mathcal{S}_{\text{wg}} g(x)) \\ &= \partial_{x_2} \int_{\Lambda} e^{i\sqrt{\kappa - \hat{\lambda}(m)} \, x_2} \Psi(m, x_1) \mathcal{F}_A g(m) \, \mathrm{d}\mu(m) \\ &= \int_{\Lambda} \partial_{x_2} e^{i\sqrt{\kappa - \hat{\lambda}(m)} \, x_2} \Psi(m, x_1) \mathcal{F}_A g(m) \, \mathrm{d}\mu(m) \\ &= \int_{\Lambda} i\sqrt{\kappa - \hat{\lambda}(m)} e^{i\sqrt{\kappa - \hat{\lambda}(m)} \, x_2} \Psi(m, x_1) \mathcal{F}_A g(m) \, \mathrm{d}\mu(m) \end{aligned}$$

Setting $x_2 = 0$ yields the desired representation of \mathcal{N} . Since $\mathcal{F}_A^{-1} : L^2(\Lambda, d\mu) \to L^2(\mathbb{R})$ is unitary, we have

$$\begin{split} \|\mathcal{N}g\|_{L^{2}(R)} &= \left\|\mathcal{F}_{A}^{-1}\left(i(\kappa-\hat{\lambda})^{1/2} \mathcal{F}_{A}g(\cdot)\right)\right\|_{\mathbb{R}} \\ &= \|i(\kappa-\hat{\lambda})^{1/2} \mathcal{F}_{A}g\|_{L^{2}(\Lambda,\mathrm{d}\mu)} \\ &\leq C\|(1+|\hat{\lambda}|)^{1/2} \mathcal{F}_{A}g\|_{L^{2}(\Lambda,\mathrm{d}\mu)} \\ &= C\|g\|_{H^{1}_{A}(\mathbb{R})}, \end{split}$$

where we used the equivalent norm from Lemma 3.2.3 in the last step.

3.5 A Laplace Problem

3.5.1 Introduction and references. This section will be concerned with an auxiliary problem, which will allow us to study the behaviour of the Diricht-to-Dirichlet operators at the junction point of the source-support and the target boundary. The only statement of relevance for the remainder of this thesis will be Theorem 3.5.10, while the rest will not play a role elsewhere.

The techniques employed in this section are rather classical. Mellin transform arguments have long been used for the study of singularities at corners of the boundary, for

3.5. A LAPLACE PROBLEM

example in [53, 81, 66], where also more information on the Fourier–Laplace transform can be found (in [53, 81]).

We also point again to the acknowledgements in Subsection 3.1.3.

3.5.2 The problem of interest. Given $f \in C_0^{\infty}(\mathbb{R})$, we are going to consider the problem of finding a solution $u \in C^2(\mathbb{R}^2_+) \cap C^0(\overline{\mathbb{R}^2_+})$ to

$$\begin{cases} \Delta u(x) = 0 & \text{for } x \in \mathbb{R}^2_+, \\ u(x_1, 0) = f(x_1) & \text{for } x_1 \in \mathbb{R}, \\ u \text{ bounded} & \text{on } \mathbb{R}^2_+, \end{cases}$$
(3.5.1)

as well as certain extensions for boundary data $f \in L^2(\mathbb{R})$ or $f \in H^1(\mathbb{R})$.

3.5.3 Lemma. For any $f \in C_0^{\infty}(\mathbb{R})$, there exists exactly one solution $u \in C^2(\mathbb{R}^2_+) \cap C^0(\overline{\mathbb{R}^2_+})$ to (3.5.1).

Proof. Let us show uniqueness: let u_1, u_2 be two solutions to (3.5.1). Define $w = u_1 - u_2$, so that w fulfils the homogeneous problem, that is, (3.5.1) with f = 0. For $x_1 \in \mathbb{R}, x_2 > 0$ define $w(x_1, -x_2) := -w(x_1, x_2)$. One easily verifies that w is defined on the whole of \mathbb{R}^2 , and w as well as its first derivatives are continuous across $\mathbb{R} \times \{0\}$. This implies that $\Delta w(x) = 0$ for any $x \in \mathbb{R}^2$, and accordingly, w is a bounded harmonic function on \mathbb{R}^2 . From Liouville's Theorem (see [28, Chapter 2, Theorem 8]) we now obtain that w is constant. Since w(0, 0) = 0, this yields w = 0.

To see the existence of a solution, we note that a solution is given by

$$u(x,y) = \frac{1}{\sqrt{2\pi}} \int_{\mathbb{R}} e^{i\xi x_1} e^{-|\xi|x_2} \hat{f}(\xi) \,\mathrm{d}\xi, \qquad (3.5.2)$$

where \hat{f} denotes the classical Fourier transform of f, that is

$$\hat{f}(\xi) = \frac{1}{\sqrt{2\pi}} \int_{\mathbb{R}} e^{-i\xi x_1} f(x_1) \,\mathrm{d}x_1.$$

The proof that u is in fact a solution can be performed almost exactly as in Lemma 3.2.5: firstly, one shows that $u \in H^2(\mathbb{R} \times (0, h))$ for any h > 0, and then that $\Delta u = 0$. The trace equality is shown exactly the same way as in Lemma 3.2.5, and the boundedness of u can be seen for example by the chain of inequalities

$$|u(x)| \le C ||u(\cdot, x_2)||_{H^1(\mathbb{R})} \le C ||f||_{H^1(\mathbb{R})} \quad \text{for } x \in \mathbb{R}^2_+$$

where we employed the Sobolev embedding $H^1(\mathbb{R}) \to C^0(\mathbb{R})$ at the first inequality, and performed a simple estimate with the help of the representation formula of $u(\cdot, x_2)$ at the second inequality. To see that $u \in C^2(\mathbb{R}^2_+)$ one now employs a standard regularity theorem (see for example [59, Theorem 4.18]).

3.5.4 Lemma. Denote by $S_{\Delta}f(x) := u(x)$ the solution operator associated with (3.5.1), which maps the boundary data $f \in C_0^{\infty}(\mathbb{R})$ to the solution $u \in C^2(\mathbb{R}^2_+) \cap C^0(\overline{\mathbb{R}^2_+})$. For any h > 0, we can extend S_{Δ} to a continuous operator $S_{\Delta} : H^{1/2}(\mathbb{R}) \to H^1(\mathbb{R} \times (0, h))$.

Proof. We have to show that there exists some C > 0 such that

$$\|u\|_{H^{1}(\mathbb{R}\times(0,h))}^{2} = \|u\|_{\mathbb{R}\times(0,h)}^{2} + \|\partial_{x_{1}}u\|_{\mathbb{R}\times(0,h)}^{2} + \|\partial_{x_{2}}u\|_{\mathbb{R}\times(0,h)}^{2} \le C\|f\|_{H^{1/2}(\mathbb{R})},$$

where u is given by (3.5.2). We start by estimating the first and second term of the $H^1(\mathbb{R} \times (0, h))$ -norm. It holds that

$$\|u\|_{\mathbb{R}\times(0,h)}^{2} + \|\partial_{x_{1}}u\|_{\mathbb{R}\times(0,h)}^{2} = \int_{0}^{h} \|u(\cdot,x_{2})\|_{\mathbb{R}}^{2} + \|\partial_{x_{1}}u(\cdot,x_{2})\|_{\mathbb{R}}^{2} \,\mathrm{d}x_{2}$$
$$= \int_{0}^{h} \|u(\cdot,x_{2})\|_{H^{1}(\mathbb{R})}^{2} \,\mathrm{d}x_{2} \cong \int_{0}^{h} \int_{\mathbb{R}} (1+\xi^{2})|\hat{u}(\xi,x_{2})|^{2} \,\mathrm{d}\xi \,\mathrm{d}x_{2},$$

where " \cong " denotes the equivalence of norms, and \hat{u} denotes the partial Fourier transform of u, that is

$$\hat{u}(\xi, x_2) = \frac{1}{\sqrt{2\pi}} \int_{\mathbb{R}} e^{-i\xi x_1} u(x_1, x_2) \, \mathrm{d}x_1.$$

From the definition of u one easily obtains $\hat{u}(\xi, x_2) = e^{-|\xi|x_2} \hat{f}(\xi)$, so that

$$\begin{aligned} \|u\|_{\mathbb{R}\times(0,h)}^2 + \|\partial_{x_1}u\|_{\mathbb{R}\times(0,h)}^2 &\cong \int_0^h \int_{\mathbb{R}} (1+\xi^2)e^{-2|\xi|x_2}|\hat{f}(\xi)|^2 \,\mathrm{d}\xi \,\mathrm{d}x_2 \\ &= \int_{\mathbb{R}} \frac{(1+\xi^2)(1-e^{-2|\xi|h})}{2|\xi|}|\hat{f}(\xi)|^2 \,\mathrm{d}\xi. \end{aligned}$$

It is now elementary to prove that there exists a constant C_1 such that

$$\frac{(1+\xi^2)(1-e^{-2|\xi|h})}{2|\xi|} \le C_1(1+\xi^2)^{1/2} \quad \text{for all } \xi \in \mathbb{R}.$$

Note in particular that

$$\lim_{\xi \to 0} \frac{1 - e^{-2|\xi|h}}{2|\xi|} = h,$$

as can be easily verified by applying L'Hospital's rule. This now yields

$$\|u\|_{\mathbb{R}\times(0,h)}^{2} + \|\partial_{x_{1}}u\|_{\mathbb{R}\times(0,h)}^{2} \cong \int_{\mathbb{R}} \frac{(1+\xi^{2})(1-e^{-2|\xi|h})}{2|\xi|} |\hat{f}(\xi)|^{2} \mathrm{d}\xi$$
$$\leq C_{1} \int_{\mathbb{R}} (1+\xi^{2})^{1/2} |\hat{f}(\xi)|^{2} \mathrm{d}\xi$$
$$= C_{1} \|f\|_{H^{1/2}(\mathbb{R})}^{2}.$$

Similarly, one easily calculates

$$\begin{aligned} \|\partial_{x_2} u\|_{\mathbb{R}^2_+}^2 &\cong \int_0^h \int_{\mathbb{R}} \left| \partial_{x_2} \left(e^{-|\xi| x_2} \hat{u}(\xi, x_2) \right) \right|^2 \, \mathrm{d}\xi \, \mathrm{d}x_2 \\ &\leq \int_{\mathbb{R}} \frac{\xi^2 (1 - e^{-2\xi h})}{2|\xi|} |\hat{f}(\xi)| \, \mathrm{d}\xi \\ &\leq C_2 \|f\|_{H^{1/2}(\mathbb{R})}^2, \end{aligned}$$

finishing the proof.

We can now define the Dirichlet-to-Dirichlet operator of (3.5.1). Since our problem is invariant with respect to translations along the x_1 -axis, we only consider the target boundary Γ_0^{θ} .

3.5.5 Definition. Let $\theta \in (0, \pi)$, and let $S_{\Delta} : H^{1/2}(\mathbb{R}) \to H^1_{\text{loc}}(\mathbb{R}^2_+)$ denote the solution operator associated with (3.5.1). For $f \in C_0^{\infty}(0, \infty)$, define the operator D_{Δ}^{θ} by

$$D_{\Delta}^{\theta} f(z) = (\mathcal{S}_{\Delta} f)(z\cos(\theta), z\sin(\theta))$$

where f is extended by zero to be in $C_0^{\infty}(\mathbb{R})$.

Let us reiterate the meaning of D_{Δ}^{θ} : for some $f \in C_0^{\infty}(0,\infty)$, it gives the trace on line $\Gamma_0^{\theta} = \{(z\cos(\theta), z\sin(\theta))^{\top} : z > 0\}$ of the solution to

$$\begin{cases} \Delta u = 0 & \text{in } \mathbb{R}^2_+, \\ u(x_1, 0) = f(x_1) & \text{for } x_1 > 0, \\ u(x_1, 0) = 0 & \text{for } x_1 < 0, \\ u \text{ bounded} & \text{on } \mathbb{R}^2_+. \end{cases}$$
(3.5.3)

We call D^{θ}_{Δ} the *Dirichlet-to-Dirichlet* operator (abbreviated as DtD operator) since it maps the Dirichlet data on $(0, \infty) \times \{0\}$ to the Dirichlet data on Γ . Note that the two sets of interest are touching each other at the origin (0, 0), as shown in Figure 3.3.



Figure 3.3: Sketch of the geometry behind the problem (3.5.3).

To obtain an explicit estimate of the norm, we transform the Laplace equation into polar coordinates, where the radius is logarithmically scaled.

3.5.6 Lemma. Let $f \in C_0^{\infty}(0, \infty)$, and let $u = u(r, \theta)$ be the solution to (3.5.3), given in polar coordinates. Define $\tilde{u} : \mathbb{R} \times (0, \pi) \to \mathbb{C}$ and $\tilde{f} : \mathbb{R} \to \mathbb{C}$ by

$$\tilde{u}(\log(r), \theta) = u(r, \theta)$$
 and $f(\log(r)) = f(r)$.

Then \tilde{u} fulfils

$$\begin{cases} \partial_{\rho}^{2} \tilde{u}(\rho,\theta) + \partial_{\theta}^{2} \tilde{u}(\rho,\theta) = 0 & \text{for } (\rho,\theta) \in \mathbb{R} \times (0,\pi), \\ \tilde{u}(\rho,0) = \tilde{f}(\rho) & \text{for } \rho \in \mathbb{R}, \\ \tilde{u}(\rho,\pi) = 0 & \text{for } \rho \in \mathbb{R}, \\ \tilde{u} \text{ bounded}, \end{cases}$$
(3.5.4)

and it holds that

$$\|g\|_{(0,\infty)}^2 = \int_{\mathbb{R}} |\tilde{g}(\rho)|^2 e^{\rho} \mathrm{d}\rho,$$

$$\|\partial_r g\|_{(0,\infty)}^2 = \int_{\mathbb{R}} |\partial_\rho \tilde{g}(\rho)|^2 e^{-\rho} \mathrm{d}\rho,$$

(3.5.5)

where $g = u(\cdot, \theta)$ and $\tilde{g} = \tilde{u}(\cdot, \theta)$, or similarly g = f and $\tilde{g} = \tilde{f}$. **Proof.** Let us recall that (see Subsection 1.2.2 to recap the notation)

$$||g||_{(0,\infty)}^2 = \int_0^\infty |g(x)|^2 \,\mathrm{d}x.$$

We have that

$$0 = \Delta_{(r,\theta)} u(r,\theta) = \left(\partial_r^2 + \frac{1}{r}\partial_r + \frac{1}{r^2}\partial_\theta^2\right) \tilde{u}(\log(r),\theta)$$
$$= \partial_\rho^2 \tilde{u}(\log(r),\theta) \frac{1}{r^2} + \frac{1}{r^2}\partial_\theta^2 \tilde{u}(\log(r),\theta).$$

Multiplying by r^2 yields the Laplace equation for \tilde{u} . The equations for the traces are rather clear. For the norm, notice that

$$\begin{split} \int_{0}^{\infty} |\partial_{r} u(r,\theta)|^{2} \, \mathrm{d}r &= \int_{0}^{\infty} |\partial_{r} [\tilde{u}(\log(r),\theta)]|^{2} \, \mathrm{d}r = \int_{0}^{\infty} |(\partial_{\rho} \tilde{u})(\log(r),\theta) \frac{1}{r}|^{2} \, \mathrm{d}r \\ &= \int_{\mathbb{R}} |\partial_{\rho} \tilde{u}(\rho,\theta)|^{2} \frac{1}{e^{2\rho}} e^{\rho} \, \mathrm{d}\rho, \end{split}$$

where we substituted $\log(r) = \rho$ in the last step. The corresponding statement for $\|u(\cdot, \theta)\|_{(0,\infty)}^2$ is obtained analogously.

We now have to solve (3.5.4) in a representation which gives access to the norms (3.5.5). The usual Fourier transform is not well suited for those spaces, so we will introduce another one.

3.5.7 Definition and Lemma. For $\beta \in \mathbb{R}$ we define the norm

$$||f||^2_{L^2(\mathbb{R},e^{2\beta\rho}\mathrm{d}\rho)} := \int_{\mathbb{R}} |f(\rho)|^2 e^{2\beta\rho} \,\mathrm{d}\rho.$$

We furthermore define the unitary transformation $\mathcal{F}_{\beta}: L^2(\mathbb{R}, e^{2\beta\rho} d\rho) \to L^2(\mathbb{R})$ by

$$\mathcal{F}_{\beta}f(\xi) = \frac{1}{\sqrt{2\pi}} \int_{\mathbb{R}} e^{-i\xi\rho + \beta\rho} f(\rho) \, \mathrm{d}\rho$$

Lastly, for f such that $f \in L^2(\mathbb{R}, e^{2\beta\rho} \mathrm{d}\rho)$ and $\partial_\rho f \in L^2(\mathbb{R}, e^{2\beta\rho} \mathrm{d}\rho)$, it holds that

$$[\mathcal{F}_{\beta}\partial_{\rho}f](\xi) = (i\xi - \beta)\mathcal{F}_{\beta}f(\xi).$$
(3.5.6)

We call \mathcal{F}_{β} the β -Fourier–Laplace transform.

66

3.5. A LAPLACE PROBLEM

Proof. The unitarity can be seen by noting that $\mathcal{F}_{\beta}f = \mathcal{F}(e^{\beta} f(\cdot))$ and exploiting the unitarity of the Fourier transform. The representation for the derivative is easily obtained by partial integration for some $f \in C_0^{\infty}(\mathbb{R})$.

It follows a technical Lemma with a rather long proof, which could be an interesting exercise in an introductory calculus course.

3.5.8 Lemma. Let $\theta \in (0, \pi)$. Then

$$\left|\frac{e^{iz\theta} - e^{i2\pi z}e^{-iz\theta}}{1 - e^{i2\pi z}}\right|^2 \le \frac{1 + \cos(\theta)}{2} \qquad \text{for } z = \pm \frac{1}{2} + i\xi, \, \xi \in \mathbb{R}.$$

Proof. ² Let $z = a + i\xi$, where $a, \xi \in \mathbb{R}$. We then obtain after a straightforward (but slightly lengthy) calculation

$$\left|\frac{e^{iz\theta} - e^{i2\pi z}e^{-iz\theta}}{1 - e^{i2\pi z}}\right|^2 = \frac{e^{-2\xi\theta} + e^{-2\xi(2\pi-\theta)} - 2e^{-2\xi\pi}\cos(2a(\pi-\theta))}{1 + e^{-4\xi\pi} - 2e^{-2\xi\pi}\cos(2\pi a)}.$$

For $a = \pm \frac{1}{2}$, we obtain $\cos(2\pi a) = -1$, and $\cos(a(2\pi - \theta)) = -\cos(\theta)$, so that

$$\left|\frac{e^{iz\theta} - e^{i2\pi z}e^{-iz\theta}}{1 - e^{i2\pi z}}\right|^2 = \frac{e^{-2\xi\theta} + e^{-2\xi(2\pi-\theta)} + 2e^{-2\xi\pi}\cos(\theta)}{1 + e^{-4\xi\pi} + 2e^{-2\xi\pi}}$$
$$= \frac{e^{2\xi(\pi-\theta)} + e^{-2\xi(\pi-\theta)} + 2\cos(\theta)}{e^{2\xi\pi} + e^{-2\xi\pi} + 2}$$
$$= \frac{\cosh(2\xi(\pi-\theta)) + \cos(\theta)}{\cosh(2\pi\xi) + 1}$$
$$=: \frac{n(\xi)}{d(\xi)},$$

where we abbreviated $n(\xi) = \cosh(2\xi(\pi - \theta)) + \cos(\theta)$ for the numerator and $d(\xi) = \cosh(2\pi\xi) + 1$ for the denominator. Note that both *n* and *d* are even functions, so that we can restrict ourselves to considering $\xi \ge 0$. We will now prove that

$$\frac{n(\xi)}{d(\xi)} \le \frac{n(0)}{d(0)} = \frac{1 + \cos(\theta)}{2} \qquad \text{for all } \xi > 0.$$
(3.5.7)

We have that

$$\frac{n(\xi)}{d(\xi)} = \frac{n(0)}{d(0)} \ \frac{1 + \frac{n(\xi) - n(0)}{n(0)}}{1 + \frac{d(\xi) - d(0)}{d(0)}},$$

so that (3.5.7) is fulfilled if and only if

$$1 + \frac{n(\xi) - n(0)}{n(0)} \le 1 + \frac{d(\xi) - d(0)}{d(0)} \quad \text{for all } \xi > 0,$$

or equivalently (since $d(\xi) - d(0) > 0$, $n(\xi) - n(0) > 0$)

$$\frac{n(\xi) - n(0)}{d(\xi) - d(0)} \le \frac{n(0)}{d(0)} \qquad \text{for all } \xi > 0.$$
(3.5.8)

 $^{^{2}}$ The Lemma was posed by the author as a "small exercise" to Uwe Zeltmann. The critical idea to use Cauchy's mean value theorem is due to him.

By Cauchy's mean value theorem, for any $\xi > 0$ there exists some $\eta_{\xi} \in (0, \xi)$ such that

$$\frac{n(\xi) - n(0)}{d(\xi) - d(0)} = \frac{n'(\eta_{\xi})}{z'(\eta_{\xi})},$$

so that (3.5.8) follows immediately if we can show that

$$\frac{n'(\eta)}{d'(\eta)} \le \frac{n(0)}{d(0)} \qquad \text{for all } \eta > 0.$$

Hence let us consider

$$f(\eta) := n(0)d'(\eta) - d(0)n'(\eta), \quad \text{for } \eta > 0.$$

We have to show that $f(\eta) \ge 0$. Explicitly calculating, we obtain

$$f(\eta) = (1 + \cos(\theta))2\pi \sinh(2\pi\eta) - 2(\pi - \theta)\sinh(2(\pi - \theta)\eta).$$

We have f(0) = 0, and furthermore, one finds that

$$f'(\eta) = (1 + \cos(\theta))4\pi^2 \cosh(2\pi\eta) - 4(\pi - \theta)^2 \cosh(2(\pi - \theta)\eta)$$

$$\geq (1 + \cos(\theta))4\pi^2 \cosh(2\pi\eta) - 4(\pi - \theta)^2 \cosh(2\pi\eta)$$

$$= 4(\pi^2(1 + \cos(\theta)) - (\pi - \theta)^2) \cosh(2\pi\eta)$$

$$\geq 0,$$

since both terms on the right hand side are strictly non-negative: this is clear for $\cosh(2\pi\eta)$, and we will consider $(\pi^2(1 + \cos(\theta)) - (\pi - \theta)^2)$ below. Hence it follows that $f'(\eta) \ge 0$ for any $\eta > 0$, and accordingly $f(\eta) \ge f(0) = 0$ for any $\eta > 0$.

Let us pick up the last bit, and consider

$$g(\theta) := (\pi^2 (1 + \cos(\theta)) - (\pi - \theta)^2) = \cos(\theta)\pi^2 + (2\pi - \theta)\theta$$

For $\theta \in (0, \pi/2)$, the cosine is positive, and hence both terms on the right hand side are positive, i.e. $g(\theta) \ge 0$. Let us show that $g(\theta)$ is monotonously decreasing on $(\pi/2, \pi)$. Note that from the convexity of the map $[\pi/2, \pi] \to \mathbb{R}, \theta \mapsto -\sin(\theta)$ one obtains

$$-\sin(\theta) \le \frac{2}{\pi}(\theta - \pi)$$
 for any $\theta \in [\pi/2, \pi],$

since the right hand side describes the secant through the points $(\pi/2, -\sin(\pi/2))$ and $(\pi, \sin(\pi))$. Hence we obtain for any $\theta \in (\pi/2, \pi)$

$$g'(\theta) = -\sin(\theta)\pi^2 + 2\theta(\pi - \theta)$$

$$\leq \pi^2 \frac{2}{\pi}(\theta - \pi) + 2\theta(\pi - \theta)$$

$$= 2(\pi - 1)(\theta - \pi)$$

$$\leq 0,$$

and consequently $g(\theta) \ge g(\pi) = 0$ for any $\theta \in (\pi/2, \pi)$, finishing the proof. The proof of the following lemma collects the main ingredients of this section.

3.5. A LAPLACE PROBLEM

3.5.9 Lemma. Let $f \in C_0^{\infty}(0, \infty)$ and let $u = S_{\Delta}f$ be the solution of (3.5.3), in polar coordinates. Then for any $\theta \in (0, \pi)$ it holds that

$$\|u(\cdot,\theta)\|_{(0,\infty)}^2 \le \frac{1+\cos(\theta)}{2} \|f\|_{(0,\infty)}^2,$$
$$\|\partial_r u(\cdot,\theta)\|_{(0,\infty)}^2 \le \frac{1+\cos(\theta)}{2} \|\partial_r f\|_{(0,\infty)}^2.$$

Proof. Let $f \in C_0^{\infty}(0,\infty)$, and let $\tilde{f} : \mathbb{R} \to \mathbb{C}$, $\tilde{u} : \mathbb{R} \times (0,\pi) \to \mathbb{C}$ be defined as in Lemma 3.5.6. Let us define the partial $\frac{1}{2}$ -Fourier–Laplace transform of \tilde{u} by

$$w(\xi,\theta) = [\mathcal{F}_{1/2}\tilde{u}(\cdot,\theta)](\xi) = \frac{1}{\sqrt{2\pi}} \int_{\mathbb{R}} e^{-i\xi\rho + \frac{1}{2}\rho} \tilde{u}(\rho,\theta) \,\mathrm{d}\rho,$$

and let $h = \mathcal{F}_{1/2}\tilde{f}$. Then we have by (3.5.5) and the unitarity of \mathcal{F}_{β} (Definition and Lemma 3.5.7) that

$$\|u(\cdot,\theta)\|_{(0,\infty)}^2 = \int_{\mathbb{R}} |\tilde{u}(\rho,\theta)|^2 e^{\rho} \,\mathrm{d}\rho = \int_{\mathbb{R}} |w(\xi,\rho)|^2 \,\mathrm{d}\xi.$$
(3.5.9)

And similarly

$$\|f\|_{(0,\infty)}^2 = \|h\|_{\mathbb{R}}^2. \tag{3.5.10}$$

Let us transform the problem for \tilde{u} by $\mathcal{F}_{1/2}$: from (3.5.4) we obtain with the help of the differentiation rule for \mathcal{F}_{β} (3.5.6) that w fulfils

$$\begin{cases} \left(i\xi - \frac{1}{2}\right)^2 w(\xi, \theta) + \partial_{\theta}^2 w(\xi, \theta) = 0 & \text{for } \xi \in \mathbb{R}, \ \theta \in (0, \pi), \\ w(\xi, 0) = h(\xi) & \text{for } \xi \in \mathbb{R}, \\ w(\xi, \pi) = 0 & \text{for } \xi \in \mathbb{R}. \end{cases}$$

One easily sees that the solution of this problem is given by (setting $z(\xi) = i\xi - \frac{1}{2}$ for abbreviation)

$$w(\xi,\theta) = h(\xi) \frac{e^{iz(\xi)\theta} - e^{i2\pi z(\xi)}e^{-iz(\xi)\theta}}{1 - e^{i2\pi z(\xi)}},$$

so that we obtain by Lemma 3.5.8

$$|w(\xi,\theta)|^{2} = |h(\xi)|^{2} \left| \frac{e^{iz(\xi)\theta} - e^{i2\pi z} e^{-iz(\xi)\theta}}{1 - e^{i2\pi z(\xi)}} \right|^{2} \le \frac{1 + \cos(\theta)}{2} |h(\xi)|^{2}.$$

Integrating ξ over \mathbb{R} we obtain

$$||w(\cdot,\theta)||_{\mathbb{R}}^2 \le \frac{1+\cos(\theta)}{2} ||h||_{\mathbb{R}}^2,$$

which by (3.5.9) and (3.5.10) is equivalent to

$$||u(\cdot,\theta)||_{(0,\infty)}^2 \le \frac{1+\cos(\theta)}{2} ||f||_{(0,\infty)}^2.$$

This finishes the proof of the bound for $||u(\cdot, \theta)||_{L^2(0,\infty)}$.

Let us prove the second bound for the $L^2(0,\infty)$ norm of $\partial_r u(\cdot,\theta)$. We begin by defining the (partial) $-\frac{1}{2}$ -Fourier–Laplace transform by

$$v(\xi,\theta) = [\mathcal{F}_{-1/2}\tilde{u}(\cdot,\theta)](\xi) = \frac{1}{\sqrt{2\pi}} \int_{\mathbb{R}} e^{-i\xi\rho - \frac{1}{2}\rho} \tilde{u}(\rho,\theta) \,\mathrm{d}\rho,$$

and similarly, $k = \mathcal{F}_{-1/2}f$. Then we obtain, as before by (3.5.5) and the unitarity of $\mathcal{F}_{-1/2}$, that

$$\begin{aligned} \|\partial_r f\|_{(0,\infty)}^2 &= \int_{\mathbb{R}} |\partial_\rho \tilde{f}(\rho)|^2 e^{-\rho} \mathrm{d}\rho = \int_{\mathbb{R}} \left| \left[\mathcal{F}_{-1/2} \partial_\rho \tilde{f} \right](\xi) \right|^2 \mathrm{d}\xi \\ &= \int_{\mathbb{R}} \left| \left(i\xi + \frac{1}{2} \right) \mathcal{F}_{-1/2} \tilde{f}(\xi) \right|^2 \mathrm{d}\xi = \left\| \left(i\xi + \frac{1}{2} \right) k \right\|_{\mathbb{R}}^2, \end{aligned}$$
(3.5.11)

where we used the rule (3.5.6) to calculate the Fourier–Laplace transform of the derivative. Similarly, one obtains that

$$\|\partial_r u(\cdot,\theta)\|_{\mathbb{R}}^2 = \left\| \left(i\xi + \frac{1}{2} \right) v(\cdot,\theta) \right\|_{\mathbb{R}}^2.$$
(3.5.12)

By applying the partial $-\frac{1}{2}$ -Fourier–Laplace transform to the system (3.5.4), we obtain that v fulfils

$$\begin{cases} \left(i\xi + \frac{1}{2}\right)^2 v(\xi,\theta) + \partial_{\theta}^2 v(\xi,\theta) = 0 & \text{for } \xi \in \mathbb{R}, \ \theta \in (0,\pi), \\ v(\xi,0) = k(\xi) & \text{for } \xi \in \mathbb{R}, \\ v(\xi,\pi) = 0 & \text{for } \xi \in \mathbb{R}, \end{cases}$$

so that (setting $z(\xi) = i\xi + \frac{1}{2}$ for abbreviation)

$$v(\xi,\theta) = k(\xi) \frac{e^{iz(\xi)\theta} - e^{i2\pi z(\xi)}e^{-iz(\xi)\theta}}{1 - e^{i2\pi z(\xi)}}.$$

Lemma 3.5.8 yields again that

$$|v(\xi,\theta)|^2 \le \frac{1+\cos(\theta)}{2}|k(\xi)|^2.$$

Multiplying by $|i\xi+1/2|^2$ and integrating ξ over $\mathbb R$ yields now

$$\left\|\left(i\xi+\frac{1}{2}\right)v(\cdot,\theta)\right\|_{\mathbb{R}}^{2} \leq \frac{1+\cos(\theta)}{2}\left\|\left(i\xi+\frac{1}{2}\right)k\right\|_{\mathbb{R}}^{2}.$$

Which by (3.5.11) and (3.5.12) actually reads as

$$\|\partial_r u(\cdot,\theta)\|_{\mathbb{R}}^2 \le \frac{1+\cos(\theta)}{2} \|\partial_r f\|_{\mathbb{R}}^2,$$

which finishes the proof.

Next we give the main theorem of this section. If the reader is not familiar with the space $H_{00}^{1/2}(0,\infty)$, we refer to the remark after the proof.

3.5.10 Theorem. Let D^{θ}_{Δ} be the DtD operator of Definition 3.5.5. If X is one of the spaces $L^2(0,\infty)$, $H^{1/2}_{00}(0,\infty)$ or $H^1_0(0,\infty)$, we have that $D^{\theta}_{\Delta}: X \to X$ is continuous and

$$\|D^{\theta}_{\Delta}\|_{\mathcal{L}(X,X)} \le \sqrt{\frac{1+\cos(\theta)}{2}}.$$

Proof. The statement for $X = L^2(0, \infty)$ and $X = H_0^1(0, \infty)$ follows directly from Lemma 3.5.9 by continuous extension. For $X = H_{00}^{1/2}(0, \infty)$, note that

$$H_{00}^{1/2}(0,\infty) = \left(L^2(0,\infty), H_0^1(0,\infty)\right)_{\frac{1}{2},2},$$

(compare for example [59, Theorem B.9]), where $(X, Y)_{\theta,q}$ denotes the interpolation space between the spaces X and Y. Then it holds (see again [59, Theorem B.2]) that

$$\|D\|_{H^{1/2}_{00}(0,\infty)} \le \|D\|_{L^{2}(0,\infty)}^{1/2} \|D\|_{H^{1}_{0}(0,\infty)}^{1/2} \le \left(\sqrt{\frac{1+\cos(\theta)}{2}}\sqrt{\frac{1+\cos(\theta)}{2}}\right)^{\frac{1}{2}},$$

finishing the proof.

3.5.11 On the space $H_{00}^{1/2}(0,\infty)$. To obtain a Sobolev space which incorporates zero boundary conditions (in the sense that they can be extended by 0), one often defines for some (sufficiently regular) open set Ω

$$H_0^s(\Omega) := \mathrm{cl}_{H^s(\Omega)} C_0^\infty(\Omega),$$

where cl denotes the closure with respect to the indicated norm. This procedure yields the space of functions which can be extended by 0, whenever $s \neq m + 1/2$ for some $m \in \mathbb{N} \cup \{0\}$. In this case, $H_0^s(\Omega)$ contains precisely the functions whose normal derivatives up to order $\lfloor s - \frac{1}{2} \rfloor$ vanish on $\partial \Omega$. However, in the case of s = m + 1/2 this characterisation fails. For example, if one takes the closure of $C_0^{\infty}(\Omega)$ in $H^{1/2}(\Omega)$ one obtains the whole space, i.e. (see [59, Theorem 3.40] and Subsection 1.2.2 for more precise references)

$$H^{1/2}(\Omega) = \operatorname{cl}_{H^{1/2}(\Omega)} C_0^{\infty}(\Omega).$$

On the other hand, if one takes a function from $H^{1/2}(\Omega)$, it generally cannot be extended by zero to obtain an $H^{1/2}(\mathbb{R}^n)$ function. How can one construct a Hilbert space incorporating Dirichlet boundary conditions for this critical exponent? The answer lies in a slightly stronger norm. There are three possible ways to construct a suitable one, which all yield equivalent norms.

- (1) Firstly, one can use the construction of the previous theorem and use the fact that it must be the norm obtained by interpolation between H_0^1 and L^2 (see for example [56, 55]).
- (2) Secondly, one might extend *before* taking the closure, hence one can define

$$H_{00}^{1/2}(\Omega) := \left[\mathrm{cl}_{H^{1/2}(\mathbb{R}^n)} C_0^{\infty}(\Omega) \right]_{\Omega},$$

which yields by definition a closed subspace of $H^{1/2}(\mathbb{R}^n)$ with the desired properties. This norm is used in [59].

(3) There are also explicit characterisations with the help of a weighted L^2 norm: for a bounded, sufficiently smooth domain Ω one has the following equivalent norm on $H_{00}^{1/2}(\Omega)$

$$\left(\int_{\Omega} \frac{1}{\operatorname{dist}(x,\partial\Omega)} |f(x)|^2 \,\mathrm{d}x + \|f\|_{H^{1/2}(\Omega)}^2\right)^{\frac{1}{2}},$$

where dist $(x, \partial \Omega)$ denotes the (Hausdorff) distance between x and the boundary $\partial \Omega$ (see [34, 37, 72]).

3.6 The DtD Operators of \mathcal{S}_{wg}

In this section we will analyse the Dirichlet to Dirichlet operators of S_{wg} , which we will be employed later in the study of the halfspace matching method.

3.6.1 Definition. Let $c \in \mathbb{R}$, $\theta \in (0, \pi)$. We define $D_c^{\theta} : H^{1/2}(\mathbb{R}) \to H^{1/2}(0, \infty)$ by

$$D_c^{\theta} f(z) = S_{\text{wg}} f(c + \cos(\theta)z, \sin(\theta)z).$$

Furthermore, for some (possibly unbounded) interval $I \subset \mathbb{R}$, the operator $D_c^{\theta} : H_{00}^{1/2}(I) \to H^{1/2}(0,\infty)$ is defined the same way, that is, for $f \in H_{00}^{1/2}(I)$, denote its extension by 0 by $\tilde{f} \in H^{1/2}(\mathbb{R})$, and let $D_c^{\theta} f := D_c^{\theta} \tilde{f}$.

Later on, we will study for $d \in \mathbb{R}$ the mapping properties of $D_c^{\theta} : H_{00}^{1/2}(d, +\infty) \to H^{1/2}(0, \infty)$. The parameter d describes the domain of D_c^{θ} , and the mapping properties will depend on it. The reader is advised to carefully take note of this.

We proceed by preparing with two auxiliary lemmata.

3.6.2 Lemma. Let $(X_1, d\nu_1)$ and $(X_2, d\nu_2)$ be two measure spaces, and let $K : L^2(X_1, d\nu_1) \to L^2(X_2, d\nu_2)$ be defined by

$$(Kf)(m_2) = \int_{X_1} k(m_2, m_1) f(m_1) \, \mathrm{d}\nu(m_1) \quad \text{for almost all } m_2 \in X_2,$$

with a kernel $k: X_2 \times X_1 \to \mathbb{C}$ such that

$$\int_{X_1} \int_{X_2} |k(m_2, m_1)|^2 \, \mathrm{d}\nu_2(m_2) \, \mathrm{d}\nu_1(m_1) < \infty.$$

Then $K: L^2(X_1, d\nu_1) \to L^2(X_2, d\nu_2)$ is compact. We call K a Hilbert-Schmidt operator.

Proof. This Lemma is contained in [47, Section IV.11.2].

3.6.3 Lemma. Let $(X, d\nu)$ be some measure space, and let the integral operator $K : L^2(X, d\mu) \to H^1(h, +\infty)$ be defined by

$$Kf(z) := \int_{X} k(z,m)f(m) \,\mathrm{d}\nu(m) \qquad \text{for } z > h,$$
3.6. THE DTD OPERATORS OF S_{WG}

with a kernel $k : (h, \infty) \times X \to \mathbb{C}$ such that

$$\int_X \int_h^{+\infty} |k(z,m)|^2 \, \mathrm{d}z \, \mathrm{d}\nu(m) < \infty \quad \text{and} \quad \int_X \int_h^{+\infty} |\partial_z k(z,m)|^2 \, \mathrm{d}z \, \mathrm{d}\nu(m) < \infty.$$

Then K is compact.

Proof. From the first assumption it follows easily that $K : L^2(X, d\nu) \to L^2(h, +\infty)$ is a compact operator by Lemma 3.6.2, and similarly the second bound implies that $K' : L^2(X, d\nu) \to L^2(\mathbb{R}), f \mapsto \partial_z K f$ is compact. Let us show that those two imply that K is compact as an operator into $H^1(h, \infty)$. Let $(v_n)_{n \in \mathbb{N}} \subset L^2(X, d\nu)$ be a bounded sequence. Then, since $K : L^2(X, d\nu) \to L^2(h, \infty)$ is compact, there is some subsequence $(v_n)_{n \in I}$ such that $(Kv_n)_{n \in I}$ converges strongly in $L^2(h, \infty)$. This sequence is still bounded, and hence by the compactness of K', there exists a second subsequence $(v_n)_{n \in \hat{I}}$ of $(v_n)_{n \in I}$ such that $(K'v_n)_{n \in \hat{I}}$ converges in $L^2(h, \infty)$. Since $\partial_z K v_n = K'v_n$, this implies that our second subsequence $(Kv_n)_{n \in \hat{I}}$ converges in $H^1(h, \infty)$. Hence any bounded sequence in $L^2(X, d\nu)$ has a subsequence such that its image under K converges strongly in $H^1(h, \infty)$, finishing the proof.

The following theorem considers the "nice" DtD operators: if the support has a positive distance to the target boundary, the DtD operator is compact.

3.6.4 Theorem. Let c < d. Then $D_c^{\theta} : H_{00}^{1/2}(d, \infty) \to H_{00}^{1/2}(0, \infty)$ is compact. **Proof.** Let $\phi : (0, \infty) \to [0, 1]$ be such that $\phi \in \mathbb{C}^{\infty}(0, \infty)$ and

$$\phi(z) = \begin{cases} 1 & \text{for } z \in [0, 1], \\ 0 & \text{for } z \in [2, \infty). \end{cases}$$

Define the two operators $D_1, D_2: H^{1/2}_{00}(d, \infty) \to H^{1/2}_{00}(0, \infty)$ by

$$D_1 f(z) = \phi(z) D_c^{\theta} f(z)$$
 and $D_2 f(z) = (1 - \phi(z)) D_c^{\theta} f(z).$

So that

$$D_c^{\theta} = D_1 + D_2.$$

For $h_1 < h_2$, let us furthermore introduce $\Gamma(h_1, h_2) := \{(c + \cos(\theta)z, \sin(\theta)z)^\top : z \in [h_1, h_2]\}$, so that $D_c^{\theta} f = \gamma_{\Gamma(0,\infty)} S_{\text{wg}} f$, where $\gamma_{\Gamma(0,\infty)}$ denotes the Dirichlet trace on $\Gamma(0,\infty)$.

(a) Let us show that D_1 is compact: denote $u = S_{wg}f$. We will apply the boundary regularity Theorem 4.18 from [59], with the domains $\Omega_1 \subset \Omega_2$ as sketched in Figure 3.4. Since the trace on $\Gamma_2 = \Omega_2 \cap \{(x_1, 0) : x_1 \in \mathbb{R}\}$ is equal to 0, we obtain that $u \in \hat{H}_0^2(\Omega_1) = \{\psi \in H^2(\Omega_1) : \psi|_{\Gamma_1} = 0\}$, so that $u \in \hat{H}_0^1(\Omega_1) = \{f \in H^1(\Omega_1) : f|_{\Gamma_1} = 0\}$. Here we denoted $\Gamma_1 = \Omega_1 \cap \{(x_1, 0) : x_1 \in \mathbb{R}\}$. Now let us apply the trace theorem [59, Theorem 3.37] to the (partial) Lipschitz boundary

$$B := ([c - 1, c] \times \{0\}) \cup \Gamma(0, 2),$$

which is also sketched in Figure 3.4. Since the B is $C^{0,1}$ -smooth, we obtain that $\gamma_B u \in H^{1/2}(B)$. Furthermore, since $\gamma_B u = 0$ on $[c - 1, c] \times \{0\}$, we obtain that



Figure 3.4: Sketch of the domains involved in the application of Theorem 4.18. in [59].

 $\gamma_{\Gamma(0,2)} u \in \hat{H}_{00}^{1/2}(0,2)$, where we denote by $\hat{H}_{00}^{1/2}(0,2)$ the subspace of $H^{1/2}(0,2)$ of functions which can be extended by 0 for x < 0. Hence we obtain that D_1 is factored as follows

$$D_1: H^{1/2}_{00}(d,\infty) \xrightarrow{\mathcal{S}_{\text{wg}}} \hat{H}^2_0(\Omega_1) \xrightarrow{I} \hat{H}^1_0(\Omega_1) \xrightarrow{\gamma_{\Gamma(0,2)}} \hat{H}^{1/2}_{00}(0,2) \xrightarrow{\phi} H^{1/2}_{00}(0,\infty)$$

where $\phi \cdot$ denotes the operator of multiplication by ϕ and extension by zero to obtain a function on $(0, \infty)$. All operators in the chain are bounded, and the embedding $I : \hat{H}_0^2(\Omega_1) \to \hat{H}_0^1(\Omega_1)$ is compact by the Sobolev embedding theorem [59, Theorem 3.27]. This implies that $D_1 = (\phi \cdot) \circ \gamma_{\Gamma(0,2)} \circ I \circ S_{wg}$ is compact.

(b) Let us now consider $D_2: H_{00}^{1/2}(d,\infty) \to H_{00}^{1/2}(0,\infty)$. Due to the multiplier $(1-\phi)$, which vanishes on [0,1], it is sufficient to show that D_c^{θ} is compact as an operator between $L^2(\mathbb{R}) \to H^1(1,\infty)$. Note that we significantly increased the domain of definition from $H_{00}^{1/2}(d,\infty)$ to $L^2(\mathbb{R})$, while we reduced the image space to $H^1(1,\infty)$.

Let us show that $D_c^{\theta} : L^2(\mathbb{R}) \to H^1(1,\infty)$ is compact, and define the operator $K : L^2(\Lambda, \mathrm{d}\mu) \to H^1(1,\infty)$ by

$$Kg(z) = \int_{\Lambda} k(z,m)g(m) d\mu(m)$$
 for $z > 1$,

where

$$k(z,m) = e^{i\sqrt{\kappa} - \hat{\lambda}(m)\sin(\theta)z}\Psi(m, d + \cos(\theta)z) \quad \text{for } m \in \Lambda, z > 1.$$

From the representation formula (3.2.1) we obtain

$$D_c^{\theta} = K \circ \mathcal{F}_A.$$

Since $\mathcal{F}_A : L^2(\mathbb{R}) \to L^2(\Lambda, d\mu)$ is bounded, it is sufficient to show that $K : L^2(\Lambda, d\mu) \to H^1(1, \infty)$ is compact. We will apply Lemma 3.6.3 and show that K is a Hilbert–

Schmidt operator. We have

$$\begin{split} &\int_{\Lambda} \int_{1}^{\infty} |k(z,m)|^2 \, \mathrm{d}z \, \mathrm{d}\mu(m) \\ &= \int_{\Lambda} \int_{1}^{\infty} e^{-2 \operatorname{Im}\left(\sqrt{\kappa - \hat{\lambda}(m)}\right) \sin(\theta) z} |\Psi(m, d + \cos(\theta) z)|^2 \, \mathrm{d}z \, \mathrm{d}\mu(m) \\ &\leq C_1^2 \int_{\Lambda} \int_{1}^{\infty} e^{-2 \operatorname{Im}\left(\sqrt{\kappa - \hat{\lambda}(m)}\right) \sin(\theta) z} \, \mathrm{d}z \, \mathrm{d}\mu(m) \\ &= C_1^2 \int_{\Lambda} \underbrace{\frac{\exp\left(-2 \operatorname{Im}\left(\sqrt{\kappa - \hat{\lambda}(m)}\right) \sin(\theta)\right)}{2 \operatorname{Im}\left(\sqrt{\kappa - \hat{\lambda}(m)}\right) \sin(\theta)}}_{=:\varphi(\hat{\lambda}(m))} \, \mathrm{d}\mu(m), \end{split}$$

where C_1 is the constant from Lemma 2.4.8, that is, the L^{∞} -bound for Ψ . We now have to check that this integral is bounded, and recall the definition of the spectral integral (see Definition 2.2.5)

$$\int_{\Lambda} \varphi(\hat{\lambda}(m)) d\mu(m) = \sum_{n=1}^{N} \varphi(\lambda_n) + \int_{q_+}^{\infty} \frac{\varphi(\lambda)}{2\sqrt{\lambda - q_+}} d\lambda + \int_{q_-}^{\infty} \frac{\varphi(\lambda)}{2\sqrt{\lambda - q_-}} d\lambda.$$
(3.6.1)

Note that since $\operatorname{Im}(\kappa) \neq 0$, we obtain that $\operatorname{Im}(\sqrt{\kappa - \lambda})$ does not vanish for any $\lambda \in \mathbb{R}$. Accordingly, the finite sum in (3.6.1) is well defined. Let us consider the first integral: note that φ is continuous on \mathbb{R} . It is straightforward to show that for $\lambda \in (q_+, \infty)$ we have

Im
$$\left(\sqrt{\kappa - \lambda}\right) = \sqrt{1 + |\lambda|} + \mathcal{O}(1)$$
 as $\lambda \to \infty$,

where \mathcal{O} denotes the Landau symbol. So we obtain

$$\varphi(\lambda) = \frac{\exp\left(-2\sqrt{1+|\lambda|}\sin(\theta)\right)}{2\sqrt{1+|\lambda|}\sin(\theta)}(1+\mathcal{O}(1)) \qquad \text{as } \lambda \to \infty.$$

The continuity at q_+ as well as the (almost exponential) decrease for $\lambda \to \infty$ now implies that

$$\int_{q_+}^{\infty} \frac{\varphi(\lambda)}{2\sqrt{\lambda - q_+}} \, \mathrm{d}\lambda < \infty,$$

since the only singularity of the integrand at zero is integrable. The same argument holds true for the second integral from q_{-} to ∞ , showing that the kernel k is square integrable.

It remains to show that $\partial_z k$ is also square integrable. We start by calculating

$$\begin{aligned} |\partial_z k(z,m)|^2 &= \left| \partial_z \left[e^{i\sqrt{\kappa - \hat{\lambda}(m)} \sin(\theta) z} \Psi(m,d + \cos(\theta)z) \right] \right|^2 \\ &= e^{-2\operatorname{Im}\left(\sqrt{\kappa - \hat{\lambda}(m)}\right) \sin(\theta) z} \left| i\sqrt{\kappa - \hat{\lambda}(m)} \sin(\theta) \Psi(m,d + \cos(\theta)z) + \partial_x \Psi(m,d + \cos(\theta)z) \cos(\theta) \right|^2 \\ &\leq e^{-2\operatorname{Im}\left(\sqrt{\kappa - \hat{\lambda}(m)}\right) \sin(\theta) z} 2 \left(|\kappa - \hat{\lambda}(m)| C_1^2 \sin^2(\theta) + (1 + |\hat{\lambda}(m)|) C_2^2 \cos^2(\theta) \right) \\ &\leq C \exp\left(-2\operatorname{Im}\left(\sqrt{\kappa - \hat{\lambda}(m)}\right) \sin(\theta) z \right) (1 + |\hat{\lambda}(m)|), \end{aligned}$$

as one easily obtains by Lemma 2.4.8, which gives the constants C_1 and C_2 to estimate both Ψ and $\partial_x \Psi$. This now implies that

$$\begin{split} &\int_{\Lambda} \int_{1}^{\infty} |\partial_z k(z,m)|^2 \, \mathrm{d}z \, \mathrm{d}\mu(m) \\ &\leq C \int_{\Lambda} \int_{1}^{\infty} \exp\left(-2 \operatorname{Im}\left(\sqrt{\kappa - \hat{\lambda}(m)}\right) \sin(\theta) z\right) (1 + |\hat{\lambda}(m)|) \, \mathrm{d}z \, \mathrm{d}\mu(m) \\ &= C \int_{\Lambda} \underbrace{\frac{\exp\left(-2 \operatorname{Im}\left(\sqrt{\kappa - \hat{\lambda}(m)}\right) \sin(\theta) z\right)}{2 \operatorname{Im}\left(\sqrt{\kappa - \hat{\lambda}(m)}\right) \sin(\theta)} (1 + |\hat{\lambda}(m)|) \, \mathrm{d}\mu(m). \\ &= : \tilde{\varphi}(\hat{\lambda}(m)) \end{split}$$

By the very same arguments as before, one now sees that this integral is finite: $\tilde{\varphi}$ is still continuous and $\tilde{\varphi}(\lambda)$ decreases almost exponentially as $\lambda \to \infty$. Hence it follows that

$$\int_{\Lambda} \int_{1}^{\infty} |\partial_z k(z,m)|^2 \, \mathrm{d}z \, \mathrm{d}\mu(m) \leq C \int_{\Lambda} \tilde{\varphi}(\hat{\lambda}(m)) \, \mathrm{d}\mu(m) < \infty.$$

This now shows by Lemma 3.6.3 that $K : L^2(\Lambda, d\mu) \to H^1(1, \infty)$ is compact, and hence $D_c^{\theta} = K \circ \mathcal{F}_A : H_{00}^{1/2}(d, \infty) \to H^1(1, \infty)$ is compact. This implies that $D_2 = (1 - \phi) \circ D_c^{\theta} : H_{00}^{1/2}(d, \infty) \to H_0^1(0, \infty)$ is also compact, which in turn yields by the continuity of the embedding $H_0^1(0, \infty) \to H_{00}^{1/2}(0, \infty)$ that this proof is at its end.

3.6.5 Corollary. For c > d, the operator $D_c^{\theta} : H_{00}^{1/2}(-\infty, d) \to H_{00}^{1/2}(0, \infty)$ is compact.

Proof. This follows from the previous theorem by mirroring along the axis $\{x_1 = 0\}$.

3.6. THE DTD OPERATORS OF S_{WG}

The last theorem deals with the mapping properties of the DtD operator when the two boundaries touch. In this case, compactness cannot be retained, however the noncompact part can be shown to be bounded and sufficiently small.

3.6.6 Theorem. Let $c = d \in \mathbb{R}, \ \theta \in (0, \pi)$. Then $D_c^{\theta} : H_{00}^{1/2}(d, \infty) \to H_{00}^{1/2}(0, \infty)$ can be split into two parts,

$$D_c^{\theta} = D_C + D_B,$$

where $D_C : H_{00}^{1/2}(d,\infty) \to H_{00}^{1/2}(0,\infty)$ is compact, and $D_B : H_{00}^{1/2}(d,\infty) \to H_{00}^{1/2}(0,\infty)$ is bounded with $||D_B|| < 1$.

Proof. Without loss of generality, we assume d = 0, so that $D_c^{\theta} = D_0^{\theta}$: else, one can redefine the parameters a, b, q to translate the solution operator to obtain c = 0. Let $\phi: (0,\infty) \to [0,1]$ be some $C^{\infty}(0,\infty)$ -function such that

$$\phi(z) = \begin{cases} 1 & \text{for } z \in [0, 1], \\ 0 & \text{for } z \in [2, \infty). \end{cases}$$

We now split D_0^{θ} into three parts, namely

$$D_0^{\theta} f(z) = \phi(z) D_{\Delta}^{\theta} f(z) + \phi(z) (D_0^{\theta} - D_{\Delta}^{\theta}) f(z) + (1 - \phi(z)) D_0^{\theta} f(z),$$

where D^{θ}_{Δ} denotes the DtD operator of the Laplacian as defined in Definition 3.5.5. Let us consider each of the three operators in the previous splitting. Firstly, the operator $D_B := \phi D_{\Delta}^{\theta} : H_{00}^{1/2}(0,\infty) \to H_{00}^{1/2}(0,\infty)$ is bounded with

$$\|D_B\| < \sqrt{\frac{\cos(\theta) + 1}{2}} < 1$$

by Theorem 3.5.10.

Secondly, the operator $(1 - \phi)D_0^{\theta} : H_{00}^{1/2}(0, \infty) \to H_{00}^{1/2}(0, \infty)$ can be shown to be compact as in the second part of the proof of Theorem 3.6.4, analogous to the operator D_2 therein.

Hence we only have to consider

$$\phi(D_0^{\theta} - D_{\Delta}^{\theta}) : H_{00}^{1/2}(0,\infty) \to H_{00}^{1/2}(0,\infty).$$

To see that this operator is compact, let $f \in H^{1/2}_{00}(0,\infty)$, denote $u = S_{wg}f$ and $v = S_{\Delta}f$. By Lemma 3.2.5 we have $u \in H^1(\mathbb{R}^2_+)$, while by Lemma 3.5.4 we have $v \in H^1(\mathbb{R} \times (0,h))$ for any h > 0. Furthermore, the two functions are solutions of

$$\begin{cases} \Delta u + (\kappa - q)u = 0 & \text{in } \mathbb{R}^2, \\ u(x_1, 0) = f(x_1) & \text{for } x_1 > 0, \\ u(x_1, 0) = 0 & \text{for } x_1 < 0, \end{cases}$$

and

$$\begin{cases} \Delta v = 0 & \text{in } \mathbb{R}^2, \\ v(x_1, 0) = f(x_1) & \text{for } x_1 > 0, \\ v(x_1, 0) = 0 & \text{for } x_1 < 0. \end{cases}$$

Accordingly, their difference $w = v - u \in H^1(\mathbb{R} \times (0, h))$ solves (in a variational sense)

$$\begin{cases} \Delta w = (\kappa - q)u & \text{in } \mathbb{R}^2, \\ w = 0 & \text{on } \mathbb{R} \times \{0\} \end{cases}$$

As in part (a) of the proof of Theorem 3.6.4, we can now apply the regularity Theorem 4.18 in [59] to obtain that $w \in \hat{H}^2(\Omega_1)$, and by again considering the traces on B, we obtain that $\gamma_{\Gamma(0,2)}w \in \hat{H}_{00}^{1/2}(0,2)$ (for the definitions of $\hat{H}_0^2(\Omega_1)$, Ω_1 , $\Gamma(0,1)$, B and $\hat{H}_{00}^{1/2}(0,2)$ we refer to the proof of Theorem 3.6.4). Hence, as before, we can factorise $\phi(D_0^{\theta} - D_{\Delta}^{\theta})$ as follows

$$\begin{split} \phi(D_0^{\theta} - D_{\Delta}^{\theta}) &= (\phi \cdot) \circ \gamma_{\Gamma(0,2)} \circ I \circ (\mathcal{S}_{\Delta} - \mathcal{S}_{wg}) : \\ H_{00}^{1/2}(d,\infty) \xrightarrow{\mathcal{S}_{\Delta} - \mathcal{S}_{wg}} \hat{H}_0^2(\Omega_1) \xrightarrow{I} \hat{H}_0^1(\Omega_1) \xrightarrow{\gamma_{\Gamma(0,2)}} \hat{H}_{00}^{1/2}(0,2) \xrightarrow{\phi \cdot} H_{00}^{1/2}(0,\infty), \end{split}$$

and due to the compactness of the embedding $I : \hat{H}_0^1(0,2) \to \hat{H}_{00}^{1/2}(0,2)$ the composition of all operators becomes compact.

3.6.7 Corollary. For c = d, the operator $D_c^{\theta} : H_{00}^{1/2}(-\infty, d) \to H_{00}^{1/2}(0, \infty)$ can be split into two parts.

$$D_c^{\theta} = D_C + D_B,$$

where $D_C: H_{00}^{1/2}(-\infty, d) \to H_{00}^{1/2}(0, \infty)$ is compact and $D_B: H_{00}^{1/2}(-\infty, d) \to H_{00}^{1/2}(0, \infty)$ is bounded with $||D_B|| < 1$.

Proof. This follows from the previous theorem by mirroring along the axis $\{x_1 = 0\}$.

3.6.8 Remark. The mapping properties of the DtD as given in Theorems 3.6.4 and 3.6.6 can be shown with the space $H_{00}^{1/2}$ replaced by L^2 or H_0^1 . For the L^2 case, some additional arguments are required, to adapt to the lower regularity of the boundary data (compare Remark 3.2.6), since we cannot use the classical trace theorems. The H_0^1 case can be studied with the same arguments as already presented. We have decided to use the $H_{00}^{1/2}$ framework, since it seems the canonical choice in the context of variational problems and allows us to use the classical trace and embedding theorems.

4 Halfspace Matching

4.1 Introduction and References

4.1.1 Introduction. In this chapter, we will introduce the method of *halfspace matching* for a rather simple model problem: it will allow us to reduce the 2D Helmholtz problem in the exterior of a triangle to a set of equations for the traces on certain boundaries, reducing the overall dimension by 1.

The resulting set of equations, called *compatibility equations* in the following, will be shown to be Fredholm and uniquely solvable, which in turn will yield the equivalence of the exterior Helmholtz problem with the compatibility equations.

A small warning: this Chapter requires some rather ghastly notation describing the geometry and the DtD operators. The author tried to reduce complications by drawing some pictures illustrating the geometrical notations.

This chapter mostly collects arguments from the previous chapter to show the well-posedness. The result itself will not be relevant for the following chapters, and serves mainly as a validation that this reformulation works. We note, however, that the notation introduced in Section 4.2 will be relevant for the following chapters, while the remainder can be skipped at first reading.



Figure 4.1: Sample geometry as considered in this chapter. The grey areas indicate the value of q (dark grey corresponds to low q(x)).

4.1.2 References. The basic structure

of this proof is not new, it can found in Tonnoir's thesis [74] for the free space, and a second publication is being prepared right now [75]. Compared to [74], our work is mostly analogous, with a few differences in the details, since we use a slightly different functional framework, and study the case of a triangle instead of a rectangular domain.

4.2 The Problem of Interest and Notations

4.2.1 The problem of interest. Let $\kappa \in \mathbb{C} \setminus \mathbb{R}$ and $\Omega \subset \mathbb{R}^2$ be some arbitrary triangle. Furthermore, let g be some boundary data in $H^{1/2}(\partial\Omega)$. We assume that $q : \mathbb{R}^2 \to \mathbb{R}$ fulfils Definition and Assumption 2.2.2 in each halfspace associated with Ω . Below we will define accurately what this means. Consider the problem of finding $U \in H^1(\mathbb{R}^2 \setminus \overline{\Omega})$ such that

$$\begin{cases} \Delta U + (\kappa - q)U = 0 & \text{in } \Omega, \\ U = g & \text{on } \partial \Omega. \end{cases}$$
(4.2.1)

This problem is uniquely solvable by Theorem 1.4.3.

Before going on, let us give a very rough description of the assumption for q: the "halfspaces associated with Ω ", denoted by $\Omega_0, \Omega_1, \Omega_2$, can be found by extending the sides of the triangle, and taking the halfspace on other side of the resulting line (see Figure 4.2). Note that the halfspaces overlap at infinite cones originating from the edges of the triangle.

In particular, we will allow for q to describe waveguides which extend perpendicular from the sides of the triangle to infinity. An example configuration is shown in Figure 4.1.



Figure 4.2: Sketch of the three exterior halfspaces associated with the triangle Ω .

4.2.2 Notation of geometry. Let us introduce a number of notations describing the geometry. In advance we point to Figure 4.3, where the following quantities are illustrated.

- The sign-variable $\sigma \in \{\pm\}$ is used more freely here: in equations, we identify it with (the real numbers) $+1 \cong "+"$ and $-1 \cong "-"$.
- Since the three halfspaces are ordered in a periodical sense, we generally consider the halfspace index $n \in \{0, 1, 2\}$ as an element of

$$n \in \mathbb{Z}/3\mathbb{Z},$$

so that for example $\Omega_{2+1} = \Omega_0$. We will write $n \in \mathbb{Z}/3\mathbb{Z}$ only when the corresponding relation actually uses this convention.

- Let Γ_0^0 , Γ_1^0 , Γ_2^0 denote the sides of the triangle, ordered in a counter clockwise sense.
- For $n \in \{0, 1, 2\}$, let $\eta^{(n)} \in \mathbb{R}^2$ denote the exterior unit normal of each side of Ω , and let

$$\xi^{(n)} = \begin{pmatrix} 0 & 1\\ -1 & 0 \end{pmatrix} \eta^{(n)} \in \mathbb{R}^2$$

denote (the clockwise oriented) tangent vectors to the three sides.

80

4.2. THE PROBLEM OF INTEREST AND NOTATIONS

• For $n \in \{0, 1, 2\}$, define $h_n \in \mathbb{R}$ and $c_n^- < c_n^+$ by

$$\Gamma_n^0 = \{ x \in \mathbb{R}^2 : \eta^{(n)} \cdot x = h_n, \ c_n^- \le \xi^{(n)} \cdot x \le c_n^+ \},\$$

so that in the coordinate system $(\xi^{(n)}, \eta^{(n)})$, the boundary Γ_n^0 is given by $[c_n^-, c_n^+] \times \{h_n\}$. In particular, the side of the triangle Γ_n^0 has the (signed) distance h_n from the origin.

• For $n \in \{0, 1, 2\}$, $\sigma \in \{\pm\}$, define the semi-infinite lines Γ_n^{σ} by

$$\Gamma_n^{\sigma} = \left\{ h_n \eta^{(n)} + c_n^{\sigma} \xi^{(n)} + \sigma t \xi^{(n)} : t \ge 0 \right\}.$$

Note that we have a small collision of notation here: the boundary Γ_n^{σ} should not be confused with the boundary Γ_c^{θ} from Chapter 3.

• For $n \in \{0, 1, 2\}$, define the halfspace $\Omega_n \subset \mathbb{R}^2$ by

$$\Omega_n = \{ x \in \mathbb{R}^2 : x \cdot \eta^{(n)} > h_n \}$$

which has the boundary

$$\Gamma_n := \partial \Omega_n = \Gamma_n^+ \cup \Gamma_n^0 \cup \Gamma_n^-.$$

• For $n \in \mathbb{Z}/3\mathbb{Z}$, $\sigma \in \{\pm\}$, let $\theta_n^{\sigma} \in (0, \pi)$ denote the angle between

$$\xi^{(n)}$$
 and $\sigma\xi^{(n-\sigma)}$.

• To simplify notation, we will use a number of canonical isomorphisms: for any $n \in \{0, 1, 2\}$, a function $f \in H^{1/2}(\mathbb{R})$ is identified with a function $\hat{f} \in H^{1/2}(\Gamma_n)$ by

$$f(\xi) = \hat{f}(\xi \,\xi^{(n)} + h_n \,\eta^{(n)}) \qquad \text{for } \xi \in \mathbb{R}.$$
(4.2.2)

Similarly, one obtains canonical isomorphism $H_{00}^{1/2}(\Gamma_n^+) \cong H_{00}^{1/2}(c_n^+,\infty), H_{00}^{1/2}(\Gamma_n^-) \cong H_{00}^{1/2}(-\infty,c_n^-), \ H_{00}^{1/2}(\Gamma_n^0) \cong H_{00}^{1/2}(c_n^-,c_n^+)$, and similarly for other spaces on the same boundaries. The same way we identify $f \in H^1(\Omega_n)$ with $\hat{f} \in H^1(\mathbb{R}^2_+)$ by

$$f(\xi \xi^{(n)} + (\eta + h_n) \eta^{(n)}) = \hat{f}(\xi, \eta) \quad \text{for } \xi \in \mathbb{R}, \ \eta > 0.$$
(4.2.3)

We can now make our assumption for q more accurate.

4.2.3 Definition and Assumption. We assume that on each halfspace Ω_n , the potential q only depends on the $\xi^{(n)}$ coordinate, that is, there exist three functions $q^{(n)}$: $\mathbb{R} \to \mathbb{R}, n \in \{0, 1, 2\}$, each fulfilling Definition and Assumption 2.2.2, such that

$$q(\xi \xi^{(n)} + (\eta + h_n) \eta^{(n)}) = q^{(n)}(\xi)$$
 for all $\xi \in \mathbb{R}, \ \eta > 0.$

We denote by $A_n = -\Delta + q^{(n)}$ the associated operator, and S_n the corresponding halfspace solution operators (see Section 3.2.2).

Note that these assumptions pose rather strong restrictions, which are not necessarily obvious in all cases: for example, if Γ_n^0 and Γ_{n+1}^0 are two sides which join at an obtuse angle, this implies that $q|_{\Omega_n} = \text{const} = q|_{\Omega_{n+1}}$. Compare also Subsection 5.1.3 in Chapter 5 below.

To simplify notation, we need a number of DtD operators, which take the coordinate systems of the different halfspaces into account.



Figure 4.3: Sketch and illustration of the geometrical notation.

4.2.4 Definition. Let $n \in \{0, 1, 2\}$. For $c \in \mathbb{R}$ and $\theta \in (0, \pi)$, let D_c^{θ} denote the Dirichlet to Dirichlet operator associated with Ω_n . For $f \in H^{1/2}(\Gamma_n)$, we define the operators \mathcal{D}_n^+ , \mathcal{D}_n^- by

$$\mathcal{D}_{n}^{+}f(\xi) = \begin{cases} D_{c_{n}^{-}}^{\theta_{n}^{-}}f(\xi - c_{n+1}^{+}) & \text{ for } \xi > c_{n+1}^{+}, \\ 0 & \text{ for } \xi < c_{n+1}^{+}, \end{cases}$$

and

$$\mathcal{D}_{n}^{-}f(\xi) = \begin{cases} D_{c_{n}^{+}}^{\theta_{n}^{+}}f(c_{n-1}^{-}-\xi) & \text{ for } \xi < c_{n-1}^{-}, \\ 0 & \text{ for } \xi > c_{n-1}^{-}. \end{cases}$$

What is the purpose of these operators? They take the trace on Γ_n and calculate the corresponding trace on $\Gamma_{n\pm 1}^{\pm}$ in the coordinate system associated with $\Gamma_{n\pm 1}$.

4.2.5 The compatibility equations. Consider a solution U of (4.2.1). For $n \in \mathbb{Z}/3\mathbb{Z}$, let us denote the traces on the boundaries of the halfspaces by

$$u_n := U|_{\Gamma_n}.$$

Since $U|_{\Omega_n}$ fulfils a halfspace problem of the type (3.1.1), we obtain (by exploiting the canonical identifications (4.2.2) and (4.2.3)) that

$$U|_{\Omega_n} = \mathcal{S}_n u_n.$$

In particular, by taking the trace of $U|_{\Omega_n}$ on the boundary Γ_{n+1}^+ , which is contained in Ω_n , we obtain (recall that we interpret n as an element of $\mathbb{Z}/3\mathbb{Z}$)

$$U|_{\Gamma_{n+1}^+} = \left[\mathcal{S}_n u_n\right]|_{\Gamma_{n+1}^+}.$$

The left hand side can be identified with u_{n+1} on the interval (c_{n+1}^+, ∞) . Denoting the Dirichlet-to-Dirichlet operator of the halfspace Ω_n by D_c^{θ} , the right hand side can be rewritten to obtain

$$\left[\mathcal{S}_{n}u_{n}\right]_{\Gamma_{n+1}^{+}}(\xi) = D_{c_{n}^{-}}^{\theta_{n}^{-}}u_{n}(\xi - c_{n+1}^{+}) = \mathcal{D}_{n}^{+}u_{n}(\xi) \quad \text{for } \xi > c_{n+1}^{+},$$

where we applied Definition 4.2.4. Accordingly, we obtain that

$$u_{n+1}(\xi) = \mathcal{D}_n^+ u_n(\xi) \qquad \text{for } \xi > c_{n+1}^+.$$
 (4.2.4)

Similarly, one obtains by considering the trace of $U|_{\Omega_n}$ on the boundary Γ_{n-1}^- (which is also contained in Ω_n) that

$$u_{n-1}(\xi) = \mathcal{D}_n^- u_n(\xi) \quad \text{for } \xi < c_{n-1}^-.$$
 (4.2.5)

The last two equations (4.2.4) and (4.2.5) are called *compatibility equations*, and are the key equations which have to be solved for the method of halfspace matching. Let us bring them into a (preliminary) operator equation.

4.2.6 The preliminary operator equation. Let us define for $n \in \{0, 1, 2\}$ and $\sigma \in \{-, +, 0\}$

$$u_n^{\sigma} := U|_{\Gamma_n^{\sigma}}.\tag{4.2.6}$$

We furthermore extend each function by 0 to get a function on the whole boundary Γ_n , so we can write

$$u_n = u_n^+ + u_n^0 + u_n^-.$$

This helps us to rewrite (4.2.4) as

$$u_{n+1}^{+} = \mathcal{D}_{n}^{+}(u_{n}^{+} + u_{n}^{0} + u_{n}^{-})$$

while (4.2.5) becomes

$$u_{n-1}^{-} = \mathcal{D}_{n}^{-}(u_{n}^{+} + u_{n}^{0} + u_{n}^{-}),$$

which can be rewritten together as

$$u_{n+\sigma}^{\sigma} - \mathcal{D}_n^{\sigma}(u_n^+ + u_n^-) = \mathcal{D}_n^{\sigma}u_n^0, \qquad \text{for } \sigma \in \{\pm\}, \, n \in \mathbb{Z}/3\mathbb{Z}.$$
(4.2.7)

This is (almost) the equation we will study.

Let us have a look at the original problem in the exterior (4.2.1). Since we aim to find $U \in H^1(\mathbb{R}^2 \setminus \overline{\Omega})$, given the boundary data $g \in H^{1/2}(\partial\Omega)$, we already know $u_n^0 = g|_{\Gamma_n^0}$. Hence we consider the right hand side of (4.2.7) as known, while the remainder of functions are unknowns.

Our goal is to define and study a functional setting, in which the compatibility equations (4.2.7) are well posed.

4.3 Fredholmness for the Exterior of a Triangle

Let us first define a space which contains all unknown traces, and define the *total* DtD operator which acts on this space.

4.3.1 Definition. We define the Hilbert space \mathcal{V} by

$$\mathcal{V} := H_{00}^{1/2}(-\infty, c_0^-) \times H_{00}^{1/2}(c_0^+, +\infty) \times \ldots \times H_{00}^{1/2}(c_2^+, +\infty).$$

We furthermore define the continuous operator $\mathcal{D}: \mathcal{V} \to \mathcal{V}$ as follows: for given $u = (u_0^-, u_0^+, \dots, u_2^+) \in \mathcal{V}$, let $\mathcal{D}u := v = (v_0^-, v_0^+, \dots, v_2^+) \in \mathcal{V}$ be given by

$$v_{n+\sigma}^{\sigma} = \mathcal{D}_n^{\sigma}(u_n^+ + u_n^-), \quad \text{for } \sigma \in \{\pm\}, \ n \in \mathbb{Z}/3\mathbb{Z}.$$

We can now rewrite (4.2.7) with the help of \mathcal{D} in the form

$$(I - \mathcal{D})u = F,$$

where F is some right hand side. Note also equation (4.3.1) below, where we have written this equation as an operator matrix.

This definition actually contains an assertion: that \mathcal{D} is well-defined and bounded from \mathcal{V} into \mathcal{V} . This, of course, has to be proved, which we will do in a side remark in the proof of Theorem 4.3.3.

Let us also point to a second peculiarity of this definition. If we consider the solution U of (4.2.1), and define u_n^{σ} via (4.2.6), the vector $u = (u_0^-, \ldots, u_2^+)$ will not be contained in \mathcal{V} : there is no reason for u_0^- to vanish at c_0^- to be in $H_{00}^{1/2}(-\infty, c_0^-)$. However, the system of equations for u can be recast into an equivalent system with unknowns in \mathcal{V} .

4.3.2 The operator equation in \mathcal{V} . Let U be a solution to (4.2.1) with some fixed boundary data $g \in H^{1/2}(\partial\Omega)$. We define $w_n^0 \in H^{1/2}(\mathbb{R})$ (utilising again the canonical identifications (4.2.2)) by

$$w_n^0 = g|_{\Gamma_n^0}$$
 for $n \in \{0, 1, 2\}$,

with some (arbitrary) extension to the whole of Γ such that $w_n^0 \in H^{1/2}(\Gamma)$. Let us now define w_n^+ and w_n^- by

$$w_n^+ = \begin{cases} U|_{\Gamma_n^+} - w_n^0 & \text{ on } (c_n^+, \infty), \\ 0 & \text{ else,} \end{cases} \quad \text{and} \quad w_n^- = \begin{cases} U|_{\Gamma_n^-} - w_n^0 & \text{ on } (-\infty, c_n^-), \\ 0 & \text{ else.} \end{cases}$$

One now easily sees that $w_n^- \in H_{00}^{1/2}(-\infty, c_n^-)$, $w_n^+ \in H_{00}^{1/2}(c_n^+, \infty)$ and that $U|_{\Gamma_n} = w_n^- + w_n^0 + w_n^+$. We now rewrite (4.2.4) as

$$w_{n+1}^+ + w_{n+1}^0 = \mathcal{D}_n^+ (w_n^- + w_n^0 + w_n^+)$$
 on (c_{n+1}^+, ∞) ,

which can be reformulated as

$$w_{n+1}^{+} - \mathcal{D}_{n}^{+}(w_{n}^{-} + w_{n}^{+}) = -w_{n+1}^{0} + \mathcal{D}_{n}^{+}w_{n}^{0} \qquad \text{on} \ (c_{n+1}^{+}, \infty)$$

It is now easy to verify that $-w_{n+1}^0 + \mathcal{D}_n^+ w_n^0 \in H^{1/2}_{00}(c_{n+1}^+,\infty)$. This yields that $w = (w_0^-, w_0^+, \dots, w_2^+)$ solves the equation

$$(I - \mathcal{D})w = f,$$

where the source term $f = (f_0^-, f_0^+, \dots, f_2^+) \in \mathcal{V}$ is given by

$$f_{n+\sigma}^{\sigma} = -w_{n+\sigma}^{0} + \mathcal{D}_{n}^{\sigma} w_{n}^{0} \quad \text{for } n \in \mathbb{Z}/3\mathbb{Z}, \ \sigma \in \{\pm\}.$$

We can now finally prove the analytical main theorem of this thesis: the Fredholm property for the compatibility equations. The proof will be rather short, since most of the work has already been done in Chapter 3. **4.3.3 Theorem.** $\mathcal{D}: \mathcal{V} \to \mathcal{V}$ is bounded, and the operator $I - \mathcal{D}: \mathcal{V} \to \mathcal{V}$ is Fredholm of index 0, that is, there exist two operators $\mathcal{B}, \mathcal{C}: \mathcal{V} \to \mathcal{V}$ such that

$$I - \mathcal{D} = \mathcal{B} + \mathcal{C},$$

where \mathcal{B} is bounded and boundedly invertible and \mathcal{C} is compact.

Proof. For $u = (u_0^+, u_0^-, \dots, u_2^+)^\top$ and $v = (v_0^-, v_0^+, \dots, v_2^+)^\top$, let us write $(I - \mathcal{D})u = v$ in an operator matrix notation. We have

$$\begin{bmatrix} I - \begin{pmatrix} & \mathcal{D}_1^- & \mathcal{D}_1^- & & \\ & & & \mathcal{D}_2^+ & \mathcal{D}_2^+ \\ \mathcal{D}_0^+ & \mathcal{D}_0^+ & & & \\ \mathcal{D}_0^- & \mathcal{D}_0^- & & & & \\ & & & \mathcal{D}_1^+ & \mathcal{D}_1^+ & & \end{pmatrix} \end{bmatrix} \begin{pmatrix} u_0^- \\ u_0^+ \\ u_1^- \\ u_1^+ \\ u_2^- \\ u_2^+ \end{pmatrix} = \begin{pmatrix} v_0^- \\ v_0^+ \\ v_1^- \\ v_1^+ \\ v_2^- \\ v_2^+ \end{pmatrix}.$$
(4.3.1)

Consider the first line of this equation: it is

$$v_0^- = u_0^- - \mathcal{D}_1^- u_1^- - \mathcal{D}_1^- u_1^+.$$

The mapping properties of the identity part are clear, and we only have to consider how \mathcal{D}_1^- acts on u_1^- and u_1^+ . For clarification, we have sketched the supports of the source and targets of \mathcal{D}_1^- in Figure 4.4. By definition of \mathcal{D}_1^- , we have that

$$[\mathcal{D}_1^- u_1^-](\xi) = [\mathcal{D}_{c_1^+}^{\theta_1^+} u_1^+](c_0^- - \xi), \qquad \text{for } \xi < c_0^-.$$

Recall that by definition $u_1^- \in H_{00}^{1/2}(-\infty, c_1^-)$. Since $c_1^- < c_1^+$, we obtain from Corollary 3.6.5 that $D_{c_1^+}^{\theta_1^+} : H_{00}^{1/2}(-\infty, c_1^-) \to H_{00}^{1/2}(0, +\infty)$ is compact. \mathcal{D}_1^- is the composition of this operator with a translation/mirroring operator; keeping this in mind, we obtain that $\mathcal{D}_1^- : H_{00}^{1/2}(-\infty, c_1^-) \to H_{00}^{1/2}(-\infty, c_0^-)$ is also compact. Let us now consider how \mathcal{D}_1^- acts on u_1^+ , that is, let us consider the mapping prop-

Let us now consider how \mathcal{D}_1^- acts on u_1^+ , that is, let us consider the mapping properties of $\mathcal{D}_1^-: H_{00}^{1/2}(c_1^+, \infty) \to H_{00}^{1/2}(-\infty, c_0^-)$. In this case, we can apply Corollary 3.6.7 to obtain that $D_{c_1^+}^{\theta_1^+}: H_{00}^{1/2}(c_1^+, \infty) \to H_{00}^{1/2}(0, +\infty)$ can be split into two operators

$$D_{c_1^+}^{\theta_1^+} = B + C,$$

where ||B|| < 1 and C is compact. Since again \mathcal{D}_1^- is the composition of $D_{c_1^+}^{\theta_1^+}$ with a mirroring/translation operator, we obtain the same splitting for \mathcal{D}_1^- , i.e. there exist operators $\mathcal{B}_1^-, \mathcal{C}_1^- : H_{00}^{1/2}(c_1^+, \infty) \to H_{00}^{1/2}(-\infty, c_0^-)$ such that

$$\mathcal{D}_1^- = \mathcal{B}_1^- + \mathcal{C}_1^-,$$

where $\|\mathcal{B}_1^-\| < 1$, and \mathcal{C}_1^- is compact. Similarly, one proceeds for all other lines of the operator matrix: whenever the sign in the operator agrees with the sign of the function upon which it acts, the operator is compact, i.e. \mathcal{D}_n^{σ} in the pairing $\mathcal{D}_n^{\sigma} u_n^{\sigma}$ is compact (by Theorem 3.6.4 or Corollary 3.6.5). If the signs differ (i.e. in the pairing $\mathcal{D}_n^{\sigma} u_n^{-\sigma}$), the corresponding operator can be split into a bounded and a compact part by Theorem



Figure 4.4: Sketch of the supports of the domains of \mathcal{D}_0^- , as considered in the proof of Theorem 4.3.3. Depending on the domain of \mathcal{D}_0^- , it is either compact or the sum of a compact and a bounded operator.

3.6.6 or Corollary 3.6.7. This way, we can split the operator matrix I - D into a bounded and a compact operator,

$$I - \mathcal{D} = (I - \mathcal{B}) + \mathcal{C},$$

where the compact part is given by

$$\mathcal{C} = \begin{pmatrix} & \mathcal{D}_1^- & \mathcal{C}_1^- & & \\ & & & \mathcal{C}_2^+ & \mathcal{D}_2^+ \\ & & & \mathcal{D}_2^- & \mathcal{C}_2^- \\ \mathcal{D}_0^- & \mathcal{C}_0^- & & & \\ & & \mathcal{C}_1^+ & \mathcal{D}_1^+ & & \end{pmatrix},$$

while the bounded part is given by

$$I - \mathcal{B} = \begin{pmatrix} & \mathcal{B}_1^- & & \\ & & \mathcal{B}_2^+ & \\ & & & \mathcal{B}_2^- \\ & & & \mathcal{B}_0^- & \\ & & \mathcal{B}_1^- & & & \end{pmatrix}$$

We claim that $I - \mathcal{B}$ is bounded and boundedly invertible: in fact, this follows since each line and each row of the matrix \mathcal{B} contains exactly one operator, so that

$$\|\mathcal{B}\| = \max_{\substack{n \in \{0,1,2\}\\\sigma \in \{\pm\}}} \|\mathcal{B}_n^{\sigma}\| < 1,$$

since $\|\mathcal{B}_n^{\sigma}\| < 1$ for any $n \in \{0, 1, 2\}$ and $\sigma \in \{\pm\}$. Using a Neumann series argument one obtains the bounded invertibility of $I - \mathcal{B}$, finishing the proof.

Having the Fredholm alternative at hand (see for example [59, Theorem 2.27]), uniqueness and existence are equivalent. In fact, we can show that uniqueness holds, and obtain the following theorem.

4.3.4 Theorem. $I - \mathcal{D} : \mathcal{V} \to \mathcal{V}$ is boundedly invertible, that is, for any $f \in \mathcal{V}$ there exists exactly one solution $u \in \mathcal{V}$ of

$$(I - \mathcal{D})u = f.$$

Proof. By Theorem 4.3.3, $I - \mathcal{D} : \mathcal{V} \to \mathcal{V}$ is Fredholm, and as a consequence of the Fredholm alternative, it is boundedly invertible if and only if

$$(I - \mathcal{D})u = 0$$

possesses only the trivial solution u = 0. Hence, let $u = (u_0^-, u_0^+, \dots, u_2^+) \in \mathcal{V}$ be such that $(I - \mathcal{D})u = 0$. For $n \in \{0, 1, 2\}$, define $u_n \in H^{1/2}(\Gamma_n)$ by

$$u_n(x) = \begin{cases} u_n^-(x) & \text{ for } x \in \Gamma_n^-, \\ 0 & \text{ for } x \in \Gamma_n^0, \\ u_n^+(x) & \text{ for } x \in \Gamma_n^+. \end{cases}$$

Note that since $u_n^- \in H_{00}^{1/2}(-\infty, c_n^-)$ and $u_n^+ \in H_{00}^{1/2}(c_n^+, \infty)$, the extension u_n by 0 in between yields a function in $H^{1/2}(\Gamma_n)$. Let us now define $U_n \in H^1(\Omega_n)$ by

$$U_n := \mathcal{S}_n u_n.$$

By definition, U_n fulfils $(\Delta + \kappa - q)U_n = 0$ in Ω_n , together with the boundary condition $U_n = u_n$ on Γ_n . We now claim that for any $n \in \mathbb{Z}/3\mathbb{Z}$ it holds that

$$U_n = U_{n+1}$$
 in $\Omega_n \cap \Omega_{n+1}$.

Since $U_n, U_{n+1} \in H^1(\Omega_n \cap \Omega_{n+1})$ are solutions to the homogeneous Helmholtz equation on this domain, it is sufficient (due to the uniqueness of Theorem 1.4.3) to prove their traces agree on $\partial(\Omega_n \cap \Omega_{n+1}) = \Gamma_n^- \cap \Gamma_{n+1}^+$. We have by definition of S_n that

$$U_n|_{\Gamma_n^-} = u_n^-,$$

as well as

$$U_n|_{\Gamma_{n+1}^+} = \mathcal{D}_n^+(u_n^+ + u_n^-).$$

From (I - D)u = 0, one obtains from the line corresponding to u_{n+1}^+ that

$$u_{n+1}^{+} - \mathcal{D}_{n}^{+}u_{n}^{+} - \mathcal{D}_{n}^{+}u_{n}^{-} = 0.$$

The last two equations imply that

$$U_n|_{\Gamma_{n+1}^+} = u_{n+1}^+.$$

Analogously, one can now show that we have $U_{n+1}|_{\Gamma_{n+1}^+} = u_{n+1}^+$ and $U_{n+1}|_{\Gamma_n^-} = u_n^-$, so that the Dirichlet traces of U_n and U_{n+1} agree on $\partial(\Omega_n \cap \Omega_{n+1})$.

Hence we have shown that $U_n = U_{n+1}$ on the intersection of their domains $\Omega_n \cap \Omega_{n+1}$, and we can define $U \in H^1(\mathbb{R}^2 \setminus \overline{\Omega})$ by

$$U(x) = U_n(x)$$
 for $x \in \Omega_n$.

U fulfils $(\Delta + \kappa - q)U = 0$ in $\mathbb{R}^2 \setminus \Omega$ and U = 0 on $\partial \Omega$, since we have set $u_n^0 = 0$ in the beginning. By Theorem 1.4.3, this implies U = 0 and finishes the proof.

4.3.5 Equivalence of the problems. Let $g \in H^{1/2}(\partial\Omega)$ be given with the corresponding solution U to (4.2.1), and let the source $f \in \mathcal{V}$ and $w \in \mathcal{V}$ be constructed as in Subsection 4.3.2. Then w and f solve

$$(I - \mathcal{D})w = f,$$

that is, from a solution to the Helmhotz equation we can obtain a solution to the compatibility equations.

Conversely, if one takes some $g \in H^{1/2}(\partial\Omega)$ and constructs f as in Subsection 4.3.2, we obtain the existence of exactly one solution $w \in \mathcal{V}$ of

$$(I - \mathcal{D})w = f.$$

From w, we can reconstruct the solution U on the exterior by setting

$$u_n = w_n^+ + w_n^- + w_n^0,$$

and then using S_n to extend it to the halfspace Ω_n . Following the proof of the previous theorem, one can than again show that this gives a (well-defined) solution U to (4.2.1). That is, from a solution to the compatibility equations we can reconstruct the solution of (4.2.1).

In other words: (I - D)w = f and (4.2.1) are equivalent.

4.3.6 Other functional frameworks. In [74, Chapter 6], other functional frameworks are proposed for halfspace matching. Most notably, by introducing an overlap between the interior triangle Ω and the halfspaces Ω_n , $n \in \{0, 1, 2\}$, it is possible to derive a Fredholm property for the problem on the full space, that is, for example for the problem: for given $f \in L^2(\Omega)$, find some $u \in H^1(\mathbb{R}^2)$ such that

$$\Delta u(x) + \kappa u(x) = f(x) \quad \text{for } x \in \mathbb{R}^2.$$

An example for the domain splitting is shown in Figure 4.5. While being more complicated on first sight, the analysis turns out to be simplified by the inclusion of the triangle Ω with overlap. This stems from the fact that the behaviour of the Dirichlet-to-Dirichlet operators at the edges is not of relevance. Again, we refer to [74, Chapter 6] for more details.

It should pose no problem to translate this analysis to the case with waveguides. We did not include this here, mostly for the reason that this functional framework is farther from the numerical implementation in the following chapter, where we did not implement an overlap with the interior domain.



Figure 4.5: An exemplary splitting into an interior domain, which overlaps with the exterior halfspaces Ω_0 , Ω_1 , Ω_2 .

5 Implementation for the Full Space

5.1 Introduction

5.1.1 Introduction. In this chapter we will describe a weak formulation for Helmholtz problems in \mathbb{R}^2 , which utilises the compatibility equations (4.2.7) to treat the exterior domain. We only introduce and describe the method, any numerical analysis is far outside the scope of this work. Accordingly, we will mostly do some formal calculations, followed by a description of the discretisation used. Let us give a quick overview.

We start in Section 5.2 to derive a weak formulation for Helmholtz problems of the type

$$\Delta u + (\kappa - q)u = f \qquad \text{in } \mathbb{R}^2,$$

where $\kappa \in \mathbb{C} \setminus \mathbb{R}$, that is, we consider again only the absorptive case. Here, we can still apply Theorem 1.4.3 to obtain well-posedness in $H^1(\mathbb{R}^2)$, and accordingly, the functional framework can be chosen in a rather simple fashion.

Afterwards, we will continue to the interesting case and consider (for certain incident fields u_i) the problem of finding $u \in H^2_{loc}(\mathbb{R}^2)$ such that

$$\begin{cases} \Delta u + (\kappa_0 - q)u = 0 & \text{ in } \mathbb{R}^2, \\ u - u_{\mathbf{i}} & \text{ outgoing.} \end{cases}$$

where $\kappa_0 \in \mathbb{R}$ is a properly real wavenumber. At this point, a number of complications arise, which we will deal with in a rather formal way. We need to define the notion of an "outgoing wave". Motivated from Section 3.3, we will use our halfspace solution expansion to define a radiation condition for each halfspace. This will allow us to derive a (distributional) weak formulation, utilising again the compatibility equations.

The discretisation we used for creating numerical examples will be shown in Section 5.4. We describe how the involved matrices can be numerically computed, and will show a few convergence results afterwards.

5.1.2 Assumption. Let $q : \mathbb{R}^2 \to \mathbb{R}$ be some real-valued $L^{\infty}(\mathbb{R}^2)$ function such that there exists a triangular (or rectangular) domain Ω such that $q|_{\mathbb{R}^2\setminus\Omega}$ fulfils Definition and Assumption 4.2.3. We denote by \boldsymbol{m} the number of sides of Ω . Furthermore, we assume that in each halfspace $n \in \{0, \ldots, \boldsymbol{m} - 1\}$, the potential $q^{(n)} : \mathbb{R} \to \mathbb{R}$ is of Pekeris type (compare Section 2.3). That is, there exist three constants $q_{-}^{(n)}, q_{+}^{(n)}, q_{m}^{(n)} \in \mathbb{R}$ and $a^{(n)} < b^{(n)}$ such that

$$q^{(n)}(\xi) = \begin{cases} q_{-}^{(n)} & \text{for } \xi < a^{(n)}, \\ q_{m}^{(n)} & \text{for } a^{(n)} \le \xi \le b^{(n)}, \\ q_{+}^{(n)} & \text{for } b^{(n)} < \xi. \end{cases}$$

We restrict ourselves to potentials of Pekeris type, since for those we can explicitly calculate the generalised Fourier transform as shown in Section 2.3. One might consider more general potentials as in Definition and Assumption 2.2.2, but to realise this, one needs to determine the spectral family Ψ by some numerical method.

Let us also recall the notation of Subsection 4.2.2, which will be exploited heavily in this chapter. Since we now allow for either a triangular or rectangular interior domain Ω , we might have $\mathbf{m} \in \{3, 4\}$ overlapping halfspaces Ω_n with boundaries $\Gamma_n = \Gamma_n^+ \cup \Gamma_n^0 \cup \Gamma_n^-$, $n \in \{0, \ldots, \mathbf{m} - 1\}.$

5.1.3 Allowed geometrical configurations. Before going on, let us discuss shortly the geometrical restrictions that follow from Definition and Assumption 4.2.3 and the previous assumptions. Note that "rectangular" means that we have a quadrangle with right angles at every corner. We restrict ourselves to this situation here. However, it is also possible to allow for splittings of the exterior where more than two halfspace overlap (see Figure 5.1). This can be also treated with our method, but we omit a detailed discussion of this situation here.



Figure 5.1: Sketch of the overlap areas for two quadrangular interior domains. The light blue areas indicate where two halfspaces overlap. For the right hand example, there is a small, red area, where three halfspaces overlap.

Note that for rectangular Ω , every halfspace can contain a waveguide. This is not necessarily the case for triangles, since one cannot fulfil Definition and Assumption 4.2.3, if there is a waveguide adjacent to an obtuse angle (see Figure 5.2).

5.2 The Sesquilinear Form of the Problem

We aim to derive a sesquilinear form which admits the numerical treatment of the following problem.

5.2.1 Problem of interest. For given $\kappa \in \mathbb{C} \setminus \mathbb{R}$ and $f \in L^2(\mathbb{R}^2)$ with $\operatorname{supp}(f) \subset \Omega$, we consider the problem of finding $u \in H^1(\mathbb{R}^2)$ such that

$$\Delta u + (\kappa - q)u = f \qquad \text{in } \mathbb{R}^2.$$

Note that Theorem 1.4.3 gives existence and uniqueness for this problem.



Figure 5.2: Two triangles with a waveguide extending perpendicularly from one side. The left geometry is allowed, while the right one is not admissible, since in the hatched area, Definition and Assumption 4.2.3 cannot be fulfilled by any $q^{(n)}$ for the left hand side halfspace.

5.2.2 The weak formulation. To make the problem suitable for halfspace matching, we will split the domain into an exterior and interior part, and define $u_{\rm e} := u|_{\mathbb{R}^2 \setminus \overline{\Omega}}$ and $u_{\Omega} = u|_{\Omega}$, so that $u_{\rm e} \in H^1(\mathbb{R}^2 \setminus \overline{\Omega})$ and $u_{\Omega} \in H^1(\Omega)$ fulfil

$$\begin{cases} \Delta u_{\rm e} + (\kappa - q)u_{\rm e} = 0 & \text{ in } \mathbb{R}^2 \setminus \overline{\Omega}, \\ u_{\rm e} - u_{\Omega} = 0 & \text{ on } \partial\Omega, \\ \partial_{\nu} u_{\Omega} - \partial_{\nu} u_{\rm e} = 0 & \text{ on } \partial\Omega, \\ \Delta u_{\Omega} + (\kappa - q)u_{\Omega} = f & \text{ in } \Omega. \end{cases}$$
(5.2.1)

Here ∂_{ν} denotes the derivative in direction of the exterior normal on Ω . We note that $u_{\rm e}$ fulfils an exterior problem as considered in Chapter 4 (see Subsection 4.2.1), and accordingly, it solves the corresponding compatibility equations. To write them down, let us introduce the trace unknowns

$$u_n = u_{\mathbf{e}}|_{\Gamma_n}$$
 for $n \in \{0, \dots, m-1\}$.

As in Subsection 4.2.5 we can derive that the traces must fulfil the compatibility equations

$$u_{n+\sigma} - \mathcal{D}_n^{\sigma} u_n = 0$$
 on $\Gamma_n^{\sigma}, \ \sigma \in \{\pm\}, \ n \in \mathbb{Z}/m\mathbb{Z}$

To realise the coupling of the Neumann traces in (5.2.1), we recall the DtN operators from Section 3.4, which are defined by

$$\mathcal{N}_n f(x) := \partial_{\nu} \mathcal{S}_n f(x) \qquad \text{for } x \in \Gamma_n^0, \ n \in \{0, \dots, m-1\},$$

where ν denotes the inward normal of Ω_n , that is, the outward normal on Ω . Hence we arrive at the following problem equivalent to (5.2.1), which contains the new unknowns, the traces (u_0, \ldots, u_{m-1}) and u_{Ω}

$$\begin{cases} u_{n+\sigma} - \mathcal{D}_{n}^{\sigma} u_{n} = 0 & \text{on } \Gamma_{n+\sigma}^{\sigma}, \ \sigma \in \{\pm\}, \ n \in \mathbb{Z}/m\mathbb{Z}, \\ u_{n} - u_{\Omega} = 0 & \text{on } \Gamma_{n}^{0}, \ n \in \{0, \dots, m-1\}, \\ \mathcal{N}_{n} u_{n} - \partial_{\nu} u_{\Omega} = 0 & \text{on } \Gamma_{n}^{0}, \ n \in \{0, \dots, m-1\}, \\ \Delta u_{\Omega} + (\kappa - q) u_{\Omega} = f & \text{in } \Omega. \end{cases}$$

$$(5.2.2)$$

Let us recast this problem into a weak formulation. For this aim, define the space

$$\mathcal{V} := \left\{ (u_{\Omega}, u_0, \dots, u_{m-1}) \in H^1(\Omega) \times \prod_{n=0}^{m-1} H^1(\Gamma_n) : u_{\Omega} = u_n \text{ on } \Gamma_n^0, \ n \in \{0, \dots, m-1\} \right\}$$

The equality of the Dirichlet traces of (5.2.2) is already incorporated by definition of our space \mathcal{V} . Note that this space is somewhat ill-balanced: one would naturally demand the traces to be only $H^{1/2}$ on the boundary. We reiterate that this functional frame is most likely ill-suited for proper numerical analysis.

Let $u = (u_{\Omega}, u_1, \ldots, u_{m-1}) \in \mathcal{V}$ be a solution of (5.2.2), and let $v = (v_{\Omega}, v_1, \ldots, v_{m-1}) \in \mathcal{V}$ be some test function. Consider the solution in the interior u_{Ω} : by multiplying the Helmholtz equation in Ω with \overline{v}_{Ω} and integrating over Ω , we obtain by partial integration that

$$\int_{\Omega} \nabla u_{\Omega} \cdot \nabla \overline{v}_{\Omega} - (\kappa - q) u_{\Omega} \overline{v}_{\Omega} \, \mathrm{d}x - \int_{\partial \Omega} (\partial_{\nu} u_{\Omega}) \overline{v}_{\Omega} \, \mathrm{d}s = -\int_{\Omega} f \overline{v}_{\Omega} \, \mathrm{d}x,$$

The volume integral on the left side will be the first part of our sesquilinear form, so we define

$$b_{\Omega}(u,v) := \int_{\Omega} \nabla u_{\Omega} \cdot \nabla \overline{v}_{\Omega} - (\kappa - q) u_{\Omega} \overline{v}_{\Omega} \,\mathrm{d}x.$$
(5.2.3)

Let us now consider the boundary integral. From $\partial \Omega = \bigcup_{n=0}^{m-1} \Gamma_n^0$, we obtain

$$-\int_{\partial\Omega} (\partial_{\nu} u_{\Omega}) \overline{v}_{\Omega} \, \mathrm{d}s = -\sum_{n=0}^{m-1} \int_{\Gamma_n^0} (\partial_{\nu} u_{\Omega}) \overline{v}_{\Omega} \, \mathrm{d}s = -\sum_{n=0}^{m-1} \int_{\Gamma_n^0} (\mathcal{N}_n u_n) \overline{v}_n \, \mathrm{d}s,$$

where we used the continuity of the Dirichlet trace of $v \in \mathcal{V}$ as well as the continuity of the Neumann trace from (5.2.2). This motivates the definition of the second part of the sesquilinear form, namely

$$b_n^{\mathcal{N}}(u,v) := -\int_{\Gamma_n^0} (\mathcal{N}_n u_n) \overline{v}_n \,\mathrm{d}s.$$
(5.2.4)

It remains to realise the compatibility equations in (5.2.2). To do this, we simply fix $\sigma \in \{\pm\}$, $n \in \mathbb{Z}/\mathbb{mZ}$, multiply the corresponding compatibility equation by $\overline{v}_{n+\sigma}$ and integrate over $\Gamma_{n+\sigma}^{\sigma}$. We obtain that the following sesquilinear form must be always equal to 0:

$$b_n^{\sigma}(u,v) = \int_{\Gamma_{n+\sigma}^{\sigma}} (u_{n+\sigma} - \mathcal{D}_n^{\sigma} u_n) \overline{v}_{n+\sigma} \,\mathrm{d}s.$$
(5.2.5)

Lastly, we denote the sum of all previously defined sesquilinear forms by

$$b(u,v) = b_{\Omega}(u,v) + \sum_{n=1}^{m-1} b_n^{\mathcal{N}}(u,v) + \sum_{n=1}^{m-1} \sum_{\sigma \in \{\pm\}} b_n^{\sigma}(u,v).$$
(5.2.6)

Summarily, we obtain that solution of (5.2.2) fulfils the following problem.

5.2.3 Weak problem. For given $f \in L^2(\mathbb{R}^2)$ such that $\operatorname{supp}(f) \subset \Omega$, find a solution $u \in \mathcal{V}$ such that

$$b(u, v) = -\int_{\Omega} f \overline{v}_{\Omega} \, \mathrm{d}x \quad \text{for all } v \in \mathcal{V},$$

where b is defined by (5.2.6) (which involves the sesquilinear forms (5.2.5), (5.2.4), (5.2.3)).

We will later discretise this variational problem by using a Galerkin method. Note that the sesquilinear form is bounded on \mathcal{V} , as can be easily seen by standard arguments (for b_{Ω}) or using the already shown mapping properties: for $b_n^{\mathcal{N}}$ check Lemma 3.4.2. To show that b_n^{σ} is bounded, one simply applies the trace theorem for one of the halfspace problems.

Let us remark here that the choice of our space and sesquilinear form merely allows us to write down a variational problem. Note that they are also somewhat ill adapted to each other, since we use H^1 -unknowns for the semi-infinite lines, while we use an L^2 -scalar product to realise the compatibility equations. Consequently, it seems questionable whether one can use this framework for numerical analysis.

5.3 The Non-absorptive Case, Incident Waves

5.3.1 Introduction. In this section, we aim to consider for

$$\kappa_0 \in \mathbb{R},$$

the Helmholtz equation without absorption, that is

$$\Delta u + (\kappa_0 - q)u = 0 \qquad \text{in } \mathbb{R}^2.$$

With this choice, the classical problems associated with the mathematical theory of scattering problems arises: we will need a radiation condition, which ensures the well-posedness of the problem, and corresponding to this we need a functional analytic framework. This is no easy task, and we discuss it only in a formal fashion.

To have a starting point, we will define an (not explicitly specified) trace space V_2 , that fulfils certain properties, allowing us to define the "outward propagation radiation condition". This will then allow us to write down a distributional formulation of our scattering problem, utilising the sesquilinear form b from the previous section for special test functions.

Another important topic of this section is the realisation of incident waves in the weak formulation, which will first be illustrated for a simple, unperturbed waveguide, and then incorporated in the general weak formulation.

We begin by specifying the properties we will require from our mystery trace space $V_{?}$.

5.3.2 Definition and Assumption. Consider again the halfspace \mathbb{R}_2^+ . Let $\kappa_0 \in \mathbb{R}$, and let $q : \mathbb{R}^2 \to \mathbb{R}$ fulfil Definition and Assumption 2.2.2. We now assume that

$$V_? \subset \{f : \mathbb{R} \to \mathbb{C}\}$$

is some function space such that:

(a) The associated solution operator $S: H^{1/2}(\mathbb{R}) \to H^1(\mathbb{R}^2_+, (1+x_2)^{-s} dx)$ (see Lemma 3.3.1) can be extended to a continuous operator

$$\mathcal{S}: V_? \to H^1_{\mathrm{loc}}(\mathbb{R}^2_+).$$

Furthermore $\mathcal{S}f|_{\{0\}\times\mathbb{R}} = f$ in $H^{1/2}_{\text{loc}}(\mathbb{R})$ for any $f \in V_{?}$.

(b) The associated DtN operator $\mathcal{N}: H^1(\mathbb{R}) \to L^2(\mathbb{R})$ can be extended to a continuous operator

$$\mathcal{N}: V_? \to L^2_{\mathrm{loc}}(\mathbb{R}).$$

(c) It contains at least the functions e^{ik} for $k \in \mathbb{R}$.

Let us comment a bit on this space, and what might be possible candidates. Note that (b) implies that functions in V_2 should be at least in $H^1_{\text{loc}}(\mathbb{R})$, since otherwise the regularity of the Neumann trace cannot be realised. On the other hand, assumption (c) requires that this space should be rather large. We conjecture that the choice

$$V_{?} = H^{1}_{\text{loc}}(\mathbb{R}) \cap L^{\infty}(\mathbb{R}),$$

might fulfil these assumptions. Why? The first critical point is the extension of the solution operator S. Let us sketch how this might be possible. Let $f \in C_0^{\infty}(\mathbb{R})$. Then we have by Fubini's theorem

$$\begin{split} \mathcal{S}f(x) &= \int\limits_{\Lambda} e^{i\sqrt{\kappa - \hat{\lambda}(m)} \, x_2} \Psi(m, x_1) \int\limits_{\mathbb{R}} \overline{\Psi}(m, y) f(y) \, \mathrm{d}y \, \mathrm{d}\mu(m) \\ &= \int\limits_{\mathbb{R}} f(y) \int\limits_{\Lambda} e^{i\sqrt{\kappa - \hat{\lambda}(m)} \, x_2} \Psi(m, x_1) \overline{\Psi}(m, y) \, \mathrm{d}\mu(m) \, \mathrm{d}y \\ &= \int\limits_{\mathbb{R}} f(y) \Phi(x, y) \, \mathrm{d}y, \end{split}$$

where we defined for $x \in \mathbb{R}^2_+$ and $y \in \mathbb{R}$ the function

$$\Phi(x,y) := \int_{\Lambda} e^{i\sqrt{\kappa - \hat{\lambda}(m)} x_2} \Psi(m,x_1) \overline{\Psi}(m,y) \, \mathrm{d}\mu(m).$$

It is now easy to see that the integral on the right hand side is well-defined. Furthermore, we conjecture that with the help of integration by parts and the stationary phase method (see [21, Section 2.7]) one can show that for fixed $x \in \mathbb{R}^2_+$

$$|\Phi(x,y)| = c(x) \frac{1}{|y|^{3/2}} (1 + \mathcal{O}(|y|^{-1}))$$
 as $|y| \to \infty$.

This would imply that the integral

$$Sf(x) = \int_{\mathbb{R}} f(y)\Phi(x,y) \,\mathrm{d}y$$

is well defined for any $f \in L^{\infty}(\mathbb{R})$, since $\Psi(x, \cdot) \in L^{1}(\mathbb{R})$ for any $x \in \mathbb{R}^{2}_{+}$. Hence, one can extend S to $L^{\infty}(\mathbb{R})$, which together with the additional regularity assumptions should be able to yield the desired properties.

94

5.3. THE NON-ABSORPTIVE CASE, INCIDENT WAVES

Let us remark that the previous approach of defining an extension to S is rather well known for the case of a free halfspace in the context of the upward propagation radiation condition. We refer to the references after Definition 5.3.4.

Let us also mention that in [74, Section 4.4], it is suggested that the space

$$V_{\kappa_0} = \{ f : \mathbb{R} \to \mathbb{C} : |\kappa_0 - \hat{\lambda}|^{1/4} \mathcal{F}_A f \in L^2(\Lambda, \mathrm{d}\mu) \}$$

might be a suitable choice. Therein, however, lie a few complications. Firstly, it is necessary to properly define the generalised Fouriertransform on \mathcal{F}_A on V_{κ_0} , since V_{κ_0} is not contained in $L^2(\mathbb{R})$, due to the zero of the weight $|\kappa_0 - \hat{\lambda}|^{1/4}$. Hence it is necessary to define appropriate distributions adapted to \mathcal{F}_A and extend the transform accordingly. Luckily, at least this problem has been overcome and has been discussed in [10] and [39]. Secondly, this space is not large enough for our purpose. Definition and Assumption 5.3.2 (c) is violated, which prohibits us from considering Example 5.3.5 below.

One might also ask the following question: assume $L^{\infty}(\mathbb{R}) \cap H^1_{\text{loc}}(\mathbb{R})$ (or something similar) is a suitable choice for $V_?$. Might it be possible to reproduce the analysis of Chapter 3 and 4, to obtain a similar Fredholm alternative? The answer is, most likely, negative. One can show that on the free halfspace the condition $f \in L^{\infty}(\mathbb{R})$ is not sufficient to ensure that Sf is in $L^{\infty}(\mathbb{R}^2_+)$ (see [13]). This, however, should be the case to guarantee $D_c^{\theta}f \in L^{\infty}(0,\infty)$ for some Dirichtlet-to-Dirichlet operator D_c^{θ} . At least in the authors opinion, this question does not allow an obvious answer.

We can now define the spaces which will be involved in our absorption free problem.

5.3.3 Definition. Define

$$\mathcal{V}_{?} := \left\{ (u_{\Omega}, u_{0}, \dots, u_{\boldsymbol{m}-1}) \in H^{1}(\Omega) \times \prod_{n=0}^{\boldsymbol{m}-1} V_{?}(\Gamma_{n}) : u_{\Omega} = u_{n} \text{ on } \Gamma_{n}^{0} \text{ for } n \in \{0, \dots, \boldsymbol{m}-1\} \right\}.$$

We furthermore define the following space of test functions

$$\mathcal{V}_{?}^{*} := \left\{ (u_{\Omega}, u_{0}, \dots, u_{m-1}) \in H^{1}(\Omega) \times \prod_{n=0}^{m-1} L^{2}(\Gamma_{n}) : \sup (u_{n}) \text{ is compact for } n \in \{0, \dots, m-1\} \right\}.$$

We will use a different test space in the variational formulation, since our assumptions for $V_{?}$ do not ensure that the Dirichlet-to-Dirichlet operators \mathcal{D}_{n}^{σ} stay bounded (for $n \in \{0, ..., m - 1\}, \sigma \in \{\pm\}$). Consequently, the sesquilinear form b_{n}^{σ} will generally be unbounded on $\mathcal{V}_{?} \times \mathcal{V}_{?}$, so we restrict ourselves to $\mathcal{V}_{?} \times \mathcal{V}_{?}^{*}$, switching effectively to a distributional formulation.

5.3.4 Definition. Let $\hat{\Omega}$ be some halfspace, $\kappa_0 \in \mathbb{R}$, and let $u \in H^2_{\text{loc}}(\mathbb{R})$ be a solution to the Helmholtz equation

$$\Delta u + (\kappa_0 - q)u = 0 \quad \text{in } \Omega.$$

We say that u fulfils the outward propagation radiation condition (OPRC) in $\hat{\Omega}$, if $u|_{\partial\hat{\Omega}} \in V_{2}(\partial\hat{\Omega})$, and

$$u = \mathcal{S}[u|_{\partial \hat{\Omega}}].$$

We motivated this radiation condition in Section 3.3, where we deduced that the limitabsorption solution on a halfspace must fulfil this condition, provided that $u|_{\partial\hat{\Omega}} \in H^{1/2}(\partial\hat{\Omega})$. For the free space, this radiation condition has been thoroughly analysed and is called the "upward propagation radiation condition", for example by Chandler-Wilde and others (see for example [2, 14, 15] and references therein) in the context of rough layer scattering.

For waveguides, the same radiation condition has already been applied, most notably by Bonnet-ben Dhia, Hazard and others. Certain well-posedness results can be found for special geometries (see [9, 10]). Note however, that these results are restricted to the case of parallel waveguides, and do not consider overlapping domains in the exterior. There are no results for non-parallel waveguides to the author's knowledge.

5.3.5 Example. Let us illustrate at a simple example how this radiation condition can be applied. We will also introduce a certain splitting in incident and outgoing parts for a solution, which might seem arbitrary at first sight, but will be used for different scattering problems, where this particular splitting is of importance. Let h > b > 0, $q_0 < 0$, $\kappa_0 > 0$ and define for $x_1 \in \mathbb{R}$

$$q(x_1) = \begin{cases} 0 & \text{for } |x_1| > b, \\ q_0 & \text{for } |x_1| \le b. \end{cases}$$

Let $A = -\partial_{x_1}^2 + q$ be the associated operator, and let $\psi \in H^2(\mathbb{R})$ be the first eigenfunction of A corresponding to the eigenvalue $\lambda_1 \in (q_0, 0)$. It has a representation of the form

$$\psi(x_1) = \begin{cases} ce^{-\sqrt{|\lambda_1|} x_1} & \text{for } x_1 > h, \\ ce^{\sqrt{|\lambda_1|} x_1} & \text{for } x_1 < -h. \end{cases}$$

Let us now consider the function

$$u(x) = e^{-i\sqrt{\kappa_0 - \lambda_1} x_2} \psi(x_1)$$
 for $x \in \mathbb{R}^2$,

which is a solution to the Helmholtz equation

$$\Delta u(x) + (\kappa_0 - q(x_1))u(x) = 0 \quad \text{for } x \in \mathbb{R}^2.$$

We will now write $u : \mathbb{R}^2 \to \mathbb{C}$ as the solution to a very particular scattering problem with the help of the halfspace representations. We define the square

$$\Omega = \{ x \in \mathbb{R}^2 : |x_1| < h, |x_2| < h \},\$$

as well as the four halfspaces (which are subset of \mathbb{R}^2)

$$\Omega_0 = \{x_1 < -h\}, \ \Omega_1 = \{x_2 < -h\}, \ \Omega_2 = \{x_1 > h\}, \ \Omega_3 = \{x_2 > h\},$$

and the boundaries $\Gamma_n := \partial \Omega_n$, $n \in \{0, 1, 2, 3\}$. As before, let us denote the traces by $u_n = u|_{\Gamma_n}$, $n \in \{0, \ldots, 3\}$. In which halfspaces can u be considered outgoing?

(0) On the halfspace Ω_0 the function q is constantly equal to 0. Hence we can write down the solution operator \mathcal{S}_0 for $g \in H^{1/2}(\mathbb{R})$ by

$$\mathcal{S}_0 g(x) = \frac{1}{\sqrt{2\pi}} \int\limits_{\mathbb{R}} e^{-i\sqrt{\kappa_0 - \xi^2} (h - x_1)} e^{i\xi x_2} \hat{g}(\xi) \,\mathrm{d}\xi,$$



Figure 5.3: Sketch of the geometrical notation (left) and the real part of the solution for Example 5.3.5. The solution on the right hand side can be considered as propagating downwards.

where \hat{g} denotes the 1D Fourier transform of g. The trace on $\Gamma_0 = \{x \in \mathbb{R}^2 : x_1 = -h\}$ is given by

$$u_0(x_2) = \psi(-h) \exp(-i\sqrt{\kappa_0 - \lambda_1} x_2).$$

We can now formally apply the Fourier transform (in the sense of distributions) to obtain

$$\hat{u}_0(\xi) = \psi(-h)\delta(\xi - \sqrt{\kappa_0 - \lambda_1}),$$

where δ denotes the δ -distribution. Interpreting the inverse Fourier transform in the sense of distribution, we obtain that

$$S_0 u_0(x) = \psi(-h) e^{-i\sqrt{\kappa_0 - (\kappa_0 - \lambda_1)} (h + x_1)} e^{-i\sqrt{\kappa_0 - \lambda_1} x_2}.$$

Keeping the branch cut of the square root in mind and using the representation for ψ , one easily sees that the right hand side is equal to u, so that we obtain

$$u(x) = S_0 u_0(x) \qquad \text{on } \Omega_0,$$

that is, u is outgoing on Ω_0 .

(1) On Γ_1 we have that

$$u_1(x_1) = \psi(x_1)e^{i\sqrt{\kappa_0 - \lambda_1}h}.$$

This function is clearly in $H^2(\mathbb{R})$, and we can employ \mathcal{F}_A in the classical sense. One easily obtains

$$(\mathcal{F}_A u_1)^p(\lambda_1) = e^{i\sqrt{\kappa_0 - \lambda_1} h}$$
 and $\mathcal{F}_A u_1(m) = 0$ for $m \in \Lambda \setminus \{(p, \lambda_1)\}.$

Employing the inverse transform, this in turn gives

$$S_1 u_1(x) = e^{-i\sqrt{\kappa_0 - \lambda_1}(x_2 + h)} e^{+i\sqrt{\kappa_0 - \lambda_1} h} \psi(x_1) = u(x),$$

showing that u can be considered outgoing in Ω_1 .

- (2) On Ω_2 the same discussion as for Ω_0 yields that u is outgoing here, too.
- (3) For the last halfspace, we obtain that

$$u_3(x_1) = \psi(x_1)e^{-i\sqrt{\kappa_0 - \lambda_1}h}.$$

Plugging this boundary data into the solution operator now yields

$$\mathcal{S}_3 u_3(x) = \psi(x_1) e^{-i\sqrt{\kappa_0 - \lambda_1} h} e^{i\sqrt{\kappa_0 - \lambda_1} (x_2 - h)}$$

which clearly is different from u, so that u is not outgoing in Ω_3 . However, if we define the incident and scattered field on Ω_3 by $u_i := u$ and $u_s := u - u_i$, we obtain that

$$u = u_{\mathbf{i}} + \mathcal{S}_3(u_{\mathbf{s}}|_{\Gamma_3}) = u_{\mathbf{i}} + \mathcal{S}_3(u_3 - u_{\mathbf{i}}|_{\Gamma_3}),$$

so we can represent the total field u as a superposition of incident and outgoing field.

Summarily, we see that u fulfils

$$\begin{cases} u = \mathcal{S}_n u_n & \text{on } \Omega_n, \ n \in \{0, 1, 2\}, \\ u = \mathcal{S}_3(u_3 - u_i|_{\Gamma_3}) + u_i & \text{on } \Omega_3, \\ \Delta u + (\kappa_0 - q)u = 0 & \text{on } \mathbb{R}^2, \end{cases}$$

which can be read as a scattering problem: find a solution u to the Helmholtz equation, such that u is outgoing on $\Omega_0, \Omega_1, \Omega_2$, while $u - u_i$ is outgoing on Ω_3 . This is the type of problem for which we now want to derive a weak formulation.

5.3.6 Weak formulation of scattering problems. Let $q \in L^{\infty}(\mathbb{R}^2)$ fulfil Assumption 5.1.2, $\kappa_0 \in \mathbb{R}$ and let $u_i \in H^2_{loc}(\Omega_0)$ be an incident field, such that there exist functions $w_1 \in V_?(\Gamma_1)$ and $w_{m-1} \in V_?(\Gamma_{m-1})$ such that

$$u_{\mathbf{i}} = \mathcal{S}_1 w_1 \quad \text{on } \Omega_0 \cap \Omega_1, \qquad u_{\mathbf{i}} = \mathcal{S}_{m-1} w_{m-1} \quad \text{on } \Omega_{m-1} \cap \Omega_0.$$

The last statement formulates that we can consider u_i as outgoing on the two halfspaces Ω_1 and Ω_{m-1} , where it is partially defined. Let $u \in H^2_{loc}(\mathbb{R}^2)$ be a solution of

$$\begin{cases} \Delta u + (\kappa_0 - q)u = 0 & \text{ in } \mathbb{R}^2, \\ u - u_i & \text{ OPRC in } \Omega_0, \\ u & \text{ OPRC in } \Omega_n, \ n \in \{1, \dots, m-1\}. \end{cases}$$
(5.3.1)

Let us reformulate this problem by utilising the compatibility equations. Define again

$$u_n = u|_{\Gamma_n}$$
 $n \in \{0, \ldots, m-1\}.$

Then we have by considering the solution on Ω_n , $n \in \{1, \ldots, m-1\}$, that

$$u_{n\pm\sigma} = [\mathcal{S}_n u_n]|_{\Gamma_{n+\sigma}^{\sigma}} = \mathcal{D}_n^{\sigma} u_n \qquad \text{on } \Gamma_{n+\sigma}^{\sigma}, \ \sigma \in \{\pm\}, \ n \in \mathbb{Z}/3\boldsymbol{m}\mathbb{Z} \setminus \{0\}$$

This is just the same set of compatibility equations as before. For those halfspaces, the DtN part also takes the same form as previously.

5.4. DISCRETISATION

Let us now consider the halfspace Ω_0 , where our incident wave is defined. Denoting $u_{i,0} := u_i|_{\Gamma_0}$, we have

$$u = S_0(u_0 - u_{i,0}) + u_i = S_0 u_0 + (u_i - S_0 u_{i,0})$$
 on Ω_0

which translates by taking the trace on Γ_{σ}^{σ} to

$$u_{\sigma} = \mathcal{D}_0^{\sigma} u_0 + (u_i|_{\Gamma_{\sigma}^{\sigma}} - \mathcal{D}_0^{\sigma} u_{i,0}) \quad \text{on } \Gamma_{\sigma}^{\sigma}, \ \sigma \in \{\pm\}.$$

Similarly, for the DtN part we obtain that

$$\partial_{\nu} u = \partial_{\nu} (\mathcal{S}(u_0 - u_{\mathbf{i},0})|_{\Gamma_0^0} + u_{\mathbf{i}}) = \mathcal{N}_0 u_0 + [\partial_{\nu} \mathcal{S}_0 u_{\mathbf{i},0} - \partial_{\nu} u_{\mathbf{i}}] \quad \text{on } \Gamma_0^0.$$

We obtain that our solution variables $(u_{\Omega}, u_0, \ldots, u_{m-1})$ fulfil

$$\begin{aligned} u_{\sigma} - \mathcal{D}_{0}^{\sigma} u_{0} &= \mathcal{D}_{0}^{\sigma} u_{i,0} - u_{i} & \text{on } \Gamma_{0+\sigma}^{\sigma}, \ \sigma \in \{\pm\}, \\ u_{n+\sigma} - \mathcal{D}_{n}^{\sigma} u_{n} &= 0 & \text{on } \Gamma_{n+\sigma}^{\sigma}, \ \sigma \in \{\pm\}, \ n \in (\mathbb{Z}/m\mathbb{Z}) \setminus \{0\}, \\ u_{n} - u_{\Omega} &= 0 & \text{on } \Gamma_{n}^{0}, \ n \in \{0, \dots, m-1\}, \\ \mathcal{N}_{0} u_{0} - \partial_{\nu} u_{\Omega} &= \partial_{\nu} u_{i} - \partial_{\nu} \mathcal{S}_{0} u_{i,0} & \text{on } \Gamma_{0}^{0}, \\ \mathcal{N}_{n} u_{n} - \partial_{\nu} u_{\Omega} &= 0 & \text{on } \Gamma_{n}^{0}, \ n \in \{1, \dots, m-1\}, \\ \Delta u_{\Omega} + (\kappa - q) u_{\Omega} &= 0 & \text{in } \Omega. \end{aligned}$$

Which yields, by the same discussion as before, the following weak formulation.

5.3.7 Weak problem. In the notation of the previous section, the weak formulation of (5.3.1) is given as follows: find $u \in \mathcal{V}_{?}$ such that

$$b(u, \psi) = f(\psi)$$
 for all $\psi \in \mathcal{V}_{?}^{*}$,

where the right hand side is given by

$$f(\psi) := \sum_{\sigma \in \{\pm\}} \int_{\Gamma_{\sigma}^{\sigma}} (\mathcal{S}_0 u_{\mathbf{i},0} - u_{\mathbf{i}}) \overline{\psi}_{\sigma} \, \mathrm{d}s + \int_{\Gamma_0^0} (\partial_{\nu} u_{\mathbf{i}} - \partial_{\nu} \mathcal{S}_0 u_{\mathbf{i},0}) \overline{\psi}_0 \, \mathrm{d}s.$$

Note that by Definition and Assumption 5.3.2, all terms of the sesquilinear form are well-defined for any $u \in \mathcal{V}_{?}$ and $\psi \in \mathcal{V}_{?}^{*}$.

5.4 Discretisation

In this section we will describe different discretisations for the weak problems of the previous two sections. Note that here we do not differentiate between the two cases, so that we use the same discretisation independent of κ . In essence, we will describe a Galerkin ansatz space \mathcal{V}^h fulfilling

$$\mathcal{V}^h\subset\mathcal{V}\subset\mathcal{V}_?$$

and show how the different terms of the sesquilinear forms b can be computed.

Let us remark that so far, we have not shown the (analytical) well-posedness of the sesquilinear form in the absorptive case, so it is not obvious how to proceed with the numerical analysis. For the absorption free case, note that with our conjectured choice $V_? = L^{\infty}(R) \cap H^1_{\text{loc}}(\mathbb{R})$, it is not trivial to choose a Galerkin ansatz space which approximates $L^{\infty}(R) \cap H^1_{\text{loc}}(\mathbb{R})$, since, for example, the functions with compact support are not dense in $L^{\infty}(\mathbb{R})$.

Let us recall that geometrically, we have to discretise the interior domain Ω and the semi-infinite lines Γ_n^{σ} , $n \in \{0, \ldots, m-1\}$, $\sigma \in \{\pm\}$.

5.4.1 Description of the discretisation subspace. Let $B_{\Omega} = \{\psi_m \in H^1(\Omega), m \in \{1, \ldots, M_{\Omega}\}\}$ be a first order Galerkin basis of the domain Ω . We define a corresponding subset of \mathcal{V} by

$$\mathcal{V}_{\Omega}^{h} = \operatorname{span}\left\{ (\psi_{m,\Omega}, \ \psi_{m,0}, \dots, \psi_{m,\boldsymbol{m}-1}) \in \mathcal{V} : \psi_{m,\Omega} = \psi_{m}, \\ \psi_{m,n} = \psi_{m}|_{\Gamma_{n}}, \ n \in \{0, \dots, \boldsymbol{m}-1\}, \ m \in \{1, \dots, M_{\Omega}\} \right\}.$$

A special case has to be made for the elements on the corners of Ω . Let $\psi_m \in H^1(\Omega)$ be such that $\psi_m(x) \neq 0$, where $x \in \partial \Omega$ is one of the corners of Ω . In this case, at least one of the traces

$$\psi_m|_{\Gamma_n}, n \in \{0,\ldots,m-1\}$$

will not be contained in $H^1(\Gamma_n)$. For these interior elements, we define $\psi_{m,n} \in H^1(\Gamma_n)$ as an $H^1(\Gamma_n)$ extension of

$$\psi_{m,n} := \psi_m|_{\Gamma_n^0}$$

So that we actually obtain a subspace of \mathcal{V} .

To discretise the boundaries, namely Γ_n^{σ} , $n \in \{0, \dots, m-1\}$, $\sigma \in \{\pm\}$, let us denote by $B_{\text{ife}} = \{\phi_m \in H_0^1(0, \infty) : m \in \{1, \dots, M_{\text{ife}}\}\}$ a basis of functions living on the positive real line, which will serve (after transformation) as a basis for the different semi-infinite lines. For this aim, let us define for $m \in \{1, \dots, M_{\text{ife}}\}$, $n \in \{0, \dots, m-1\}$ and $\sigma \in \{\pm\}$ the transformed basis function $\psi_{m,n}^{\sigma} \in \mathcal{V}$ as follows. On Γ_n^{σ} , $\psi_{m,n}^{\sigma}$ is given by

$$\psi_{m,n}^{\sigma}(\sigma z + c_n^{\sigma}) = \phi_n(z) \qquad \text{for } z > 0,$$

where we used the canonical identification (4.2.2). Elsewhere, we set $\psi_{m,n}^{\sigma} = 0$ (that is, in the interior of Ω and on the remaining semi-infinite lines). So we define for $n \in$ $\{0, \ldots, m-1\}, \sigma \in \{\pm\}$ the finite element ansatz space for Γ_n^{σ} by

$$\mathcal{V}_{\Gamma_n^{\sigma}}^h := \operatorname{span}\left\{\psi_{m,n}^{\sigma} : m \in \{1, \dots, M_{\operatorname{ife}}\}\right\}.$$

So that our full ansatz space will be

$$\mathcal{V}^h = \mathcal{V}^h_\Omega \oplus \bigoplus_{\substack{\sigma \in \{\pm\} \ n \in \{0, ..., \boldsymbol{m}-1\}}} \mathcal{V}^h_{\Gamma^\sigma_n}.$$

In Subsection 5.4.5, we will give explicit formulae for the basis B_{ife} .

5.4.2 Assembly of the matrices. Let us note that for $\psi, \phi \in \mathcal{V}^h$ the matrices associated with the volume part of the sesquilinear form $b_{\Omega}(\psi, \phi)$ can be computed by standard techniques. Accordingly, we will only describe how the matrices of the sesquilinear forms $b_n^{\mathcal{N}}$ and b_n^{σ} will be computed.

We fix some $n \in \{0, \ldots, m-1\}$, that is, we operate in the halfspace Ω_n . Let A be the associated operator with the generalised Fourier transform \mathcal{F}_A , and also recall the spectral space $L^2(\Lambda, d\mu)$ with the corresponding family of (generalised) eigenfunctions $\Psi : \Lambda \times \mathbb{R} \to \mathbb{C}$ (compare Chapter 2).

(a) DtN part $b_n^{\mathcal{N}}$. Let ψ_s , $\psi_t \in \mathcal{V}^h$ be two basis functions. We assume that $\psi_s|_{\Gamma_n}$ and $\psi_t|_{\Gamma_n^0}$ do not vanish, since elsewise $b_n^{\mathcal{N}}(\psi_s, \psi_t) = 0$. Let now $\phi_s := \psi_s|_{\Gamma_n}$ and $\phi_t := \psi_t|_{\Gamma_n^0}$. Utilising the canonical identification (4.2.2), we have to compute

$$b_n^{\mathcal{N}}(\psi_s, \psi_t) = -\int_{\Gamma_n^0} (\mathcal{N}\phi_s)\overline{\phi}_t \,\mathrm{d}s = -\int_{\mathbb{R}} (\mathcal{N}\phi_s)(x_1)\overline{\phi}_t(x_1) \,\mathrm{d}x_1.$$

5.4. DISCRETISATION

Denoting $\tilde{\phi}_s := \mathcal{F}_A \phi_s$ and $\tilde{\phi}_t = \mathcal{F}_A \tilde{\phi}_t$, we obtain by applying Lemma 3.4.2 to express \mathcal{N} , and by Fubini's theorem

$$-b_n^{\mathcal{N}}(\psi_s,\psi_t) = \int_{\mathbb{R}} \Psi(m,x_1) \int_{\Lambda} i\sqrt{\kappa - \hat{\lambda}(m)} \,\tilde{\phi}_{s}(m) \,\mathrm{d}\mu(m) \,\overline{\phi}_{t}(x_1) \,\mathrm{d}x_1$$
$$= \int_{\Lambda} i\sqrt{\kappa - \hat{\lambda}(m)} \,\tilde{\phi}_{s}(m) \int_{\mathbb{R}} \Psi(m,x_1) \overline{\phi}_{t}(x_1) \,\mathrm{d}x_1 \,\mathrm{d}\mu(m)$$
$$= \int_{\Lambda} i\sqrt{\kappa - \hat{\lambda}(m)} \,\tilde{\phi}_{s}(m) \overline{\tilde{\phi}_{t}(m)} \,\mathrm{d}\mu(m).$$

This formula helps us to reduce the problem of computing $b_n^{\mathcal{N}}(\psi_s, \psi_t)$ to two smaller sub-problems:

- (i) We have to calculate the transforms $\tilde{\phi}_s = \mathcal{F}_A \phi_s$, $\tilde{\phi}_t = \mathcal{F}_A \phi_t$. Later in Subsection 5.4.4, we will present explicit but somewhat complicated formulae for those quantities.
- (ii) We have to evaluate the integrals over the spectral space Λ , that is, we have to give some quadrature for the integral

$$\int_{\Lambda} f(m) \,\mathrm{d}\mu(m)$$

for an explicitly known function f. This will be the topic of Subsection 5.4.3.

(b) DtD part of b_n^{σ} : Fix $\sigma \in \{\pm\}$. Note that we can write

$$b_n^{\sigma}(\psi_{\mathbf{s}},\psi_{\mathbf{t}}) = \int\limits_{\Gamma_{n+1}^{\sigma}} [(I - \mathcal{D}_n^+)\psi_s]\overline{\psi}_t \,\mathrm{d}s = \int\limits_{\Gamma_{n+1}^{\sigma}} \psi_s \overline{\psi}_t \,\mathrm{d}s - \int\limits_{\Gamma_{n+1}^{\sigma}} [\mathcal{D}_n^+\psi_s]\overline{\psi}_t \,\mathrm{d}s$$

We discuss the two integrals on the right hand side separately. First consider the part involving the DtD operator \mathcal{D}_n^{σ} . Let ψ_s , $\psi_t \in \mathcal{V}^h$ be such that $\psi_s|_{\Gamma_n}$ and $\psi_s|_{\Gamma_{n+\sigma}^{\sigma}}$ do not vanish, since otherwise the corresponding integral would be zero. Let $\phi_s = \psi_s|_{\Gamma_n}$. We have to express $\psi_t|_{\Gamma_{n+\sigma}^{\sigma}}$ in the coordinate system of the halfspace Ω_n . For this purpose, let us define

$$\phi_{\mathbf{t}}(z) := \psi_{\mathbf{t}}(c_{n+\sigma}^{\sigma} + \sigma z) \quad \text{for } z > 0.$$

Furthermore, let us introduce the vector

$$d = \begin{pmatrix} d_1 \\ d_2 \end{pmatrix} := \begin{pmatrix} \cos(\theta_n^{-\sigma}) \\ \sin(\theta_n^{-\sigma}) \end{pmatrix},$$

where we refer again to Subsection 4.2.2, where we defined the angles θ_n^{σ} . This now

allows us to write (denoting $\tilde{\phi}_{s} = \mathcal{F}_{A}\phi_{s}$)

$$\int_{\Gamma_{n+1}^{\sigma}} (\mathcal{D}_{n}^{+}\psi_{s})\overline{\psi}_{t} \,\mathrm{d}s$$

$$= \int_{0}^{\infty} \int_{\Lambda} e^{i\sqrt{\kappa-\hat{\lambda}(m)} \,d_{2}z} \Psi(m, c_{n}^{-\sigma} + d_{1}z)\widetilde{\phi}_{s}(m) \,\mathrm{d}\mu(m) \,\overline{\phi}_{t}(z) \,\mathrm{d}z$$

$$= \int_{\Lambda} \widetilde{\phi}_{s}(m) \int_{0}^{\infty} e^{i\sqrt{\kappa-\hat{\lambda}(m)} \,d_{2}z} \Psi(m, c_{n}^{-\sigma} + d_{1}z)\overline{\phi}_{t}(z) \,\mathrm{d}z \,\mathrm{d}\mu(m)$$

This leaves us with another sub-problem: we have to compute the integral

$$\tilde{\phi}_{t}(z) := \int_{0}^{\infty} e^{i\sqrt{\kappa - \hat{\lambda}(m)} \, d_{2}z} \Psi(m, c_{n}^{-\sigma} + d_{1}z) \overline{\phi}_{t}(z) \, \mathrm{d}z$$
(5.4.1)

explicitly. This will be dealt with in Subsection 5.4.4.

(c) Identity part of b_n^{σ} : If ψ_s and ψ_t both have parts which are supported on $\Gamma_{n+\sigma}^{\sigma}$, we have to compute the integral

$$\int\limits_{\Gamma_{n+1}^{\sigma}} \psi_s \overline{\psi}_t \,\mathrm{d}s$$

This will be done explicitly when we describe our bases in Subsection 5.4.5.

5.4.3 The quadrature of the spectral space. Let $f : \Lambda \to \mathbb{C}$ be some function, $f = (f^p, f^+, f^-)$. We approximate the integral

$$\int_{\Lambda} f(m) \, \mathrm{d}\mu(m) = \sum_{n=1}^{N} f^p(\lambda_n) + \frac{1}{2\pi} \sum_{\sigma \in \{\pm\}} \int_{q_{\sigma}}^{\infty} f^{\sigma}(\lambda) \frac{\mathrm{d}\lambda}{2\sqrt{\lambda - q_{\sigma}}}$$

as follows. The computation of the finite sum is clear. For $\sigma \in \{\pm\}$, we obtain by substituting $\lambda = \xi^2 + q_\sigma$

$$\frac{1}{2\pi} \int_{q_{\sigma}}^{\infty} f^{\sigma}(\lambda) \frac{\mathrm{d}\lambda}{2\sqrt{\lambda - q_{\sigma}}} = \frac{1}{2\pi} \int_{0}^{\infty} f^{\sigma}(q_{\sigma} + \xi^2) \,\mathrm{d}\xi.$$

This integral is now approximated as follows: we truncate the integral at some maximal value $T_{\xi} > 0$, and divide the interval $(0, T_{\xi})$ into a number of subintervals. For each subinterval, we apply Gauss quadrature weights and knots. Since the integrands (as functions of λ) will depend on $\sqrt{\kappa - \lambda}$, $\sqrt{\lambda - q_-}$, $\sqrt{\lambda - q_+}$, which are not differentiable at certain points, we will choose the subintervals so that the discontinuities are boundary points of the subintervals. The critical points are given by

$$\xi \in \left\{\sqrt{\kappa - q_{\sigma}}, \sqrt{q_{+} - q_{\sigma}}, \sqrt{q_{-} - q_{\sigma}}\right\}$$

5.4. DISCRETISATION

5.4.4 Explicit formulae for transforms. To keep the presentation somewhat contained, we will only consider the integral from (5.4.1) here. The other integrals (in particular the transforms of the basis functions) can be calculated analogously. Let $\theta \in (0, \pi), c \in \mathbb{R}, d = (\cos(\theta), \sin(\theta))$, and let us consider

$$I(m) := \int_{0}^{\infty} e^{i\sqrt{\kappa - \hat{\lambda}(m)} \, d_2 z} \Psi(m, c + d_1 z) \overline{\phi}_{t}(z) \, \mathrm{d}z.$$

Let us make a few additional assumptions to avoid a tedious case-by-case analysis. Let a < b be the quantities from the definition of q (the potential of A, see Section 2.3). We now assume that $d_1 < 0$ and that c < a. This implies that $c + d_1 z < a$ for all z > 0, so that $\Psi(m, c + d_1 z)$ is given by a closed representation for any z > 0. Otherwise, the different cases of (2.3.1) have to be considered separately. Under these assumptions, we now have that

$$\Psi(m, c + d_2 z) = \alpha_1(m) e^{i\sqrt{\lambda - q_-}(c + d_1 z - a)} + \alpha_2(m) e^{-i\sqrt{\lambda - q_-}(c + d_1 z - a)}$$
 for $z > 0$,

where $\alpha_1, \alpha_2 : \Lambda \to \mathbb{C}$ are explicitly known functions (see Section 2.3). If we now introduce for $k \in \mathbb{C}$, $\operatorname{Re}(k) \leq 0$, the function

$$E(k) := \int_{0}^{\infty} e^{kz} \overline{\phi}_{t}(z) \, \mathrm{d}z.$$

we can easily compute

$$I(m) = \alpha_1(m) e^{i\sqrt{\hat{\lambda}(m) - q_-}(c-a)} E\left(i\sqrt{\kappa - \hat{\lambda}(m)} \, d_2 + i\sqrt{\hat{\lambda}(m) - q_-} \, d_1\right) + \alpha_1(m) e^{-i\sqrt{\hat{\lambda}(m) - q_-}(c-a)} E\left(i\sqrt{\kappa - \hat{\lambda}(m)} \, d_2 - i\sqrt{\hat{\lambda}(m) - q_-} \, d_1\right).$$

We will compute the function E explicitly for the different types of basis functions in the following Subsection 5.4.5.

5.4.5 Infinite element discretisation. We will define a few discretisations for the semi-infinite lines, and compute the required terms of the previous section. In effect, we only need to specify a basis B_{ife} , whose span is a sufficiently large subspace of $H_0^1(0,\infty)$. We will give two different choices in the following for the basis $B_{\text{ife}} = \{\phi_m \in H_0^1(0,\infty) : m \in \{1,\ldots,M_{\text{ife}}\}\}.$

FE-IFE. Our first basis is taken from [74], where it has been used in the very same context. We can truncate the semi-infinite line and discretise the functions on the bounded interval by first order finite elements. Let h > 0, then we choose for $m \in \{1, \ldots, M_{\text{ife}}\}$ the basis functions

$$\phi_m(z) = \begin{cases} \frac{x - (m-1)h}{h} & \text{for } z \in ((m-1)h, mh), \\ \frac{(m+1)h - x}{h} & \text{for } z \in (mh, (m+1)h), \\ 0 & \text{for } z \in (0, (m-1)h) \cup ((m+1)h, \infty). \end{cases}$$

From this, one easily computes

$$E_m(k) = \int_{\mathbb{R}} e^{kz} \overline{\phi}_m(z) \, \mathrm{d}z = \frac{4}{k^2 h} \sinh(\frac{kh}{2}) e^{mkh} \quad \text{for } k \in \mathbb{C},$$

as well as for $m, m' \in \{1, \ldots, M_{ife}\}$

$$\int_{0}^{\infty} \phi_m(z)\overline{\phi}_{m'}(z) \,\mathrm{d}z = \begin{cases} \frac{2}{3}h & \text{for } m = m', \\ \frac{1}{6}h & \text{for } |m - m'| = 1, \\ 0 & \text{else.} \end{cases}$$

FT-IFE. We propose a type of Fourier–Laplace infinite elements. Let $\gamma > 0$ and $\delta > 0$ be two arbitrary tuning parameters. We now choose for $m \in \{1, \ldots, M_{\text{ife}}\}$

$$\phi_m(z) = e^{-\gamma z} \sin(m\delta z).$$

Let us comment further on those two parameters: $\gamma > 0$ assures the finiteness of the integrals we will compute later. The parameter δ describes how fine we discretise in the "Fourier domain". Again, one easily calculates for $m, m' \in \{1, \ldots, M_{\text{ife}}\}$ and $k \in \mathbb{C}$ with $\text{Re}(k) < \gamma$

$$E_m(k) = \frac{\delta m}{(k-\gamma)^2 + \delta^2 m^2},$$
$$\int_0^\infty \phi_m(z)\overline{\phi}_{m'}(z) \,\mathrm{d}z = \frac{\gamma}{\delta^2 (m-m')^2 + 4\gamma^2} - \frac{\gamma}{\delta^2 (m+m')^2 + 4\gamma^2}.$$

5.4.6 Extension of corner elements. As mentioned in Subsection 5.4.1, we have to extend the interior elements living on the corners of $\partial\Omega$ to the adjacent semi-infinite lines Γ_n^{σ} . This will be done by an additional function $\phi_0 : [0, \infty) \to \mathbb{C}$, which fulfils $\phi_0(0) = 1$. We will choose this function corresponding to the rest of the IFE basis, and have to compute the corresponding integrals.

FE-IFE. In this case we choose

$$\phi_0(z) = \begin{cases} \frac{h-z}{h} & \text{for } z \in (0,h), \\ 0 & \text{else.} \end{cases}$$

Again, one easily computes

$$E_0(k) = \frac{1}{k^2 h} (e^{kh} - 1 - kh)$$

and for $m \in \{0, \ldots, M_{\text{ife}}\}$

$$\int_{0}^{\infty} \phi_0(z) \overline{\phi}_m(z) \, \mathrm{d}z = \begin{cases} \frac{1}{3}h & \text{for } m = 0, \\ \frac{1}{6}h & \text{for } m = 1, \\ 0 & \text{else.} \end{cases}$$

FT-IFE. Here we set

$$\phi_0(z) = e^{-\gamma z} \qquad \text{for } z > 0.$$

Then

$$E_0(k) = \frac{1}{k+\gamma},$$

$$\int_0^\infty \phi_0(z)\overline{\phi}_m(z) \,\mathrm{d}z = \begin{cases} \frac{1}{2\gamma} & \text{for } m = 0, \\ \frac{\delta m}{2^2\gamma^2 + \delta^2 m^2} & \text{for } m \ge 1. \end{cases}$$

5.4.7 Implementation and convergence tests. The method we just described was implemented with first order elements for the interior in MATLAB, and it turned out that this restriction to first order elements was a rather bad choice, since it significantly inhibits convergence. Nonetheless we will show a few results, but keep this section very brief, and restrict ourselves to the non-absorptive case including incident waves. For a more detailed study of the numerical convergence in the case of an unperturbed free space, we refer to Tonnoir's PhD thesis [74].

We will test the two different discretisations FT-IFE and FE-IFE on two examples of scattering problems of the type of Section 5.3. For these examples we know the exact solution, and will try to reconstruct it by our method. The first benchmark is the same problem as sketched in Example 5.3.5: a mode propagating in a simple open waveguide. The solution we expect to obtain is exactly the mode without any disturbance. The second benchmark situation is a homogeneous space, with a plane wave incident on a triangular finite element domain. Again, we expect the plane wave to propagate through our domain, which will be our reference solution.

The error we show is the relative L^2 -error on a square $\hat{\Omega}$, which contains the finite element domain Ω as well as parts of the surrounding halfspaces. Outside of the finite element domain, we used the solution operators S_n to compute the solution.



Figure 5.4: The two test situations for the convergence test. The coloured field shows the real part of the incident field, which is only supported on the top halfspace. They are straightforwardly extendable to the whole \mathbb{R}^2 , which serves as the reference solution. As before, the grey area indicates the waveguide.

FE-IFE. There are five parameters that influence the convergence:

(1) the mesh size of the finite element discretisation in the interior,



Figure 5.5: Convergence for the FE-IFE. The black lines are guides for the eye, to give orientation of the order of convergence. Note that in particular for the free space case, there is no proper convergence.

- (2) the truncation of the integration interval for the quadrature of the spectral integrals (see Subsection 5.4.3),
- (3) the order and number of subintervals for the quadrature,
- (4) the mesh size h of discretisation of the infinite elements,
- (5) the truncation of the semi-infinite lines, which is given by

$$T_{\text{ife}} = M_{\text{ife}}h,$$

where we recall that M_{ife} is the number of infinite elements.

This is a rather large parameter space we should cover, but for the sake of brevity we will restrict ourselves to showing some results regarding point (4) and (5), while we will deal with (1) to (3) just in a few remarks.

The convergence with respect of the mesh-size in the interior is well-analysed, and we omit any analysis here. For the integration parameters, one needs to choose the truncation distance T_{ξ} larger than

$$T_{\xi} > \sqrt{\|\kappa_0 - q\|_{L^{\infty}}}.$$

[74] found that the error depends exponentially on $T_{\xi} - \sqrt{\|\kappa_0 - q\|_{L^{\infty}}}$, which we can confirm. The number of subintervals was chosen as 500, each one integrated by a (Gauss) quadrature of order 5. Increasing the order did not significantly reduce the number of function evaluations necessary. We suspect, however, that in particular the resolution close to the critical points of discontinuity (see Subsection 5.4.3) is of importance. The convergence with respect to the remaining two parameters is shown in Figure 5.5. One notes that the convergence for the free space plane wave is particularly bad, which is actually not very surprising, since each of the traces we try to retrieve is not decaying at all. For the waveguide case, we see a reasonable polynomial convergence with respect to both the truncation and the mesh size.



Figure 5.6: Convergence for the FT-IFE discretisation of the semi-infinite lines. The convergence seems to be faster than the FE-IFE. In particular the convergence with respect to δ seems to be exponential, but flattens out due to residual error from the finite elements in the interior.

FE-IFE. Let us quickly have a look at the second type of discretisation, the Fourier–Laplace infinite elements. Here, the parameters (4) and (5) from the previous list are replaced by

- (4) the mesh-size in the Fourier domain δ ,
- (5) the truncation in Fourier domain,

$$T = M_{\text{ife}}\delta.$$

We omitted the parameter γ , which was chosen as 0.1 for the examples shown.

The corresponding convergence results are shown in Figure 5.6. Let us point here to the fact that these infinite elements seem to provide a higher order of convergence, which is not easy to quantify. One main reason is the limiting influence of the finite elements in the interior, which proved to be a bottleneck for this type of discretisation (for the waveguide case).

5.4.8 Comparison to other transparent boundary conditions. During the introduction in Subsection 1.1.3, we mentioned two other methods which in principle allow dealing with the same problem as ours: the perfectly matched layer (PML) and the Hardy space infinite elements (HSIE). How do these three methods compare?

In the authors opinion, our method is at a clear disadvantage, due to a number of reasons.

• Halfspace matching is much more complicated in terms of implementation than the other two methods. It requires the explicit availability of the generalised Fourier transform, which is only given for very particular cases. Furthermore, the assembly of the different types of matrices is rather complicated.

- Related to this is the second, severe drawback: each matrix entry is computed by the numerical evaluation of an integral. For a fine discretisation of the semiinfinite lines (i.e. for small δ or h), one needs to carefully increase the accuracy of the quadrature, additionally increasing the cost of the matrix assembly, making it very costly compared to the other methods.
- A third disadvantage is the lower rate of convergence we obtain. For PML and HSIE, the convergence in the non-absorptive case is effectively exponential [63] with respect to the number of exterior degrees of freedom. As seen above, this kind of convergence is only partly shared by our method, and due to the high number of discretisation parameters, there is some balancing required to obtain good results.
- Lastly, our method imposes a number of additional restrictions on the geometrical arrangements of the waveguides in the exterior, which is not shared by either PML or HSIE.

However, it should be mentioned, that our method also has a few advantages.

- For incident waves, our problem formulation allows for very close cropping of the interior domain (see Figure 5.4). Since the mode far outside the interior is incorporated, one can practically cut directly at the waveguide. This is not shared for other methods, since the in-coupling of the incident mode is usually done by a Dirichlet and Neumann jump.
- Our method gives easy access to the solution outside of the interior domain.
- The additional degrees of freedom introduced by our method do not depend on the size of the interior domain, but only on the number of corners. For PML and HSIE, the number of degrees of freedom to realise the transparent boundary condition is rougly proportional to the circumference of the interior domain.
6 Optimisation of Waveguide Junctions

6.1 Introduction

In this chapter we want to illustrate the physical background behind the method developed in the last chapters. To do this, we will study scattering problems at waveguide junctions, and try to manipulate them to create a special scattering behaviour. To be able to introduce the calculus required for the optimisation, we will again make a detour into the absorptive world, so the outline of this chapter is as follows.

We will start by introducing a formulation for incident waves in the absorptive case, to get an analogous problem to the non-absorptive problem from Section 5.3. This will allow us to study the differentiability properties of the solution as a function of the potential q, and gives rise to an interesting property of waveguide junctions. If we try to optimise the transmission into a particular mode, the resulting adjoint state will take a very peculiar form: it is again a guided incident mode.

The required derivatives will be introduced in Section 6.3, where we will keep up with a rigorous mathematical definition-lemma style. This will end with the next section, where we will switch back to the non-absorptive case and formal calculations. To give a proper meaning to the *energy flow* associated with some mode, we will derive an energy conservation relation for junctions of open waveguides in Section 6.4. In Section 6.5, we will describe the optimisation algorithm we will use in the examples afterwards.

Let us again clarify the scope of this chapter: it is not to give a thorough analysis of the optimisation algorithms or an extensive testing thereof, but rather to illustrate the physics of our problem.

6.1.1 References. Shape, material, and topology optimisation are a large field, and we will give only a few general references. The general method we use originates from Bendsøe and Kikuchi's papar [3]. A large number of examples and a general introduction into topology/material optimisation can be found in Bendsøe's and Sigmund's book [4] and the references therein, which is, however, more application oriented. The notation and approach here is more heavily influenced by shape optimisation, for which we reference [26], as well as by inverse problem scattering, for which we reference to [41, 50].

6.2 Incident Fields in the Absorptive Case

To give a more precise treatment of the shape calculus and the adjoint states, we will switch back to a scattering problem with absorption, that is, we choose again

$$\kappa \in \mathbb{C} \setminus \mathbb{R}$$

and let $q \in L^{\infty}(\mathbb{R}^2)$ be some arbitrary real-valued potential. We start with an incident field u_i that solves

$$\Delta u_{\rm i} + (\kappa - q)u_{\rm i} = 0 \qquad \text{in } \Omega_0,$$

where Ω_0 is a halfspace. We furthermore assume that $u_i|_{\Gamma_0} \in H^{1/2}(\Gamma_0)$, where $\Gamma_0 = \partial \Omega_0$ as before. These assumptions are already rather restrictive for u_i . However, for example a guided mode of the form

$$u_{\mathbf{i}}(x) = e^{-i\sqrt{\kappa - \lambda_n} x_2} \Psi^p(\lambda_n, x_1)$$

fulfils these conditions on $\Omega_0 := \{x \in \mathbb{R}^2 : x_2 < 0\}$, provided q is chosen appropriately. Note that u_i is exponentially increasing as $x_2 \to -\infty$.

One can easily repeat the following procedure solely under the condition that u_i fulfils the Helmholtz equation in the proximity of Γ_0 , so that sources contained in Ω_0 can be incorporated in u_i as well.

We aim to give meaning to the problem

$$\begin{cases} \Delta u + (\kappa - q)u = 0 & \text{ in } \mathbb{R}^2, \\ u - u_{i} & \text{ outgoing in } \mathbb{R}^2, \end{cases}$$
(6.2.1)

in a variational setting. We start by slightly redefining our incident field as $\tilde{u}_i := u_i - w$, where $w \in H^1(\Omega_0)$ solves

$$\begin{cases} \Delta w + (\kappa - q)w = 0 & \text{ in } \Omega_0, \\ w = u_{\mathbf{i}} & \text{ on } \Gamma_0, \end{cases}$$

in the variational sense. w exists and is unique by Theorem 1.4.3. The new incident field \tilde{u}_i has the advantage that it vanishes on Γ_0 , so that we expect that the scattered field, defined by

$$u_{\mathbf{s}} := \begin{cases} u - \tilde{u}_{\mathbf{i}} & \text{in } \Omega_0, \\ u & \text{in } \mathbb{R}^2 \setminus \overline{\Omega}_0 \end{cases}$$

will be in $H^1(\mathbb{R}^2)$: we choose to demand that $u_s \in H^1(\Omega_0)$ as well as $u_s \in H^1(\mathbb{R}^2 \setminus \overline{\Omega}_0)$, since this has turned out to give a notion of "outgoing" in the absorptive case. Furthermore, since \tilde{u}_i vanishes on Γ_0 , we obtain that the left-sided and right-sided traces on Γ_0 of u_s agree, which implies that u_s is in $H^1(\mathbb{R}^2)$.

Let us derive a variational formulation for u_s . Since $u \in H^2_{loc}(\mathbb{R}^2)$, we can multiply the Helmholtz equation (6.2.1) by some $\psi \in C_0^{\infty}(\mathbb{R}^2)$, and integrate over $\operatorname{supp}(\psi)$ to obtain by partial integration

$$\begin{split} 0 &= \int_{D} -\nabla u \cdot \nabla \overline{\psi} + (\kappa - q) u \overline{\psi} \, \mathrm{d}x \\ &= \int_{\Omega_0 \cap D} -\nabla (u_\mathrm{s} + \tilde{u}_\mathrm{i}) \cdot \nabla \overline{\psi} + (\kappa - q) (u_\mathrm{s} + \tilde{u}_\mathrm{i}) \overline{\psi} \, \mathrm{d}x \\ &+ \int_{(\mathbb{R}^2 \setminus \Omega_0) \cap D} -\nabla u_\mathrm{s} \cdot \nabla \overline{\psi} + (\kappa - q) u_\mathrm{s} \overline{\psi} \, \mathrm{d}x, \end{split}$$

which can be rearranged to get

$$\int_{\mathbb{R}^2} \nabla u_s \cdot \nabla \overline{\psi} - (\kappa - q) u \overline{\psi} \, \mathrm{d}x = \int_{\Omega_0 \cap D} -\nabla \widetilde{u}_i \cdot \nabla \psi + (\kappa - q) \widetilde{u}_i \overline{\psi} \, \mathrm{d}x$$
$$= \int_{\Gamma_0} (\partial_\nu \widetilde{u}_i) \overline{\psi} \, \mathrm{d}s,$$

where we used that u_i solves the Helmholtz equation in the last step. Let us reformulate (6.2.1) with a sesquilinear form.

6.2.1 Weak problem. Let $\tilde{u}_i \in H^2_{loc}(\Omega_0)$ be some incident field such that $\tilde{u}_i|_{\Gamma_0} = 0$. We say that u is a variational solution to (6.2.1), if

$$u = \begin{cases} u_{\rm s} + \tilde{u}_{\rm i} & \text{on } \Omega_0, \\ u_{\rm s} & \text{on } \mathbb{R}^2 \setminus \overline{\Omega}_0, \end{cases}$$

where $u_{\rm s} \in H^1(\mathbb{R}^2)$ is the unique solution of

$$\int_{\mathbb{R}^2} \nabla u_{\mathbf{s}} \cdot \nabla \overline{\psi} - (\kappa - q) u_{\mathbf{s}} \overline{\psi} \, \mathrm{d}x = \int_{\Gamma_0} (\partial_{\nu} \tilde{u}_{\mathbf{i}}) \overline{\psi} \, \mathrm{d}s \quad \text{for all } \psi \in H^1(\mathbb{R}^2).$$

This variational problem can be shown to be uniquely solvable by the Lax–Milgram theorem, as in the proof of Theorem 1.4.3, since the right hand side is a bounded conjugate linear functional on $H^1(\mathbb{R}^2)$ by the trace theorem.

Let us also comment on the meaning of the problem we just derived: essentially, we implemented an incident wave by a Neumann jump of the scattered field u_s . Let us now formulate the underlying conjecture, which will allow us to translate the result to the absorption free case, i.e. $\kappa_0 \in \mathbb{R}$.

6.2.2 Conjecture. Let $q : \mathbb{R}^2 \to \mathbb{R}$ fulfil Assumption 5.1.2, and let Ω_0 contain a waveguide. Consider the incident field

$$\tilde{u}_{i}(\xi\xi^{(0)} + \eta\eta^{(0)}) = \left(e^{-i\sqrt{\kappa - \lambda_{l}}\eta} - e^{i\sqrt{\kappa - \lambda_{l}}\eta}\right)\Psi^{(p)}(\lambda_{l},\xi) \qquad \xi \in \mathbb{R}, \ \eta > 0,$$

where λ_l and $\Psi^{(p)}$ denotes an eigenvalue and eigenfunction of the operator A associated with Ω_0 (also note that we heavily employ the notation of Subsection 4.2.2), which fulfils the requirement of the previous subsection. Fix now $\kappa_0 \in \mathbb{R}$, and let $\epsilon > 0$ and set

$$\kappa = \kappa_0 + i\epsilon.$$

Denote by u_{ϵ} the solution of

$$\begin{cases} \Delta u_{\epsilon} + (\kappa_0 + i\epsilon - q)u_{\epsilon} = 0 & \text{in } \mathbb{R}^2, \\ u_{\epsilon} - u_{i} & \text{is outgoing,} \end{cases}$$

which has to be understood in the variational sense. Let furthermore u_0 denote the solution of the last problem for $\epsilon = 0$, which is understood in the sense of Subsection 5.3.6. We now conjecture that

$$u_{\epsilon} \to u_0$$
 as $\epsilon \to 0, \epsilon > 0$ in $H^2_{\text{loc}}(\mathbb{R}^2)$.

Let us point again to the references at the end of Section 1.3, where we already discussed the limit-absorption principle in open waveguides. Note however, that with the somewhat unsure formulation we use in our non-absorptive framework, there are still large gaps to be filled.

Note that this assumption holds for the scattering problem of a straight mode (see Example 5.3.5).

For the remainder of this chapter we will proceed as follows: we will develop material optimisation on the level of the absorptive, variational problem, and in the end pull all results over to the non-absorptive case in a rotating-arm argument¹.

6.3 The Fréchet Derivative

In this section we will consider the dependence of the solution on the potential q. To reflect the change of scope, we will introduce a bit of notation.

6.3.1 Definition. Let us define the space

$$H^1(\mathbb{R}^2)^* := \{F : H^1(\mathbb{R}^2) \to \mathbb{C} : F \text{ conjugate linear and continuous}\}$$

which we equip with the norm

$$||F||_{H^1(\mathbb{R}^2)^*} := \sup_{u \in H^1(\mathbb{R}^2)} |F(u)|.$$

Furthermore we denote by

$$L^{\infty}(\mathbb{R}^2,\mathbb{R}) = \{ f : \mathbb{R}^2 \to \mathbb{R} : \|f\|_{L^{\infty}(\mathbb{R}^2)} < \infty \},\$$

the space of real-valued L^{∞} -functions on \mathbb{R}^2 . For $q \in L^{\infty}(\mathbb{R}^2, \mathbb{R})$, we define the sesquilinear form $b_q : H^1(\mathbb{R}^2) \times H^1(\mathbb{R}^2) \to \mathbb{C}$ by

$$b_q(u,v) := \int_{\mathbb{R}^2} \nabla u \cdot \nabla \overline{\psi} - (\kappa - q) u \overline{\psi} \, \mathrm{d}x.$$

We recall a few facts, which fall out of the proof of Theorem 1.4.3, and give the following corollary.

6.3.2 Corollary. Let $\kappa \in \mathbb{C} \setminus \mathbb{R}$, $q \in L^{\infty}(\mathbb{R}^2, \mathbb{R})$. Then there exist constants $0 < c_1 = c_1(\|q\|_{L^{\infty}(\mathbb{R})}, \kappa)$ and $0 < c_2 = c_2(\|q\|_{L^{\infty}(\mathbb{R})}, \kappa)$, which only depend on $\|q\|_{L^{\infty}(\mathbb{R})}$ and κ , such that

$$\begin{aligned} |b_q(u,v)| &\leq c_1 ||u||_{H^1(\mathbb{R}^2)} ||v||_{H^1(\mathbb{R}^2)} & \text{for all } u, v \in H^1(\mathbb{R}^2) \\ |b_q(u,u)| &\geq c_2 ||u||_{H^1(\mathbb{R}^2)}^2 & \text{for all } u \in H^1(\mathbb{R}^2). \end{aligned}$$

For any $F_{s} \in H^{1}(\mathbb{R}^{2})^{*}$, there exists exactly one solution u_{q} of

$$b_q(u,\psi) = F_{\mathbf{s}}(\psi) \quad \text{for all } \psi \in H^1(\mathbb{R}^2),$$

¹In physics, a "hand-waving proof" is an argument, which seems somewhat plausible, but is actually build on very shaky assumptions. A "very hand-waving" argument is sometimes raised to be a "rotatingarm" argument for obvious reasons. Whether another comparative to rotating-arm exists is not known to the author.

and it holds that

$$||u_q||_{H^1(\mathbb{R}^2)} \le \frac{1}{c_2} ||F_{\mathbf{s}}||_{H^1(\mathbb{R}^2)^*}.$$
(6.3.1)

Proof. Rechecking the proof of Theorem 1.4.3, the bounds are easily obtained. The existence of the solution for any functional follows by the Lax–Milgram lemma, and the bound for the $H^1(\mathbb{R}^2)$ norm follows by exploiting the coerciveness of b_q .

6.3.3 Definition. Fix $\kappa \in \mathbb{C} \setminus \mathbb{R}$ and $F_s \in H^1(\mathbb{R}^2)^*$. We define the potential-to-solution operator

$$\mathcal{L}: L^{\infty}(\mathbb{R}^2, \mathbb{R}) \to H^1(\mathbb{R}^2), \qquad \mathcal{L}(q) := u_q,$$

where $u_q \in H^1(\mathbb{R}^2)$ is the solution of

$$b_q(u,\psi) = F_{\mathrm{s}}(\psi) \quad \text{for all } \psi \in H^1(\mathbb{R}^2).$$

We now aim to take the Fréchet-derivative of the operator \mathcal{L} . It will be denoted by $d\mathcal{L}: L^{\infty}(\mathbb{R}^2, \mathbb{R}) \times L^{\infty}(\mathbb{R}^2, \mathbb{R}) \to H^1(\mathbb{R}^2)$. Since it is non-linear in the first and linear in the second argument, we choose the notation

$$d\mathcal{L}(q)p \in H^1(\mathbb{R}^2)$$

for the derivative at q in direction p.

6.3.4 Lemma. Fix $\kappa \in \mathbb{C} \setminus \mathbb{R}$, $F_{s} \in H^{1}(\mathbb{R}^{2})^{*}$. Then the Fréchet-derivative of \mathcal{L} at $q \in L^{\infty}(\mathbb{R}^{2}, \mathbb{R})$, denoted by $d\mathcal{L}(q) : L^{\infty}(\mathbb{R}^{2}, \mathbb{R}) \to H^{1}(\mathbb{R}^{2})$, is given by

$$d\mathcal{L}(q)p = u'_p,$$

where $u'_p \in H^1(\mathbb{R}^2)$ is the solution of

$$b_q(u'_p, \psi) = -\int_{\mathbb{R}^2} p u_q \overline{\psi} \, \mathrm{d}x \quad \text{for all } \psi \in H^1(\mathbb{R}^2),$$

where $u_q = \mathcal{L}(q)$.

Proof. Let $q \in L^{\infty}(\mathbb{R}^2, \mathbb{R})$ be fixed for the following, and let us define the ball with radius 1 around 0 by

$$B_1 := \left\{ \tilde{p} \in L^{\infty}(\mathbb{R}^2, \mathbb{R}) : \|\tilde{p}\|_{L^{\infty}(\mathbb{R}^2)} < 1 \right\}.$$

First note that u'_p depends linearly and continuously on p, which are the usual properties of the derivative. To prove differentiability, we will show that there exists a constant C > 0 such that

$$||u_{q+p} - u_q - u'_p||_{H^1(\mathbb{R}^2)} < C ||p||^2_{L^{\infty}(\mathbb{R}^2)} \quad \text{for all } p \in B_1 \setminus \{0\},$$

where $u_{p+q} = \mathcal{L}(p+q)$. Let us define $C_2 > 0$ by

$$C_2 := \inf \left\{ c_2(\|q + \tilde{p}\|_{L^{\infty}(\mathbb{R}^2)}, \kappa) : \tilde{p} \in B_1 \right\},\$$

where $c_2(||q + \tilde{p}||_{L^{\infty}(\mathbb{R}^2)}, \kappa)$ is the constant from Corollary 6.3.2. One now easily obtains by the definition of b_q that

$$b_q(u_{q+p} - u_q, \psi) = -\int\limits_{\mathbb{R}^2} p \, u_{q+p} \overline{\psi} \, \mathrm{d}x.$$

Setting $\psi = u_{q+p} - u_q$ in the last equation and exploiting (6.3.1) as well as the Cauchy–Schwarz inequality now yields

$$\begin{aligned} \|u_{q+p} - u_q\|_{H^1}^2 &\leq \frac{1}{C_2} \left| \int_{\mathbb{R}^2} p u_{q+p} (u_{q+p} - u_q) \, \mathrm{d}x \right| \\ &\leq \frac{1}{C_2} \|p\|_{L^{\infty}} \|u_{p+q}\|_{H^1} \|u_{q+p} - u_q\|_{H^1} \\ &\leq \frac{\|F\|_{H^1(\mathbb{R}^2)^*}}{C_2^2} \|p\|_{L^{\infty}} \|u_{q+p} - u_q\|_{H^1}, \end{aligned}$$

where we omitted the domain of the spaces (which is always \mathbb{R}^2) for better readability. Dividing by the norm of $u_{q+p} - u_q$ yields

$$\|u_{q+p} - u_q\|_{H^1} \le \frac{\|F\|_{H^1(\mathbb{R}^2)^*}}{C_2^2} \|q\|_{L^{\infty}}.$$
(6.3.2)

Let $\psi \in H^1(\mathbb{R}^2)$, then we obtain by the definitions of u_{q+p} , u_q and u'_p

$$\begin{split} b_q(u_{q+p} - u_q - u'_p, \psi) \\ &= b_{q+p}(u_{q+p}, \psi) - \int_{\mathbb{R}^2} p \, u_{q+p} \overline{\psi} \, \mathrm{d}x - b_q(u_q, \psi) - b_q(u'_p, \psi) \\ &= F(\psi) - \int_{\mathbb{R}^2} p \, u_{q+p} \overline{\psi} \, \mathrm{d}x - F(\psi) + \int_{\mathbb{R}^2} p \, u_q \overline{\psi} \, \mathrm{d}x \\ &= - \int_{\mathbb{R}^2} p(u_{q+p} - u_q) \overline{\psi} \, \mathrm{d}x. \end{split}$$

Setting $\psi = u_{q+p} - u_q - u'_p$, we obtain by (6.3.2), (6.3.1), and the Cauchy–Schwarz inequality

$$\begin{aligned} \|u_{q+p} - u_q - u'_p\|_{H^1}^2 &\leq \frac{1}{C_2} \left| b_q (u_{q+p} - u_q - u'_p, u_{q+p} - u_q - u'_p) \right| \\ &\leq \frac{1}{C_2} \left| \int_{\mathbb{R}^2} p(u_{q+p} - u_q) (u_{q+p} - u_q - u'_p) \, \mathrm{d}x \right| \\ &\leq \frac{1}{C_2} \|p\|_{L^{\infty}} \|u_{q+p} - u_q\|_{H^1} \|u_{q+p} - u_q - u'_p\|_{H^1} \\ &\leq \frac{\|F_s\|_{H^1(\mathbb{R}^2)^*}}{C_2^3} \|p\|_{L^{\infty}}^2 \|u_{q+p} - u_q - u'_p\|_{H^1}. \end{aligned}$$

Dividing the last equation by the H^1 -norm of $u_{q+p} - u_q - u'_p$ now gives the estimate we set out to prove in the beginning, finishing the proof.

In order to optimise, we need a figure of merit (abbreviated as FOM). Let us define the class of functionals we will consider.

6.3.5 Definition and Lemma. Let $F_t \in H^1(\mathbb{R}^2)^*$. We will consider figures of merit $J: H^1(\mathbb{R}) \to \mathbb{R}$, given by

$$J(u) := |F_{t}(u)|^{2}.$$

It holds in the sense of Fréchet derivatives for $u, v \in H^1(\mathbb{R}^2)$

$$dJ(u)v = 2\operatorname{Re}\left(F_{t}(u)\overline{F_{t}(v)}\right).$$

Proof. Note that J is not complex-differentiable, and hence one has to interpret previous derivative as a real derivative, that is, one has to identify $H^1(\mathbb{R}^2, \mathbb{C}) \cong H^1(\mathbb{R}^2, \mathbb{R}^2)$ and take the real linear structure of the latter space. The proof gets rather trivial by writing $J(u) = [\operatorname{Im}(F_t(u))]^2 + [\operatorname{Re}(F_t(u))]^2$, noting that the two functionals $\operatorname{Im}(F_t(u))$ and $\operatorname{Re}(F_t(u))$ are real-linear and real-differentiable.

Our goal is to determine the Fréchet-derivative of the potential-to-FOM function

$$\mathcal{J}: L^{\infty}(\mathbb{R}^2, \mathbb{R}) \to \mathbb{R}, \qquad \mathcal{J}(q) := J(\mathcal{L}(q)).$$

By the chain rule, one obtains

$$d\mathcal{J}(q)p = d(J \circ \mathcal{L})(q)p = dJ(\mathcal{L}(q)) \, d\mathcal{L}(q) \, p$$

= 2 Re $\left(F_{t}(u_{q})\overline{F_{t}(u'_{p})}\right),$ (6.3.3)

where we used the previous Definition and Lemma 6.3.5 as well as Lemma 6.3.4. Let us now consider that we want to calculate the derivative of \mathcal{J} with respect to a number of directions $p_1, \ldots, p_N \in L^{\infty}(\mathbb{R}^2, \mathbb{R})$: for each p_n , we have to calculate u'_{p_n} , that is, we have to solve a large linear system of equations. To compute all derivatives, one needs to solve in total N + 1 large systems. With the help of the adjoint state, this can be reduced to solving two systems.

6.3.6 Definition. Fix $\kappa \in \mathbb{C} \setminus \mathbb{R}$, F_s , $F_t \in H^1(\mathbb{R}^2)^*$, and let $q \in L^{\infty}(\mathbb{R}^2, \mathbb{R})$. Denote $u_q = \mathcal{L}(q)$. We now define the adjoint state w_q by

$$b_q(w_q,\psi) = 2F_t(u_q)\overline{F_t(\overline{\psi})}$$
 for all $\psi \in H^1(\mathbb{R}^2)$.

Note that due to the double conjugation, the right hand side is still conjugate linear in ψ .

6.3.7 Lemma. Let $\kappa \in \mathbb{C} \setminus \mathbb{R}$, F_{s} , $F_{t} \in H^{1}(\mathbb{R}^{2})^{*}$. Then we have for all $q, p \in L^{\infty}(\mathbb{R}^{2}, \mathbb{R})$

$$d\mathcal{J}(q)p = -\int_{\mathbb{R}^2} p \operatorname{Re}\left(u_q w_q\right) \, \mathrm{d}x,$$

where $u_q = \mathcal{L}(q)$ and w_q is the adjoint state at q.

Proof. Let us start by noting the following symmetry relation for b_q . We have

$$b_q(u, v) = b_q(\overline{v}, \overline{u}) \quad \text{for } u, v \in H^1(\mathbb{R}^2).$$

Let $u'_p = d\mathcal{L}(q)p$ be the derivative of \mathcal{L} in direction of p. Setting $\psi = \overline{u}'_p$ in the sesquilinear form for w_q we obtain

$$2F_{t}(u_{q})\overline{F_{t}(u_{p}')} = b_{q}(w_{q},\overline{u}_{p}') = b_{q}(u_{p}',\overline{w}_{q}) = -\int_{\mathbb{R}^{2}} p \, u_{q} w_{q} \, \mathrm{d}x,$$

where we used the definition the sesquilinear form for u'_p (see Lemma 6.3.4) at the last step. Taking the real part yields by (6.3.3) that

$$d\mathcal{J}(q)p = -2\operatorname{Re}\left(F_{t}(u_{q})\overline{F_{t}(u_{p}')}\right) = -\operatorname{Re}\left(\int_{\mathbb{R}^{2}}p\,u_{q}w_{q}\,\mathrm{d}x\right),$$

finishing the proof.

Let us also state (without proof) the second derivative, which we later use to determine the step width in iterative optimisation schemes.

6.3.8 Lemma. Let $\kappa \in \mathbb{C} \setminus \mathbb{R}$, F_s , $F_t \in H^1(\mathbb{R}^2)^*$ and $q, p_1, p_2 \in L^{\infty}(\mathbb{R}^2, \mathbb{R})$. Then the second Fréchet derivative $d^2 \mathcal{J}(q) : L^{\infty}(\mathbb{R}^2, \mathbb{R}) \times L^{\infty}(\mathbb{R}^2, \mathbb{R}) \to \mathbb{R}$ is given by

$$d^{2}\mathcal{J}(q)[p_{1},p_{2}] = 2\operatorname{Re}\left(F_{t}(u_{p_{1},p_{2}}'')\overline{F_{t}(u_{q})}\right) + 2\operatorname{Re}\left(F_{t}(u_{p_{1}}')\overline{F_{t}(u_{p_{2}}')}\right)$$

where $u_{p_1,p_2}' \in H^1(\mathbb{R}^2)$ is the unique solution of

$$b_q(u_{p_1,p_2}'',\psi) = -\int\limits_{\mathbb{R}^2} p_1 u_{p_2}' \overline{\psi} \, \mathrm{d}x - \int\limits_{\mathbb{R}^2} p_2 u_{p_1}' \overline{\psi} \, \mathrm{d}x \quad \text{for all } \psi \in H^1(\mathbb{R}^2),$$

where $u'_{p_1} = d\mathcal{L}(q)p_1$, $u'_{p_2} = d\mathcal{L}(q)p_2$.

Proof. We omit the proof here, but remark that not many additional arguments compared to the proof of the first derivative are necessary.

6.4 Energy Conservation

We now pass back to a more specific example of waveguide junctions.

6.4.1 Setting. Let us fix $\kappa \in \mathbb{C} \setminus \mathbb{R}$, and let q fulfil Definition and Assumption 5.1.2. Let Ω denote the corresponding interior domain, and let $\Omega_0, \ldots, \Omega_{m-1}$ be the overlapping exterior halfspaces. For $n \in \{0, \ldots, m-1\}$, $k \in \{1, \ldots, N_n\}$, we denote by $\lambda_k^{(n)}$ the kth eigenvalue of $A_n = -\Delta + q_n$, with the eigenfunction $\Psi_n^p(\lambda_k^{(n)}, \cdot)$.

6.4.2 Definition. For $n \in \{0, \ldots, m-1\}$ and $k \in \{1, \ldots, N_n\}$, we define the amplitude functional $F_{n,k} \in H^1(\mathbb{R}^2)^*$ by

$$F_{n,k}(\psi) = \int_{\Gamma_n} \Psi_n^p(\lambda_k^{(n)}, \cdot) \overline{\psi} \, \mathrm{d}s.$$

We name it amplitude functional, since it yields the amplitude of the kth guided mode in the field on the halfspace Ω_n : if one unwraps the definition of the solution operator S_n , including the different coordinate systems, one obtains that the sum over the eigenmodes reads as follows

$$S_n^{\text{guided}} g(\xi \xi^{(n)} + (\eta + h_n)\eta^{(n)})$$

= $\sum_{k=1}^{N_n} e^{i\sqrt{\kappa - \lambda_k^{(n)}} \eta} \Psi_n^p(\lambda_k^{(n)}, \xi) \int_{\Gamma_n} \Psi_k^p(\lambda_k^{(n)}, \cdot) g \, \mathrm{d}s$
= $\sum_{k=1}^{N_n} e^{i\sqrt{\kappa - \lambda_k^{(n)}} \eta} \Psi_n^p(\lambda_k^{(n)}, \xi) \overline{F_{n,k}(g)},$

where $\eta > 0, \xi \in \mathbb{R}$. These functionals will come in handy later.

6.4.3 Lemma. For $n \in \{0, ..., m-1\}$ and $k \in \{1, ..., N_n\}$ let \tilde{u}_i be the incident field corresponding to the kth incident mode on Ω_n with amplitude 1 at Γ_n , that is

$$\tilde{u}_{i}(\xi\xi^{(n)} + (\eta + h_{n})\eta^{(n)}) = \left(e^{-i\sqrt{\kappa - \lambda_{k}^{(n)}} \eta} - e^{i\sqrt{\kappa - \lambda_{k}^{(n)}} \eta}\right)\Psi_{n}^{p}(\lambda_{k}^{(n)},\xi)$$

for $\xi \in \mathbb{R}$, $\eta > 0$. Then the corresponding source functional in the sense of the weak problem 6.2.1 is given by

$$F_{n,k}^{\text{inc}}(\psi) := \int_{\Gamma_n} (\partial_\nu \tilde{u}_i) \overline{\psi} \, \mathrm{d}s = 2i \sqrt{\kappa - \lambda_k^{(n)}} F_{n,k}(\psi).$$

Proof. This follows immediately by noting that that $\partial_{\nu} = \partial_{\eta}$ in the corresponding coordinate system.

Let us now define the functional we will optimise: the energy contained in an outgoing mode.

6.4.4 Definition. Let $n \in \{0, \ldots, m-1\}, k \in \{1, \ldots, N_n\}$. We define the energy functional $J_{n,k}: H^1(\mathbb{R}^2) \to \mathbb{R}$ by

$$J_{n,k}(u) := \left| \sqrt{\kappa - \lambda_k^{(n)}} \right| |F_{n,k}(u)|^2.$$

It will become clearer later on, why this functional represents the energy flow. Let us now derive the corresponding adjoint state representation of the derivative.

6.4.5 Lemma. Let $\kappa \in \mathbb{C} \setminus \mathbb{R}$, $F_{s} \in H^{1}(\mathbb{R}^{2})^{*}$, and let $q \in L^{\infty}(\mathbb{R}^{2}, \mathbb{R})$ fulfil Assumption 5.1.2. Let $\mathcal{L} : L^{\infty}(\mathbb{R}^{2}, \mathbb{R}) \to H^{1}(\mathbb{R}^{2})$ be the potential-to-solution operator (Definition 6.3.3). Let $n \in \{0, \ldots, m-1\}, k \in \{1, \ldots, N_{n}\}$. Then we have for any $p, q \in L^{\infty}(\mathbb{R}^{2})$

$$d(J_{n,k} \circ \mathcal{L})(q)p = -\int_{\mathbb{R}^2} p \operatorname{Re}(u_q w_q) \, \mathrm{d}x,$$

where $u_q = \mathcal{L}(q)$ and $w_q \in H^1(\mathbb{R}^2)$ is the solution of

$$b_q(w_q,\psi) = -i \frac{\left|\sqrt{\kappa - \lambda_k^{(n)}}\right|}{\sqrt{\kappa - \lambda_k^{(n)}}} F_{n,k}(u_q) F_{n,k}^{\text{inc}}(\psi) \quad \text{for all } \psi \in H^1(\mathbb{R}^2).$$

Proof. Up to a real constant of $|\kappa - \lambda_k^{(n)}|^{1/2}$, the energy functional is exactly of the same type as considered in Definition and Lemma 6.3.5: accordingly, we obtain by Lemma 6.3.7 that the adjoint state is given by

$$\begin{split} b_q(w_q,\psi) &= 2 \left| \sqrt{\kappa - \lambda_k^{(n)}} \right| F_{n,k}(u_q) \overline{F_{n,k}(\overline{\psi})} \\ &= 2 \left| \sqrt{\kappa - \lambda_k^{(n)}} \right| F_{n,k}(u_q) \int_{\Gamma_n} \Psi_n^p(\lambda_k^{(n)}, \cdot) \psi \, \mathrm{d}s \\ &= 2 \frac{\left| \sqrt{\kappa - \lambda_k^{(n)}} \right|}{2i\sqrt{\kappa - \lambda_k^{(n)}}} F_{n,k}(u_q) 2i\sqrt{\kappa - \lambda_k^{(n)}} \int_{\Gamma_n} \Psi_n^p(\lambda_k^{(n)}, \cdot) \overline{\psi} \, \mathrm{d}s \\ &= -i \frac{\left| \sqrt{\kappa - \lambda_k^{(n)}} \right|}{\sqrt{\kappa - \lambda_k^{(n)}}} F_{n,k}(u_q) F_{n,k}^{\mathrm{inc}}(\psi), \end{split}$$

where we used that $\Psi_n^p(\lambda_k^{(n)}, \cdot)$ is real-valued and the definition of $F_{n,k}^{\text{inc}}$ (see Lemma 6.4.3).

The last lemma tells us that the adjoint state of the energy functional is nothing but a rescaled incident guided mode. This will become important later on in the optimisation section.

6.4.6 Back to the absorption free case. To justify the notion of energy functional, let us now go back to the absorption free case, that is, let now

 $\kappa_0 \in \mathbb{R}$

be properly real. For $n \in \{0, \ldots, m-1\}$, let $u_i^{(n)} \in H^2_{loc}(\Omega_n)$ be a superposition of guided modes, that is, there exists coefficients $\alpha_k^{(n)} \in \mathbb{C}$ for $n \in \{0, \ldots, m-1\}$, $k \in \{1, \ldots, N_n\}$, such that on Ω_n we have

$$u_{i}^{(n)}(\xi\xi^{(n)} + (\eta + h_{n})\eta^{(n)}) = \sum_{k=1}^{N_{n}} \alpha_{k}^{(n)} e^{-i\sqrt{\kappa_{0} - \lambda_{l}^{(n)}} \eta} \psi_{n}^{p}(\lambda_{p}^{(n)}, \xi).$$

For this particular incident field let us now consider the solution of

$$\begin{cases} \Delta u + (\kappa_0 - q)u = 0 & \text{ in } \mathbb{R}^2, \\ u - u_i^{(n)} & \text{ OPRC in } \Omega_n, \ n \in \{0, \dots, m - 1\}, \end{cases}$$
(6.4.1)

which has to be understood in the sense of Section 5.3. Our goal is to formally derive an energy conservation property for this solution.

6.4.7 The energy flow. Take a solution u of (6.4.1). By multiplying the Helmholtz equation by \overline{u} , integrating over some bounded domain $D \subset \mathbb{R}^2$ and applying integration by parts we obtain

6.4. ENERGY CONSERVATION

that is, we have obtained the energy conservation relation

$$0 = \int_{\partial D} \operatorname{Im}[(\partial_{\nu} u)\overline{u}] \,\mathrm{d}s.$$
(6.4.2)

We interpret this equation as the conservation of the energy-flow, which is defined as the real valued vector field

$$\operatorname{Im}[(\nabla u)\overline{u}],$$

so that the boundary integral over ∂D gives the energy flow through the closed surface ∂D .

6.4.8 The energy flow of a halfspace. We now aim to calculate the energy-flow of our solution u across the boundary of some halfspace. Hence, let us fix again $n \in \{0, \ldots, m-1\}$. To simplify the calculation, we will transform into the coordinate system of Ω_n . Let us denote

$$\hat{u}(x) := u(x_1\xi^{(n)} + (x_2 + h_n)\eta^{(n)}) \quad \text{for all } x \in \mathbb{R}^2_+,$$
$$\hat{u}_i(x) := u_i^{(n)}(x_1\xi^{(n)} + (x_2 + h_n)\eta^{(n)}) \quad \text{for all } x \in \mathbb{R}^2_+.$$

Let us furthermore denote the traces

$$g_{\mathbf{i}}(x_1) := \hat{u}_{\mathbf{i}}^{(n)}(x_1, 0), \qquad g_{\mathbf{s}}(x_1) := \hat{u}(x_1, 0) - \hat{u}_{\mathbf{i}}(x_1, 0), \qquad \text{for } x_1 \in \mathbb{R}.$$

By definition, we have for $x \in \mathbb{R}^2_+$

$$\hat{u}(x) = \mathcal{S}_n g_s(x) + \hat{u}_i(x) = \mathcal{F}_A^{-1} \left[e^{i\sqrt{\kappa_0 - \lambda(\cdot)} x_2} \mathcal{F}_A g_s \right](x_1) + u_i(x),$$

and note furthermore that similarly

$$\hat{u}_{\mathbf{i}}(x) = \mathcal{F}_{A}^{-1} \left[e^{-i\sqrt{\kappa_{0} - \lambda(\cdot)} x_{2}} \mathcal{F}_{A} g_{\mathbf{i}} \right] (x_{1}).$$

We can now calculate with the help of Parseval's equality (see Theorem 2.2.12)

$$\begin{split} \operatorname{Im} & \int_{\Gamma_n} (\partial_{\nu} u) \overline{u} \, \mathrm{d}s = \operatorname{Im} \int_{\mathbb{R}} (\partial_{x_2} \hat{u})(x_1) \overline{u}(x_1) \, \mathrm{d}x_1 \\ &= \operatorname{Im} \int_{\mathbb{R}} \mathcal{F}_A^{-1} \left[i \sqrt{\kappa_0 - \hat{\lambda}(\cdot)} \mathcal{F}_A g_{\mathrm{s}} - i \sqrt{\kappa_0 - \hat{\lambda}(\cdot)} \mathcal{F}_A g_{\mathrm{i}} \right] (\overline{g}_{\mathrm{s}} + \overline{g}_{\mathrm{i}}) \, \mathrm{d}x \\ &= \operatorname{Im} \int_{\Lambda} i \sqrt{\kappa_0 - \hat{\lambda}(\cdot)} \left(\mathcal{F}_A g_{\mathrm{s}} - \mathcal{F}_A g_{\mathrm{i}} \right) \left(\overline{\mathcal{F}}_A g_{\mathrm{s}} + \overline{\mathcal{F}}_A g_{\mathrm{i}} \right) \, \mathrm{d}\mu \\ &= \int_{\Lambda} \operatorname{Re} \left(\sqrt{\kappa_0 - \hat{\lambda}(\cdot)} \right) |F_A g_{\mathrm{s}}|^2 \, \mathrm{d}\mu - \int_{\Lambda} \operatorname{Re} \left(\sqrt{\kappa_0 - \hat{\lambda}(\cdot)} \right) |F_A g_{\mathrm{i}}|^2 \, \mathrm{d}\mu. \end{split}$$

Let us now assume that $\kappa_0 > q_-$, so that $\sqrt{\kappa_0 - \lambda_k^{(n)}} \in \mathbb{R}$ for all $k \in \{1, \ldots, N_n\}$. One now easily obtains that

$$\int_{\Lambda} \operatorname{Re}\left(\sqrt{\kappa_0 - \hat{\lambda}(\cdot)}\right) |F_A g_i|^2 \,\mathrm{d}\mu = \sum_{k=1}^{N_s} \sqrt{\kappa_0 - \lambda_k^{(n)}} \,|\alpha_k^{(n)}|^2.$$

Furthermore, by definition of \mathcal{F}_A we have

$$\beta_k^{(n)} := (\mathcal{F}_A g_{\mathbf{s}})^p (\lambda_k^{(p)}) = \int_{\mathbb{R}} \Psi_n^p (\lambda_k^{(n)}, x_1) g_{\mathbf{s}}(x_1) \, \mathrm{d}x_1.$$

Note that we defined $\beta_k^{(n)}$ here. This in turn implies that

$$\int_{\Lambda} \operatorname{Re}\left(\sqrt{\kappa_{0} - \lambda(\cdot)}\right) |F_{A}g_{s}|^{2} d\mu$$

$$= \sum_{k=1}^{N_{n}} \sqrt{\kappa_{0} - \lambda_{k}^{(n)}} |\beta_{k}^{(n)}|^{2} + \sum_{\sigma \in \{\pm\}} \int_{q_{\sigma}}^{\kappa_{0}} \sqrt{\kappa - \lambda} |(F_{A}g_{s})^{\sigma}(\lambda)|^{2} \frac{d\lambda}{2\sqrt{\lambda - q_{\sigma}}}$$

$$= \sum_{k=1}^{N_{n}} \sqrt{\kappa_{0} - \lambda_{k}^{(n)}} |\beta_{k}^{(n)}|^{2} + E_{\operatorname{free}}(u_{s}),$$

where we defined the scattered energy $E_{\text{free}}(u_s)$ in the last equation. Now, let us note the meaning of the energy functional $J_{n,k}(u)$: we have

$$J_{n,k}(u-u_{i}) = \sqrt{\kappa_{0} - \lambda_{k}^{(n)}} \left| \int_{\Gamma_{n}} \Psi_{n}^{p}(\hat{\lambda}_{k}^{(n)}, \cdot) \overline{(u-u_{i})} \, \mathrm{d}s \right|^{2}$$
$$= \sqrt{\kappa_{0} - \lambda_{k}^{(n)}} \, |\beta_{k}^{(n)}|^{2},$$

so that $J_{n,k}(u - u_i^{(n)})$ yields the energy contained in the *k*th outgoing mode of the waveguide in Ω_n . Similarly, we can also deduce that

$$J_{n,k}(u_{i}^{(n)}) = \sqrt{\kappa_{0} - \lambda_{k}^{(n)}} \, |\alpha_{k}^{(n)}|^{2}$$

To sum up, we have obtained the following expression for the energy flow

$$\int_{\Gamma_n} \operatorname{Im}[(\partial_{\nu} u)\overline{u}] \, \mathrm{d}s = -\left(\sum_{k=1}^{N_n} J_{n,k}(u_1^{(n)})\right) + \left(\sum_{k=1}^{N_n} J_{n,k}(u - u_1^{(n)}) + E_{\operatorname{free}}(u - u_1^{(n)})\right).$$
(6.4.3)

6.4.9 Energy conservation for the halfspace flows. Let us, again in a rather handwaving fashion, derive an energy conservation equality. First note that carefully checking the calculation of the last section, we see that the energy flow of a halfspace does not change if we take a surface parallel to Γ_n , that is

$$\int_{\Gamma_n} \operatorname{Im}[(\partial_{\nu} u)\overline{u}] \,\mathrm{d}s = \int_{\Gamma_n^R} \operatorname{Im}[(\partial_{\nu} u)\overline{u}] \,\mathrm{d}s,$$

where $\Gamma_n^R := \{\xi \xi^{(n)} + (h_n + R) \eta^{(n)} : \xi \in \mathbb{R}\}$. This now motivates the following discussion. Assume that $0 \in \Omega$. Denote the stretched version of Ω by

$$\Omega_R = (1+R)\Omega = \{(1+R)x : x \in \Omega\}.$$

6.5. OPTIMISATION

We get by the energy conservation relation (6.4.2) that

$$0 = \int_{\partial\Omega_R} \operatorname{Im}[(\partial_{\nu} u)\overline{u}] \,\mathrm{d}s = \sum_{n=0}^{m-1} \int_{\Gamma_n^{(1+R)h_n} \cap \partial\Omega_R} \operatorname{Im}[(\partial_{\nu} u)\overline{u}] \,\mathrm{d}s$$
$$\longrightarrow \sum_{n=0}^{m-1} \int_{\Gamma_n} \operatorname{Im}[(\partial_{\nu} u)\overline{u}] \,\mathrm{d}s \quad \text{as } R \to \infty.$$

Applying now (6.4.3) to rewrite the expression after the limit, we obtain that

$$\sum_{n=1}^{m-1} \sum_{k=1}^{N_n} J_{n,k}(u_i^{(n)}) = \sum_{n=1}^{m-1} \left(E_{\text{free}}(u - u_i^{(n)}) + \sum_{k=1}^{N_n} J_{n,k}(u - u_i^{(n)}) \right).$$
(6.4.4)

6.5 Optimisation

6.5.1 The generic setting. Let $\kappa_0 \in \mathbb{R}$, and let $q_e \in L^{\infty}(\mathbb{R}^2 \setminus \overline{\Omega})$ fulfil Definition and Assumption 5.1.2. In the following, we will consider a few optimisation examples of the following type: let $U_i = (u_i^{(0)}, \ldots, u_i^{(m-1)})$ be some incident field consisting of guided modes. Let furthermore

$$J(u) := J_{n_t,k_t}(u)$$

be the energy contained in the k_t th outgoing mode in Ω_{n_t} . Let $q_{\min} < q_{\max}$ be two fixed constants. We define the design space of admissible potentials by

$$\mathcal{D} := \{ q \in L^{\infty}(\Omega, \mathbb{R}) : q_{\min} \le q(x) \le q_{\max} \text{ for almost all } x \in \Omega \}.$$

Now consider the following optimisation problem

Maximise
$$J(u_q)$$
 for $q \in \mathcal{D}$, where u_q solves

$$\begin{cases}
\Delta u_q + (\kappa_0 - q)u_q = 0 & \text{in } \mathbb{R}^2, \\
u_q - u_i^{(n)} & \text{OPRC in } \Omega_n, n \in \{0, \dots, m-1\}, \\
\text{where } q \text{ is extended by } q \text{ outside of } \Omega
\end{cases}$$

where q is extended by $q_{\rm e}$ outside of Ω .

We interpret this problem as follows: given an exterior potential q_e as well as an incident field, we try to construct a structure contained in Ω , which allows us to transmit the incident energy into the k_t th outgoing mode of the waveguide in Ω_n .

Recall that we defined $\mathcal{J}: L^{\infty}(\Omega, \mathbb{R}) \to \mathbb{R}$ by

$$\mathcal{J}(q) := J(u_q).$$

6.5.2 An iterative optimisation algorithm. To solve this problem numerically, let us give a rough iterative algorithm yielding such structures. Start by choosing two tuning parameters, the maximal step size $h_{\text{max}} > 0$ and a regularisation parameter $\alpha \in (0, 1]$. Begin with some starting guess $q_1 \in \mathcal{D}$, and construct the next iteration by the following algorithm in the *l*th iteration, $l \in \mathbb{N}$.

(a) Compute the forward solution $u_l = u_{q_l}$.

(b) Compute the adjoint state w_l as the solution to

$$\begin{cases} \Delta w_l + (k_0 - q_l)w_l = 0 & \text{in } \mathbb{R}^2, \\ w_l - w_i^{(n)} & \text{OPRC in } \Omega_n, \ n \in \{0, \dots, m-1\}, \end{cases}$$

where the incident field is given on Ω_{n_t} by

$$w_{i}^{(n_{t})}(\xi\xi^{(n)} + (\eta + h_{n})\eta^{(n)}) = -iF_{n_{t},k_{t}}(u_{l})e^{-i\sqrt{k_{0}-\lambda_{k_{t}}^{(n_{t})}\eta}}\Psi_{n_{t}}^{p}(\lambda_{k_{t}}^{(n_{t})}\xi),$$

for $\xi \in \mathbb{R}$, $\eta > 0$, and by $w_i^{(n)} = 0$ on the other halfspaces Ω_n , $n \neq n_t$.

(c) Define the gradient $g_l \in L^{\infty}(\Omega, \mathbb{R})$ by

$$g_l(x) := -\operatorname{Re}(u_l(x)w_l(x)) \quad \text{for } x \in \Omega.$$

Note that the gradient is the L^2 -gradient of $d\mathcal{J}(q_l) : L^{\infty}(\mathbb{R}^2, \mathbb{R}) \to \mathbb{R}$, that is, the unique element $q_l \in L^2(\Omega)$ such that

$$\int_{\Omega} g_l p \, \mathrm{d}x = d\mathcal{J}(q_l) p = \int_{\Omega} -\operatorname{Re}(u_l w_l) p \, \mathrm{d}x \quad \text{for all } p \in L^2(\Omega),$$

where we used Lemma 6.4.5 to obtain the representation of the derivative. To be able to use this representation, we invoke again the limit-absorption principle of Conjecture 6.2.2.

(d) Based on the already computed gradients g_0, \ldots, g_l , choose some preliminary ascent direction $\tilde{d}_l \in L^{\infty}(\Omega, \mathbb{R})$. We will use a non-linear conjugate gradient method with the Polak-Ribière update formula [29].

(e) Project the descent direction, so that it points into \mathcal{D} : that is, define

$$d_{l}(x) := \begin{cases} 0 & \text{if } q_{l}(x) = q_{\max} \text{ and } d_{l}(x) > 0, \\ 0 & \text{if } q_{l}(x) = q_{\min} \text{ and } d_{l}(x) < 0, \\ \tilde{d}_{l}(x) & \text{else.} \end{cases}$$

(f) To calculate a step size h_l , we use utilise the second derivative, so we calculate the first material derivative u'_l in direction d_l , which is the solution of

$$\begin{cases} \Delta u'_l + (\kappa_0 - q_l)u'_l = -d_l u_l & \text{in } \mathbb{R}^2, \\ u'_l & \text{OPRC in } \Omega_n, \ n \in \{0, \dots, m-1\}, \end{cases}$$

as well as the second derivative u_l'' in direction d_l , which solves

$$\begin{cases} \Delta u_l'' + (\kappa_0 - q_l)u_l'' = -2d_l u_l' & \text{ in } \mathbb{R}^2, \\ u_l'' & \text{ OPRC in } \Omega_n, \ n \in \{0, \dots, m-1\}, \end{cases}$$

Now the first and second derivative of \mathcal{J} in direction d_l are given by

$$\delta_{1} := d\mathcal{J}(q_{l})d_{l} = \sqrt{\kappa_{0} - \lambda_{k_{t}}^{(n_{t})}} 2\operatorname{Re}\left(F_{n_{t},k_{t}}(u_{l})\overline{F_{n_{t},k_{t}}(u_{l}')}\right)$$

$$\delta_{2} := d^{2}\mathcal{J}(q_{l})[d_{l},d_{l}] = \sqrt{\kappa_{0} - \lambda_{k_{t}}^{(n_{t})}} \left(2\operatorname{Re}\left(F_{n_{t},k_{t}}(u_{l}')\overline{F_{n_{t},k_{t}}(u_{l})}\right) + 2\operatorname{Re}\left(F_{n_{t},k_{t}}(u_{l}')\overline{F_{n_{t},k_{t}}(u_{l}')}\right)\right),$$

where we employed Lemmata 6.3.4, 6.3.8 and Definition and Lemma 6.3.5. We also used the limit-absorption principle of Conjecture 6.2.2 again. Now we choose the step size as follows

$$h_l = \begin{cases} \min\left\{h_{\max} \|d_l\|_{L^{\infty}(\Omega)}^{-1}, -\alpha \frac{\delta_1}{\delta_2}\right\} & \text{if } \delta_2 < 0, \\ h_{\max} \|d_l\|_{L^{\infty}(\Omega)}^{-1} & \text{else.} \end{cases}$$

Let us quickly explain the purpose of this step size: if \mathcal{J} is negatively curved along the direction d_l , that is, if $\delta_2 < 0$, we have a maximum and choose the step size to step onto this maximum (of the second order Taylor expansion). The parameter α serves as a security value to avoid overstepping.

On the other hand, if $\delta_2 > 0$, we choose the pre-defined maximal step distance h_{max} .

(g) Calculate the preliminary update by

$$\tilde{q}_{l+1} = q_l + h_l d_l,$$

and project back into the set of admissible designs \mathcal{D} , that is define

$$q_{l+1}(x) = \begin{cases} q_{\max} & \text{if } \tilde{q}_l(x) > q_{\max}, \\ q_{\min} & \text{if } \tilde{q}_l(x) < q_{\min}, \\ \tilde{q}_{l+1}(x) & \text{else.} \end{cases}$$

With the new iterate q_{l+1} , restart the process at step (a).

6.6 Examples

In this section, we will consider different examples of waveguide scattering problems and study equations of the form

$$\Delta u(x) + p(x)u(x) = 0 \quad \text{for } x \in \mathbb{R}^2, \tag{6.6.1}$$

with real-valued $p : \mathbb{R}^2 \to \mathbb{R}$. From p, one easily finds suitable q and $\kappa_0 \in \mathbb{R}$ to get back to our conventional setting by setting $\kappa_0 = 0$ and q = -p. The situations we consider are motivated from design problems surrounding optical waveguides, which have been studied in the physical literature to a large extend. For more information, we refer to [31, 33, 32, 69].

6.6.1 A bend of a waveguide. Let us start with the first example of a bent waveguide, which turns by 90 degrees, with a first mode incident from one side to the bend. An example solution is shown in Figure 6.1 together with the basic geometry of the computational domain: note that the waveguide bend forms a quarter circle, with an inner radius r_i and an outer radius r_o . Since the waveguide has a width of h, we have that $r_o = r_i + h$.

For many applications it is of relevance to construct bends with minimal loss. The first, somewhat intuitive idea is to increase the radius of curvature. How does the transmission of energy change for bends of different radius? Figure 6.2 shows the dependence of the transmission on the radius of curvature. The transmission is defined by

$$\tau := \frac{J_{1,1}(u - u_{\rm i})}{J_{0,1}(u_{\rm i})},$$



Figure 6.1: Sketch of the geometrical situation for the bent waveguide (left) and the real part of the solution for a bend radius of $r_0 = 6$ (right).



Figure 6.2: Transmission of energy for different radii of curvature.



Figure 6.3: Optimisation of a waveguide bend. On the left, the transmission as a function of the number of iterations is shown. The middle shows the structure after the optimisation. Dark grey corresponds to to p(x) = 3, while white corresponds to p(x) = 1. On the right the real part of the solution after optimisation is shown. Note that the transmission was increased from 45.1% to 65.1% without increasing the effective curvature of the waveguide. The initial curvature corresponds to $r_o = 6$, which is the same as in Figure 6.1.

6.6. EXAMPLES

that is, as the energy contained in the first mode in Ω_1 of the scattered field, divided by the energy contained in the first mode of the incident field on Ω_0 , which is the only part of the incident field. From (6.4.4), we obtain that

$$J_{0,1}(u_{i}^{(n)}) = \sum_{n=1}^{m-1} \left(E_{\text{free}}(u - u_{i}^{(n)}) + \sum_{k=1}^{N_{n}} J_{n,k}(u - u_{i}^{(n)}) \right) \ge J_{1,1}(u - u_{i}^{(n)}),$$

which in turn implies

 $\tau \leq 1,$

which is why we consider $\tau = 1$ a *perfectly lossless* transmission.

We have now applied the optimisation algorithm from subsection 6.5.2 to this problem, to find a structure which allows a waveguide bend with low losses, without effectively increasing the "curvature" of the bend (in the sense that input and output waveguide stay at the same place). A reasonable increase of transmission from 45% to 65% was obtained (see Figure 6.3).

6.6.2 A mode splitter. As a second example, we want to construct a junction, which splits a single mode into two. For this aim, let again u_i denote the first incident mode in the top waveguide in Ω_0 , as sketched in Figure 6.4. We demand again that $J_{0,1}(u_i) = 1$.



Figure 6.4: Sketch of the waveguide splitter.

Note that we used a triangular interior domain. Our goal is now to construct a junction, which evenly splits the mode in the top waveguide into the two bottom waveguides. We model this with help of the following optimisation problem:

Find some $p \in \{\tilde{p} \in L^{\infty}(\mathbb{R}^2, \mathbb{R}) : 1 \leq \tilde{p}(x) \leq 2 \text{ for all } x \in \Omega\}$ such that the functional

$$J(u) = J_{1,1}(u - u_i)J_{2,1}(u - u_i)$$

gets maximal. Here u is the outgoing solutions to (6.6.1) with incident fields u_i .



Figure 6.5: Evolution of the FOM during the optimisation. Note that we reach almost 50% transmission for each of the two waveguides.



Figure 6.6: Resulting structure and real part of the solution after the optimisation.

Note that we chose the product of the two energies as the figure of merit. By the energy conservation relation we have

$$1 = J_{0,1}(u_{\rm i}) \ge J_{1,1}(u - u_{\rm i}) + J_{2,1}(u - u_{\rm i}).$$

Assuming that no energy is lost due to some scattered part, that is, if equality holds in the last inequality, i.e. if

$$J_{1,1}(u - u_{i}) + J_{2,1}(\tilde{u} - \tilde{u}_{i}) = 1,$$

one easily sees that J gets maximal if and only if

$$J_{1,1}(u - u_{\rm i}) = J_{2,1}(\tilde{u} - \tilde{u}_{\rm i}),$$

that is, if the energy is evenly distributed between the two lower waveguides. In this case, half the energy would be transmitted into each of the waveguides, and we would obtain

$$J(u) = \frac{1}{2}\frac{1}{2} = \frac{1}{4}.$$

6.6. EXAMPLES

So that our choice J ensures that a optimal solution will both ensure a low scattering loss of energy, while evenly balancing out the energy between the two output waveguides. We note that this optimisation problem does not fall in the scheme we studied in the previous section. A straightforward, but somewhat technical discussion, which we omit for brevity, yields that the adjoint state of 6.5.2(b) can be computed as the solution of

$$\begin{cases} \Delta w_l + (k_0 - q_l)w_l = 0 & \text{in } \mathbb{R}^2, \\ w_l - w_i^{(n)} & \text{OPRC in } \Omega_n, \ n \in \{0, \dots, m-1\}, \end{cases}$$

where the incident field is given (in a slightly sloppy notation) by

$$w_{i}(x) = \begin{cases} 0 & \text{on } \Omega_{0}, \\ -iJ_{2,1}(u_{l})F_{1,1}(u_{l})e^{-i\sqrt{k_{0}-\lambda_{1}^{(1)}}\eta}\Psi_{1}^{p}(\lambda_{1}^{(1)}\xi) & \text{on } \Omega_{1}, \\ -iJ_{1,1}(u_{l})F_{2,1}(u_{l})e^{-i\sqrt{k_{0}-\lambda_{1}^{(2)}}\eta}\Psi_{2}^{p}(\lambda_{1}^{(2)}\xi) & \text{on } \Omega_{2}. \end{cases}$$

for $\xi \in \mathbb{R}$, $\eta > 0$. Similarly, δ_1 and δ_2 in 6.5.2(f) have to be replaced by slightly bulky expressions

$$\begin{split} \delta_{1} = &\sqrt{\kappa_{0} - \lambda_{1}^{(2)}} \sqrt{\kappa_{0} - \lambda_{1}^{(1)}} \\ &\cdot \left(2 \operatorname{Re} \left(F_{1,1}(u_{l}) \overline{F_{1,1}(u_{l}')} \right) |F_{1,1}(u_{l})|^{2} \\ &+ 2 \operatorname{Re} \left(F_{2,1}(u_{l}) F_{2,1}(u_{l}') \right) |F_{2,1}(u_{l})|^{2} \right), \\ \delta_{2} = &\sqrt{\kappa_{0} - \lambda_{1}^{(2)}} \sqrt{\kappa_{0} - \lambda_{1}^{(1)}} \\ &\cdot \left(2 \operatorname{Re} \left(F_{1,1}(u_{l}) \overline{F_{1,1}(u_{l}')} \right) |F_{2,1}(u_{l})|^{2} \\ &+ 2 \operatorname{Re} \left(F_{2,1}(u_{l}) \overline{F_{2,1}(u_{l}')} \right) |F_{1,1}(u_{l})|^{2} \\ &+ 2 \left| F_{1,1}(u_{l}') \right|^{2} |F_{2,1}(u_{l})|^{2} + 2 \left| F_{2,1}(u_{l}') \right|^{2} |F_{1,1}(u_{l})|^{2} \\ &+ 8 \operatorname{Re} \left(F_{1,1}(u_{l}) \overline{F_{1,1}(u_{l}')} \right) \operatorname{Re} \left(F_{2,1}(u_{l}) \overline{F_{2,1}(u_{l}')} \right) \right). \end{split}$$

With those modifications, we can now optimise the waveguide splitter to obtain a better splitting. The results can be seen in Figures 6.5 and 6.6.

6.6.3 A mode flipper. For the next example, we aim to construct a waveguide which optimises two scattering problems at the same time. Consider a straight waveguide, as shown in Figure 6.7. This waveguide has two guided modes, and we can sent in both as incident fields on Ω_0 . Let u_i be the first incident mode, and let \tilde{u}_i be the second, which are normalised so that

$$J_{0,1}(u_i) = 1$$
 and $J_{0,2}(\tilde{u}_i) = 1$,

that is, they both contain the same energy. In the sketched configuration, both modes will simply propagate through the domain without changing their shape.



Figure 6.7: Geometry of the mode flipping problem: the modes are incident in the straight waveguide from the left, and we try to manipulate the output on the right.



Figure 6.8: Evolution of the figure of merit for the waveguide flipping during the optimisation. Note that we get a total figure of merit of up to 1.988, which is rather close to the optimal value of 2.



Figure 6.9: Optimised material configuration (top) and corresponding real part of the total field u (middle) and \tilde{u} (bottom). Note that all figures are almost point symmetrical to the origin.

6.6. EXAMPLES

We can now ask the following question: can we perturb the waveguide in such way that the modes are *flipped* at the end of the finite element domain Ω ? That is, so that u_i produces a second outgoing mode on the right, while \tilde{u}_i produces a first outgoing mode? We can reformulate this problem in the following (simultaneous) optimisation problem:

Find some $p \in \{\tilde{p} \in L^{\infty}(\Omega, \mathbb{R}) : 3 \leq \tilde{p}(x) \leq 9 \text{ for all } x \in \Omega\}$ such that the functional

 $J(u, \tilde{u}) = J_{2,2}(u - u_{i}) + J_{2,1}(\tilde{u} - \tilde{u}_{i})$

gets maximal. Here u and \tilde{u} are the outgoing solutions to (6.6.1) with incident fields u_i and \tilde{u}_i .

In other words: we aim to maximise the sum of two energies: the energy produced by the first incident mode in the second outgoing mode, and the energy produced by the second incident mode in the first outgoing mode. Note that we obtain that by (6.4.4)

$$J(u,\tilde{u}) = J_{2,2}(u-u_{\rm i}) + J_{2,1}(\tilde{u}-\tilde{u}_{\rm i}) \le J_{0,1}(u_{\rm i}) + J_{0,2}(\tilde{u}_{\rm i}) = 2,$$

so that we can consider $J(u, \tilde{u}) = 2$ as a perfect flipping of the modes. Note that in Figures 6.8 and 6.9, we almost achieve this perfect flipping (we obtain $J(u, \tilde{u}) = 1.988$). This is a somewhat simpler version of a mode converter (see [49, 44]).

6.6.4 A mode multiplexer. The last example we want to study is a mode multiplexer: assume we have a junction of three waveguides as shown in Figure 6.10. Note the width of the waveguides: the two output waveguides on the right and top have a smaller width, while the input waveguide in Ω_0 has a larger width. This causes it to have two guided modes, while the output waveguides each have one guided mode. Note that we put this arrangement of waveguides on top of a substrate, whose material parameter $p_{subs} = 6$ is between the free space $p_{free} = 3$ and the waveguide $p_{wg} = 9$. We now aim to construct



Figure 6.10: Sketch of the geometry of the waveguide multiplexer.

a junction which transmits the first mode incident in the left waveguide into the right hand waveguide, while the second incident mode is transmitted into the top waveguide. Let again u_i be the first incident mode on Ω_0 , and let \tilde{u}_i be the second mode on Ω_0 ,



Figure 6.11: Evolution of the figure of merit for the waveguide multiplexer. Note that around iteration 75, the algorithm oversteps and actually reduces the figure of merit. Note that at the final shape, around 2% of the energy in the first mode is not transmitted to right. The second mode has a higher loss of around 16%.

which are normalised so that

$$J_{0,1}(u_i) = 1$$
 and $J_{0,2}(\tilde{u}_i) = 1$.

To add an additional restriction, we consider the substrate as fixed, that is, we cannot change the coefficient p on the domain Ω_{subs} , which is the light grey area in Figure 6.10. Let us now reformulate the multiplexer problem as the following optimisation problem:

Find some $p \in \{\tilde{p} \in L^{\infty}(\Omega, \mathbb{R}) : 3 \leq \tilde{p}(x) \leq 9 \text{ for all } x \in \Omega, q(x) = 6 \text{ for } x \in \Omega_{\text{subs}}\}$ such that the functional $J(u, \tilde{u}) = J_{2,1}(u - u_{\text{i}}) + J_{3,1}(\tilde{u} - \tilde{u}_{\text{i}})$ gets maximal. Here u and \tilde{u} are the outgoing solutions to (6.6.1) with incident fields u_{i} and \tilde{u}_{i} .

This problem has been considered recently in the context of nanophotonics, see for example [69, 32]. Conventional mode multiplexers require relatively much space, and do not allow the integration in a small, compact device, so it seems a rather interesting prospect to integrate them on a smaller scale.



Figure 6.12: Result of the optimisation process for the waveguide multiplexer. At the top one can see the resulting material distribution in the computational domain, while the real parts of the two solutions u and \tilde{u} are shown below.

Bibliography

- S. Agmon. Spectral properties of Schrödinger operators and scattering theory. Ann. Scuola Norm. Sup. Pisa Cl. Sci. (4), 2(2):151–218, 1975.
- [2] T. Arens and T. Hohage. On radiation conditions for rough surface scattering problems. IMA J. Appl. Math., 70(6):839–847, 2005.
- [3] M. P. Bendsøe and N. Kikuchi. Generating optimal topologies in structural design using a homogenization method. *Comput. Methods Appl. Mech. Engrg.*, 71(2):197– 224, 1988.
- [4] M. P. Bendsøe and O. Sigmund. *Topology optimization*. Springer-Verlag, Berlin, 2003. Theory, methods and applications.
- [5] J.-P. Berenger. A perfectly matched layer for the absorption of electromagnetic waves. J. Comput. Phys., 114(2):185-200, 1994.
- [6] Y. M. Berezansky, Z. G. Sheftel, and G. F. Us. Functional analysis. Vol. I, volume 85 of Operator Theory: Advances and Applications. Birkhäuser Verlag, Basel, 1996. Translated from the 1990 Russian original by Peter V. Malyshev.
- [7] Y. M. Berezansky, Z. G. Sheftel, and G. F. Us. Functional analysis. Vol. II, volume 86 of Operator Theory: Advances and Applications. Birkhäuser Verlag, Basel, 1996. Translated from the 1990 Russian original by Peter V. Malyshev.
- [8] C. Besse, J. Coatléven, S. Fliss, I. Lacroix-Violet, and K. Ramdani. Transparent boundary conditions for locally perturbed infinite hexagonal periodic media. *Commun. Math. Sci.*, 11(4):907–938, 2013.
- [9] A.-S. Bonnet-Ben Dhia, G. Dakhia, C. Hazard, and L. Chorfi. Diffraction by a defect in an open waveguide: a mathematical analysis based on a modal radiation condition. SIAM J. Appl. Math., 70(3):677–693, 2009.
- [10] A.-S. Bonnet-Ben Dhia, B. Goursaud, and C. Hazard. Mathematical analysis of the junction of two acoustic open waveguides. *SIAM J. Appl. Math.*, 71(6):2048–2071, 2011.
- [11] A. Boutet de Monvel-Berthier and D. Manda. Spectral and scattering theory for wave propagation in perturbed stratified media. J. Math. Anal. Appl., 191(1):137– 167, 1995.
- [12] J. Chabrowski. The Dirichlet problem with L²-boundary data for elliptic linear equations, volume 1482 of Lecture Notes in Mathematics. Springer-Verlag, Berlin, 1991.

- [13] S. N. Chandler-Wilde. Boundary value problems for the Helmholtz equation in a half-plane. In *Mathematical and numerical aspects of wave propagation (Mandelieu-La Napoule, 1995)*, pages 188–197. SIAM, Philadelphia, PA, 1995.
- [14] S. N. Chandler-Wilde and P. Monk. Existence, uniqueness, and variational methods for scattering by unbounded rough surfaces. SIAM J. Math. Anal., 37(2):598–618, 2005.
- [15] S. N. Chandler-Wilde, C. R. Ross, and B. Zhang. Scattering by infinite onedimensional rough surfaces. R. Soc. Lond. Proc. Ser. A Math. Phys. Eng. Sci., 455(1990):3767–3787, 1999.
- [16] W. C. Chew and W. H. Weedon. A 3d perfectly matched medium from modified maxwell's equations with stretched coordinates. *Microwave and Optical Technology Letters*, 7(13):599–604, 1994.
- [17] T. Christiansen. Scattering theory for perturbed stratified media. J. Anal. Math., 76:1–43, 1998.
- [18] G. Ciraolo. A radiation condition for the 2-D Helmholtz equation in stratified media. Comm. Partial Differential Equations, 34(10-12):1592–1606, 2009.
- [19] G. Ciraolo and R. Magnanini. Analytical results for 2-D non-rectilinear waveguides based on a Green's function. *Math. Methods Appl. Sci.*, 31(13):1587–1606, 2008.
- [20] G. Ciraolo and R. Magnanini. A radiation condition for uniqueness in a wave propagation problem for 2-D open waveguides. *Math. Methods Appl. Sci.*, 32(10):1183– 1206, 2009.
- [21] R. Ciraolo. Non-rectilinear waveguides: Analytical and numerical results based on the Green's function. PhD thesis, Università degli Studi di Firenze, 2005.
- [22] E. A. Coddington and N. Levinson. Theory of ordinary differential equations. McGraw-Hill Book Company, Inc., New York-Toronto-London, 1955.
- [23] S. De Bièvre. Spectral analysis for optical fibres and stratified fluids. In *Rigorous results in quantum dynamics (Liblice, 1990)*, pages 143–146. World Sci. Publ., River Edge, NJ, 1991.
- [24] S. De Bièvre and D. W. Pravica. Spectral analysis for optical fibres and stratified fluids. I. The limiting absorption principle. J. Funct. Anal., 98(2):404–436, 1991.
- [25] S. De Bièvre and D. W. Pravica. Spectral analysis for optical fibres and stratified fluids. II. Absence of eigenvalues. *Comm. Partial Differential Equations*, 17(1-2):69– 97, 1992.
- [26] M. C. Delfour and J.-P. Zolésio. Shapes and geometries, volume 22 of Advances in Design and Control. Society for Industrial and Applied Mathematics (SIAM), Philadelphia, PA, second edition, 2011. Metrics, analysis, differential calculus, and optimization.
- [27] N. Dunford and J. T. Schwartz. Linear operators. Part II: Spectral theory. Self adjoint operators in Hilbert space. With the assistance of William G. Bade and Robert G. Bartle. Interscience Publishers John Wiley & Sons New York-London, 1963.

- [28] L. C. Evans. Partial differential equations, volume 19 of Graduate Studies in Mathematics. American Mathematical Society, Providence, RI, second edition, 2010.
- [29] R. Fletcher. *Practical methods of optimization*. Wiley-Interscience [John Wiley & Sons], New York, second edition, 2001.
- [30] S. Fliss and P. Joly. Exact boundary conditions for time-harmonic wave propagation in locally perturbed periodic media. *Appl. Numer. Math.*, 59(9):2155–2178, 2009.
- [31] L. Frellsen, L. Frandsen, and Y. Ding. Topology Optimized Components for Modeand Wavelength Division Multiplexing. PhD thesis, 2016.
- [32] L. F. Frellsen, Y. Ding, O. Sigmund, and L. H. Frandsen. Topology optimized mode multiplexing in silicon-on-insulator photonic wire waveguides. *Opt. Express*, 24(15):16866–16873, Jul 2016.
- [33] L. F. Frellsen, L. H. Frandsen, Y. Ding, Y. Elesin, O. Sigmund, and K. Yvind. Topology optimized design for silicon-on-insulator mode converter. In 2015 IEEE Photonics Conference (IPC), pages 162–163, Oct 2015.
- [34] H. Fujita, N. Saito, and T. Suzuki. Operator theory and numerical methods, volume 30 of Studies in Mathematics and its Applications. North-Holland Publishing Co., Amsterdam, 2001.
- [35] D. V. Ginste, H. Rogier, and D. D. Zutter. Perfectly matched layer based modelling of layered media: Overview and perspective. In 2010 URSI International Symposium on Electromagnetic Theory, pages 803–806, Aug 2010.
- [36] J. W. Goodman. Introduction to Fourier optics. Roberts, Englewood, Colo., 3. ed. edition, 2005.
- [37] P. Grisvard. Elliptic problems in nonsmooth domains, volume 69 of Classics in Applied Mathematics. Society for Industrial and Applied Mathematics (SIAM), Philadelphia, PA, 2011. Reprint of the 1985 original.
- [38] M. Halla and L. Nannen. Hardy space infinite elements for time-harmonic twodimensional elastic waveguide problems. Wave Motion, 59:94–110, 2015.
- [39] C. Hazard. A distribution framework for the generalized fourier transform associated with a sturmliouville operator. *Rapport de Recherche INRIA 6885*, 2009.
- [40] C. Hazard. On the absence of trapped modes in locally perturbed open waveguides. IMA J. Appl. Math., 80(4):1049–1062, 2015.
- [41] F. Hettlich. The Domain Derivative in Inverse Obstacle Problems. Habilitation, Universität Erlangen-Nürnberg, 1999.
- [42] T. Hohage, F. Schmidt, and L. Zschiedrich. Solving time-harmonic scattering problems based on the pole condition. I. Theory. SIAM J. Math. Anal., 35(1):183–210, 2003.
- [43] T. Hohage, F. Schmidt, and L. Zschiedrich. Solving time-harmonic scattering problems based on the pole condition. II. Convergence of the PML method. SIAM J. Math. Anal., 35(3):547–560, 2003.

- [44] B. M. Holmes and D. C. Hutchings. Realization of novel low-loss monolithically integrated passive waveguide mode converters. *IEEE Photonics Technology Letters*, 18(1):43–45, Jan 2006.
- [45] T. Ikebe. Eigenfunction expansions associated with the Schroedinger operators and their applications to scattering theory. Arch. Rational Mech. Anal., 5:1–34 (1960), 1960.
- [46] C. Jerez-Hanckes and J.-C. Nédélec. Asymptotics for Helmholtz and Maxwell solutions in 3-D open waveguides. *Commun. Comput. Phys.*, 11(2):629–646, 2012.
- [47] K. Jörgens. Lineare Integraloperatoren. B. G. Teubner Stuttgart, 1970. Mathematische Leitfäden.
- [48] E. M. Kartchevski, A. I. Nosich, and G. W. Hanson. Mathematical analysis of the generalized natural modes of an inhomogeneous optical fiber. SIAM J. Appl. Math., 65(6):2033–2048, 2005.
- [49] S.-H. Kim, R. Takei, Y. Shoji, and T. Mizumoto. Single-trench waveguide TE-TM mode converter. Opt. Express, 17(14):11267–11273, Jul 2009.
- [50] A. Kirsch. An introduction to the mathematical theory of inverse problems, volume 120 of Applied Mathematical Sciences. Springer-Verlag, New York, 1996.
- [51] A. Kirsch and A. Lechleiter. The limiting absorption principle and a radiation condition for the scattering by a periodic layer. SIAM J. Math. Anal. (submitted), 2017.
- [52] K. Kodaira. The eigenvalue problem for ordinary differential equations of the second order and Heisenberg's theory of S-matrices. Amer. J. Math., 71:921–945, 1949.
- [53] V. A. Kozlov, V. G. Maz'ya, and J. Rossmann. Elliptic boundary value problems in domains with point singularities, volume 52 of Mathematical Surveys and Monographs. American Mathematical Society, Providence, RI, 1997.
- [54] N. Lindenmann, G. Balthasar, D. Hillerkuss, R. Schmogrow, M. Jordan, J. Leuthold, W. Freude, and C. Koos. Photonic wire bonding: a novel concept for chip-scale interconnects. *Opt. Express*, 20(16):17667–17677, Jul 2012.
- [55] J.-L. Lions and E. Magenes. Non-homogeneous boundary value problems and applications. Vol. I. Springer-Verlag, New York-Heidelberg, 1972. Translated from the French by P. Kenneth, Die Grundlehren der mathematischen Wissenschaften, Band 181.
- [56] J.-L. Lions and E. Magenes. Non-homogeneous boundary value problems and applications. Vol. II. Springer-Verlag, New York-Heidelberg, 1972. Translated from the French by P. Kenneth, Die Grundlehren der mathematischen Wissenschaften, Band 182.
- [57] L. Liu, Y. Qin, Y. Xu, and Y. Zhao. The uniqueness and existence of solutions for the 3-D Helmholtz equation in a stratified medium with unbounded perturbation. *Math. Methods Appl. Sci.*, 36(15):2033–2047, 2013.

- [58] R. Magnanini and F. Santosa. Wave propagation in a 2-D optical waveguide. SIAM J. Appl. Math., 61(4):1237–1252, 2000/01.
- [59] W. McLean. Strongly elliptic systems and boundary integral equations. Cambridge University Press, Cambridge, 2000.
- [60] K. A. Michalski and D. Zheng. Electromagnetic scattering and radiation by surfaces of arbitrary shape in layered media. i. theory. *IEEE Transactions on Antennas and Propagation*, 38(3):335–344, Mar 1990.
- [61] K. A. Michalski and D. Zheng. Electromagnetic scattering and radiation by surfaces of arbitrary shape in layered media. ii. implementation and results for contiguous half-spaces. *IEEE Transactions on Antennas and Propagation*, 38(3):345–352, Mar 1990.
- [62] P. Monk. Finite element methods for Maxwell's equations. Numerical Mathematics and Scientific Computation. Oxford University Press, New York, 2003.
- [63] L. Nannen and A. Schädle. Hardy space infinite elements for Helmholtz-type problems with unbounded inhomogeneities. *Wave Motion*, 48(2):116–129, 2011.
- [64] A. I. Nosich and V. P. Shestopalov. Radiation conditions and uniqueness theorems for open waveguides. *Radiotekhn. i Èlektron.*, 33(12):2483–2491, 1988.
- [65] C. Pekeris. Theory of propagation of explosive sound in shallow water. Geol. Soc. Am. Mem., 27, 1948.
- [66] B. A. Plamenevskij. Elliptic boundary value problems in domains with piecewise smooth boundary. In *Partial differential equations*, *IX*, volume 79 of *Encyclopaedia Math. Sci.*, pages 217–281. Springer, Berlin, 1997. Translated from the Russian by the author.
- [67] M. Reed and B. Simon. Methods of modern mathematical physics. I. Functional analysis. Academic Press, New York-London, 1972.
- [68] M. Reed and B. Simon. Methods of modern mathematical physics. II. Fourier analysis, self-adjointness. Academic Press, New York-London, 1975.
- [69] B. Shen, P. Wang, R. Polson, and R. Menon. An integrated-nanophotonics polarization beamsplitter with $2.4 \times 2.4 \mu m^2$ footprint. *Nat Photon*, 9(6):378–382, Jun 2015. Letter.
- [70] A. W. Snyder and J. Love. Optical Waveguide Theory. Springer, 1st edition, 1983.
- [71] A. Sommerfeld. Die Greensche Funktion der Schwingungsgleichung. Deutsche Math.-Ver., 21:309–353, 1912.
- [72] L. Tartar. An introduction to Sobolev spaces and interpolation spaces, volume 3 of Lecture Notes of the Unione Matematica Italiana. Springer, Berlin; UMI, Bologna, 2007.
- [73] E. C. Titchmarsh. Eigenfunction expansions associated with second-order differential equations. Part I. Second Edition. Clarendon Press, Oxford, 1962.

- [74] A. Tonnoir. Conditions transparentes pour la diffraction d'ondes en milieu èlastique anisotrope. PhD thesis, ENSTA ParisTech, 2015.
- [75] A. Tonnoir, A.-S. Bonnet Ben-Dhia, and S. Fliss. The halfspace matching method: a general method to solve scattering problems in infinite media. *to be submitted*.
- [76] R. Weder. Spectral and scattering theory for wave propagation in perturbed stratified media, volume 87 of Applied Mathematical Sciences. Springer-Verlag, New York, 1991.
- [77] H. Weyl. über gewöhnliche Differentialgleichungen mit Singularitäten und die zugehörigen Entwicklungen willkürlicher Funktionen. Math. Ann., 68(2):220–269, 1910.
- [78] C. H. Wilcox. Sound propagation in stratified fluids, volume 50 of Applied Mathematical Sciences. Springer-Verlag, New York, 1984.
- [79] Y. Xu. Scattering of acoustic waves by an obstacle in a stratified medium. In Partial differential equations with real analysis, volume 263 of Pitman Res. Notes Math. Ser., pages 147–168. Longman Sci. Tech., Harlow, 1992.
- [80] Y. Xu. Radiation condition and scattering problem for time-harmonic acoustic waves in a stratified medium with a nonstratified inhomogeneity. *IMA J. Appl. Math.*, 54(1):9–29, 1995.
- [81] A. I. Zayed. Handbook of function and generalized function transformations. Mathematical Sciences Reference Series. CRC Press, Boca Raton, FL, 1996.